



PhD course in

"Science and Management of Climate Change"

 34^{th} cycle

Thesis title

Ocean-atmosphere interaction at the Gulf Stream sea surface temperature front: variability and impacts on midlatitude atmospheric circulation

Author:

Luca Famooss Paolini

Supervisor:

Alessio Bellucci

Co-supervisors:

Panos Athanasiadis Paolo Ruggieri

Abstract

Recent observational and modelling studies show that the strong sea surface temperature (SST) gradients (oceanic fronts) associated with the western boundary currents affect the atmospheric circulation both on local and large spatial scale and on different timescales. Despite these advancements, the nature of the oceanatmosphere interactions associated with oceanic fronts variability is still not fully understood. Indeed, the character of the atmospheric response to oceanic fronts variability has not been established yet. Furthermore, the effective role of the atmospheric forcing on the oceanic fronts variability is unclear.

The present PhD thesis analyses the ocean–atmosphere interactions associated with the Gulf Stream SST front (GSF) variability. Specifically, the first part of the thesis assesses the character of the atmospheric response to the interannual GSF meridional shifts and its dependence on the model horizontal resolution. The second part of the thesis assesses the spectral features of the North Atlantic Oscillation (NAO)–GSF interaction and the mechanims through which the NAO forces the GSF meridional shifts on the decadal timescale.

The character of the atmospheric response to the interannual GSF meridional shifts was assessed in the ERA5 reanalysis dataset and in an ensemble of multimember atmosphere-only historical simulations forced with observed SSTs (1950– 2014). The role of the model horizontal resolution was assessed analyzing models with the horizontal resolution from 25 km to 100 km. Results show that the atmospheric response is strongly resolution dependent, with the response in the high-resolution simulations (resolution finer than 50 km) resembling the observed anomalies. More specifically: (i) analysis of the atmospheric thermodynamic balance close to the GSF showed that the anomalous diabatic heating associated with the GSF displacement is mainly balanced in the atmosphere by vertical motion and by meridional transient eddy heat transport (not the case for low-resolution models), while (ii) the large-scale response includes a meridional shift of the North Atlantic eddy-driven jet and stormtrack homo-directional to the GSF displacement. This atmospheric response is accompanied by changes in low-level baroclinicity close to and north of the GSF, resulting from the oceanic forcing and the zonal atmospheric circulation anomalies respectively. The low-level baroclinicity anomalies lead to changes in baroclinic eddy activity and, ultimately, in the jet via eddy-mean flow interaction.

The spectral features of the NAO–GSF interaction in the past decades and the mechanisms underlying this interaction were assessed in a set of global atmosphere and ocean reanalyses. Results show that the NAO and the GSF indices covary on the decadal timescales but only during 1972–2018. A secondary peak in the NAO– GSF covariability emerges on multiannual timescales but only for a limited period of time (2005–2015). The non-stationarity in the decadal NAO–GSF covariability is also manifested through the dependency of their lead-lag relationship on the analyzed time period. Indeed, the NAO leads the GSF shifts by 3 years during 1972–1990 and by 2 years during 1990–2018. Results show that the lag between the NAO forcing and the GSF response on decadal timescales can be interpreted as the joint effect of the fast response of wind-driven oceanic circulation, the lagged response of deep oceanic circulation, and the propagation of Rossby waves. However, not all the mechanisms are stationary. There is evidence of Rossby wave propagation only before 1990. Here it is suggested that the non-stationarity of the Rossby wave propagation causes the time lag between the NAO and the GSF latitudinal position on decadal timescales to differ before and after 1990.

Considering the impact that the GSF variability has on the North Atlantic vari-

ability, this PhD thesis helps to improve our understanding of extratropical climate variability.

Acknowledgements

I decided to start this PhD at the age of 28 years old, after some years of work in private companies in a completely different field. An hard decision. Especially in a cultural context like the Italian one, where leaving a permanent, well-paying but unsatisfying job for the unstable, poorly paid yet passionate scientific research is simply and hastily considered crazy. At least, this was the most common reaction around me at that time.

After 4 years of "studio matto e disperatissimo", sleepless nights spent in front of the black screen of Python, moments of stellar enthusiasm where one thinks to be able to publish in Nature journal followed by moments of dark and gloomy discouragement where one wonders what sense it makes to "play scientist", I can say that choosing that unstable, poorly paid yet passionate scientific research was one of the most important and right decisions of my life.

I would like to thank my supervisor Alessio Bellucci and my co-supervisors Paolo Ruggieri and Panos Athanasiadis, who made it possible to make this choice a reality. The tense moments faced in the past years within the group certainly cannot erase the invaluable help they have given me with scientific and non-scientific discussions, lunches, beers, phone calls, "psicological support", comments, corrections and advice. All of this has allowed me to grow not only as a scientist but also, and more importantly, as a person.

Furthermore, I thank Noureddine Omrani at the University of Bergen. In him, I have found not only one of the brightest and most passionate scientists of today but also a person with whom to share a friendship. At the University of Bergen, I also thank Noel Keenlyside. It is especially thanks to him that I managed to spend my surprising period abroad in Bergen. Few words shared with him were able to help me in my research more than reading tens of scientific papers.

In addition, it is essential to thank external reviewers Gerard McCarthy and Fumiaki Ogawa who, with their punctual and valuable comments, have made this thesis a significantly better work.

Finally, I want to thank all the other people (scientists and non-scientists) that I have met, and in some cases "lost", along this amazing journey. Rome (yes, it is like a person) and all the people that represent my roots, friends and colleagues at CMCC in Bologna, my PhD classmates, Venice (again, other human identity) and my new Venetian family, Duegradi's friends, Bergen community, etc. With their presence and their passing, they have given me the necessary energy to complete this experience.

Declaration

I confirm that this is my own work and the use of all material from other sources has been properly and fully acknowledged. Part of this work has been published before submission of the thesis¹.

¹Famooss Paolini, L., Athanasiadis, P. J., Ruggieri, P., Bellucci, A. (2022). The Atmospheric Response to Meridional Shifts of the Gulf Stream SST Front and Its Dependence on Model Resolution, *Journal of Climate*, 35(18), 6007-6030

Contents

\mathbf{A}	Abbreviations				
1	Motivation of work			12	
2	Inti	troduction			
	2.1	The G	ulf Stream system and the North Atlantic circulation	16	
	2.2	The w	estern intensification	20	
	2.3	The N	orth Atlantic Oscillation	22	
	2.4	2.4 The Gulf Stream SST front impact on the atmospheric circulation		32	
		2.4.1	Oceanic-forcing mechanisms and impact on the time-mean at-		
			mospheric circulation	33	
		2.4.2	Impact on the atmospheric variability	42	
3	$Th\epsilon$	The atmospheric response to meridional shifts of the Gulf Stream			
	SST front and its dependence on model resolution		45		
	3.1	Introd	uction	45	
	3.2	2 Data and methodological approach		48	
		3.2.1	Data	48	
		3.2.2	Gulf Stream sea surface temperature front shift	50	
		3.2.3	Methods	54	
		3.2.4	SST front-following coordinate system	55	
	3.3	Atmos	spheric response to Gulf Stream SST front shift	56	

	3.4	Heat budget	65						
	3.5	Discussion	76						
	3.6	Chapter conclusions	81						
4	Nor	n-stationarity in the NAO-Gulf Stream SST front interaction	86						
	4.1	Introduction	86						
	4.2	Data and methodological approach	93						
		4.2.1 Data	93						
		4.2.2 Methods	94						
	4.3	Results	99						
	4.4	Discussion	117						
	4.5	Chapter conclusions	125						
5	Cor	onclusions 12							
Re	References 132								

Abbreviations

AC	Azores current
AGCM	atmospheric general circulation model
AMM	Atlantic meridional mode
AMO	Atlantic multidecadal oscillation
AMOC	Atlantic Meridional Overturning Circulation
AO	Arctic Oscillation
CMA	cold meridional advection
DWBC	deep western boundary current
eGSI	extended Gulf Stream index
EHF	eddy heat flux
ENSO	El Niño–Southern Oscillation
GCM	general circulation model
GS	Gulf Stream
GSF	Gulf Stream sea surface temperature front
GSI	Gulf Stream index

GSNW	Gulf Stream North Wall
GST	Gulf Stream temperature
HighResMIP	High Resolution Model Intercomparison Project
IL	Icelandic Low
LC	Labrador Current
LS	Labrador Sea
LSW	Labrador Sea water
MABL	marine atmospheric boundary layer
MEHF	meridional eddy heat flux
MET	meridional eddy heat transport
MVA	mean vertical advection
NAC	North Atlantic current
NAM	Northern Hemisphere Annular Mode
NAO	North Atlantic Oscillation
NRG	northern recirculation gyre
PDF	probability density function
PV	potential vorticity
SHF	surface heat fluxes
SLP	mean sea-level pressure
SPG	subpolar gyre
SRG	southern recirculation gyre

SSH	sea surface height
SST	sea surface temperature
STG	subtropical gyre
WBC	western boundary current
WMA	warm meridional advection
WRF	Weather Research and Forecasting

| Chapter _

Motivation of work

The seminal work by Frankignoul and Hasselmann (1977) has shown that the largescale, low-frequency upper ocean variability in the extratropics can be largely interpreted as the oceanic integral response to short timescale randomic atmospheric disturbances. These results have led to the emergence of the well-known paradigm of white-noise atmospheric forcing and red-noise oceanic passive response, which has been an important paradigm to interpret the low-frequency sea surface temperature (SST) variability in middle latitudes during the last 40 years. As an example, this paradigm has been used to explain the SST tripole variability as the oceanic response to the North Atlantic Oscillation (NAO; Hurrell, 1995) forcing via surface heat fluxes (SHF) and Ekman currents (Cayan, 1992; Deser et al., 2010). On the other hand, the ocean has been shown to exert a weak effect on the short timescale atmospheric variability, with the ocean-induced atmospheric response projecting onto intrinsic atmospheric variability modes (Rodwell et al., 1999; Czaja and Frankignoul, 2002; Kushnir et al., 2002; Peng et al., 2003; Cassou et al., 2007).

However, several observational and modelling studies have shown that this interpretation of air-sea interaction in the extratropics is altered by the presence of meso-scale oceanic features, such as the western boundary currents (WBCs; Chelton et al., 2001, 2004; Xie, 2004; Bellucci et al., 2021). In these areas, a great part of the oceanic variability is induced by intrinsic oceanic processes, such as advection and diffusion (Kelly et al., 2010; Kwon et al., 2010; Minobe et al., 2010; Patrizio and Thompson, 2021). Furthermore, the WBCs are characterized by strong SST gradients (oceanic fronts), which strongly influence the atmospheric circulation within the marine atmospheric boundary layer (MABL; Chelton et al., 2001; Small et al., 2008). The impact of the oceanic fronts extend also beyond the MABL, influencing the atmospheric circulation on the large-scale and on different timescales (O'Reilly and Czaja, 2015; Sato et al., 2014; Wills et al., 2016; Joyce et al., 2019). These results show that the oceanic fronts may be acting as a source of atmospheric predictability on a wide range of timescales (Joyce et al., 2019; Athanasiadis et al., 2020). The oceanic fronts are also important because their presence shapes many features of the time-mean atmospheric circulation. In particular, they anchor zones of intense time-mean upward motion thanks to different pressure conditions across the fronts themselves (Minobe et al., 2008a). In addition, they shape the extratropical stormtracks and eddy-driven jets maintaining the near-surface baroclinicity necessary for baroclinic instability growth (Sampe et al., 2010). Finally, they affect the time-mean stratospheric circulation enhancing the upward wave propagation (Omrani et al., 2019).

Despite these advancements, the nature of the ocean-atmosphere interactions associated with oceanic fronts variability is still not fully understood. Indeed, the character of the atmospheric response to oceanic fronts variability has not been established yet. Furthermore, the effective role of the atmospheric forcing on the oceanic fronts variability is unclear. This is especially true for the SST front associated with the Gulf Stream (GS), on which this PhD thesis focuses. As one of the major WBCs, the GS plays a key role in driving the northern hemisphere climate and variability (Minobe et al., 2008a; Omrani et al., 2019). Therefore, improving our understanding of its variability and interaction with the atmosphere is highly relevant for the comprehension of extratropical climate variability on a wide range of timescales.

Regarding the ocean-to-atmosphere forcing, there are observational studies showing that positive SST anomalies in the GS region induce positive NAO-like anomalies in the atmospheric circulation (Sato et al., 2014; Joyce et al., 2019), whereas other studies show atmospheric circulation anomalies of opposite sign (Kwon and Joyce, 2013; Wills et al., 2016). Taking into account that a positive (negative) NAO phase is associated with positive (negative) SST anomalies in the GS region (Frankignoul et al., 2001), the first results shed light on the possible existence of a positive feedback between the NAO and the GS SST front (GSF) variability, whereas the second results show an opposite relationship between them. These differences may be induced either by discrepancies in the methodological approach or by the chaotic nature of the atmospheric variability, especially on timescale shorter than the interannual one.

The use of general circulation models (GCMs) could help to disentangle this issue, increasing the signal-to-noise ratio associated with the phenomena of interest. For example, ad-hoc simulations can be performed with observed SSTs in the area of interest and climatological SSTs elsewhere, thus reducing possible impact from remote SST variability. Using multi-member simulations also helps in letting the forced signal emerge from the chaotic atmospheric variability. Furthemore, simulations performed with atmosphere-only models allow to focus on the single ocean-to-atmosphere direction of the two-way interaction between the two realms. However, important inter-model discrepancies still emerge in studies analysing the impact of the oceanic fronts variability on the atmospheric circulation (Czaja et al., 2019). In this context, there are indications that the horizontal resolution may play a role (Smirnov et al., 2014), highlighting the need for models with grids adequately resolving the oceanic fronts.

Regarding the atmosphere-to-ocean forcing, the variability of the GSF has been mainly linked with the NAO (Taylor and Stephens, 1998; Wolfe et al., 2019), which is the dominant mode of variability of the surface atmospheric circulation in the North Atlantic region (Hurrell, 1995). Interestingly, the discrepancies in the lead– lag relationship between the NAO forcing and the GSF response suggested in the literature show the possible existence of non-stationarity in their interaction. Such non-stationarity may explain why there is also a lack of consensus on the mechanisms through which the atmosphere influences the GSF latitudinal position (Rossby and Benway, 2000; Zhang and Vallis, 2007; Sasaki and Schneider, 2011; Gangopadhyay et al., 2016).

The present PhD thesis analyses the ocean–atmosphere interactions associated with the GSF variability, addressing the following scientific questions:

- 1. What is the character of the atmospheric response to the interannual GSF meridional shifts and its dependence on the model horizontal resolution?
- 2. What are the spectral features of the NAO–GSF interaction and the mechanisms through which the NAO forces the GSF meridional shifts on the decadal timescale?

Specifically, Chapter 2 gives a general introduction on the GS system and the North Atlantic circulation. Then, it provides a brief description of the NAO and an overview about the role of the GSF on the time-mean atmospheric circulation and the atmospheric variability. Chapter 3 responds to the first question above, assessing the atmospheric response to the interannual meridional shift of the GSF and its dependence on model resolution. Chapter 4 responds to the second question above, assessing the spectral features of the NAO–GSF interaction and the mechanisms through which the NAO may be forcing changes in the GSF latitudinal position on the decadal timescale. Finally, Chapter 5 summarizes the most important findings of the PhD thesis.

Chapter 2

Introduction

2.1 The Gulf Stream system and the North Atlantic circulation

The GS is the WBC located in the North Atlantic basin and it is an essential component of northern hemisphere climate. Indeed, it constitutes the upper limb of the Atlantic Meridional Overturning Circulation (AMOC) and it transports huge amounts of warm and salty water poleward, partly balancing the heat disequilibrium between the tropics and the higher latitudes (Peixoto and Oort, 1992). Once the GS reaches higher latitudes, the heat gets partly released into the atmosphere, which becomes predominant in redistributing the heat over extra-tropical and polar areas (Fasullo and Trenberth, 2008; Yang et al., 2015). The ocean-to-atmosphere heat exchanges are particularly intense in the winter season between Cape Hatteras (75°W–37°N) and the area south of the Grand Banks, where the GS enters in contact with cold, dry air masses coming from the North America continent (Kallberg et al., 2005; Yu and Weller, 2007).

The intense ocean-to-atmosphere heat exchanges in the western North Atlantic determine large-scale asymmetry in the horizontal distribution of surface diabatic heating within the subtropical gyre (STG). The STG is the large-scale anticyclonic oceanic circulation mainly driven by the wind forcing between the equator and mid-latitudes in the North Atlantic (Sverdrup, 1947). This aspect is fundamental in setting the mid-latitude stationary and quasi-stationary atmospheric eddies (Smagorinsky, 1953; Ambaum and Athanasiadis, 2007), thus affecting the climate at extratropics. Furthermore, the GS plays an important role in northern hemisphere climate because it is characterized by a strong SST gradient (oceanic front), which can affect the atmospheric circulation both on local and large spatial scale and on different timescales (Minobe et al., 2010; Sampe et al., 2010; Brayshaw et al., 2011; Joyce et al., 2019). Refer to section 2.4 for more details about the impact of GS front on the atmospheric circulation.

The GS originates in the Gulf of Mexico and it flows northward along the eastern coast of North America until Cape Hatteras. Then, it detaches from the coast and turns northeastward, flowing along the continental shelf break until the Grand Banks. The location of the GS detachment from the coast has been explained through several mechanisms, such as the position of zero wind-stress curl (strictly linked with the zero Sverdrup transport; Stommel, 1948; Munk, 1950), the outcropping of isopycnal surfaces (Parsons, 1969; Veronis, 1973) and the interaction with adjacent currents (Ezer and Mellor, 1992). Furthermore, the position of the separation point has been shown to be controlled by local factors such as the coastline geometry, the bottom topography and the interaction with the deep western boundary current (DWBC; Thompson and Schmitz, 1989; Spall, 1996a; Tansley and Marshall, 2000; Zhang and Vallis, 2007).

Once detached from the coast, the GS starts meandering around 70°W and an intense eddy activity develops up to the Grand Banks (Ducet et al., 2000). Some of the large-amplitude, propagating meanders can give rise to warm (cold) core rings north (south) of the time-mean GS path (Brown et al., 1986; Cornillon, 1986; Cornillon et al., 1987).

Near 50°W, the GS decreases its transport and branches into several current bands. The main ones are the northern recirculation gyre (NRG; Hogg et al., 1986), the southern recirculation gyre (SRG; Worthington, 1976), the Azores current (AC; Krauss et al., 1990) and the North Atlantic current (NAC; Krauss, 1986). The NRG (SRG) is a barotropic cyclonic (anticyclonic) recirculation pattern, located north (south) of the GS (Hogg, 1992). The NRG and SRG recirculate part of the GS water westward, rejoining the main stream at different longitudes downstream Cape Hatteras (Halkin and Rossby, 1985; Johns et al., 1995). Because of these two recirculation gyres, the GS transport almost doubles between the separation point and the Grand Banks, going from 88 Sv at 73°W (Halkin and Rossby, 1985) to 150 Sv at 60°W (Hogg, 1992). The AC is the branch of the GS that flows eastward after the Grand Banks and it represents the eastern side of the STG (Sverdrup, 1947). Due to its position, this current separates the subtropical water to the south from the subpolar water to the north (Krauss et al., 1990). The GS water that is not recirculated westward by the NRG and SRG and does not proceed eastward along the AC, flows northward off the eastern flank of the Grand Banks and then turns abruptly eastward around 52°N. This branch of the GS is the NAC and it transports warm and saline water to much higher latitudes than currents in any other ocean basin. The transport of warm water reaches the Nordic Seas, thus playing a fundamental role in determining the mild climate over northern Europe (Krauss, 1986). The NAC flows poleward via three main channels: the Irminger Current and the subpolar front west and east of Iceland respectively, and the eastern NAC flowing in the eastern side of North Atlantic. The rest of the NAC water interacts with Arctic Sea water and turns westward, flowing along the coast of Greenland, as Eastern and Western Greenland currents, and then along the coast of eastern North America, as Labrador Current (LC). These currents form a large-scale cyclonic circulation pattern called subpolar gyre (SPG), which is also driven by the prevailing winds (Sverdrup, 1947).

Within the SPG, the heat loss from the ocean to the atmosphere and the interaction between regional oceanic currents facilitate vertical mixing and convection in the ocean (Marshall and Schott, 1999; Lazier et al., 2001). Once the deep water is formed, it is partly recirculated within the Labrador Sea (LS) and along the NAC and partly travels southward. The deep water is transported southward via the DWBC (Stommel, 1958), flowing from the LS along the continental shelf of the North American sea-board and then passing underneath the GS at the crossover region (near Cape Hatteras; Spall, 1996a,b). This current has been shown to strongly interact with the GS, affecting the position of the separation point and the GS path downstream Cape Hatteras (Zhang and Vallis, 2007; Chassignet and Marshall, 2008). Furthermore the deep water is transported southward via interior pathways which pass underneath the NAC close to the Grand Banks (Bower et al., 2009, 2019). Differently from what was thought before, these interior pathways have been shown to be the principal conduits through which the ventilated water is transported from the SPG to the STG (Getzlaff et al., 2006; Bower et al., 2011). The DWBC and the interior pathways constitute the deep limb of the AMOC and then they constitute the deep branch of the meridional–vertical thermohaline circulation of the Atlantic ocean (Lozier, 2010).



Figure 2.1: Maps of the main oceanic currents connected with the GS. The figure is reproduced from Imawaki et al. (2013).

2.2 The western intensification

The GS is the western portion of STG. The STG is a large-scale oceanic circulation pattern mainly driven by the stress exerted by the prevailing winds on the sea surface (Sverdrup, 1947). The same is true for other ocean gyres in the globe. In the North Atlantic, the prevailing winds are westerlies at mid-latitudes and easterlies (the trades) at tropics. Then, the ocean is forced to have an anticyclonic circulation between the equator and around 40°N because of the negative wind-stress torque.

Wind-driven ocean gyres are asymmetric in the zonal direction, showing narrow, deep and strong poleward currents in their western side (i.e. the GS in the North Atlantic) and broad, shallow and weaker equatorward currents in their eastern side. The intensification of currents along the western side of the ocean gyres has been referred to as "western intensification" and it is caused by the variation of Coriolis force with latitude (Stommel, 1948) and the friction exerted by the coasts (Munk, 1950). This section provides a brief description of the theory explaining the western intensification.

Under the assumption of barotropic flow in a rotating system, starting from the momentum and the continuity equations it is possible to obtain the following potential vorticity (PV) equation:

$$\frac{\mathrm{d}}{\mathrm{d}t} \left(\frac{\xi + f}{h} \right) = \frac{F}{h} \tag{2.1}$$

Here $\boldsymbol{\xi}$ represents the relative vorticity; f is the planetary vorticity; h is the height of the water column; F is the forcing and the dissipation of vorticity, while the quantity inside the parenthesis is the PV. The equation 2.1 states that the PV of a column of water is conserved following the motion in the absence of PV sources and sinks on the fluid (i.e. F is equal to zero).

In the subtropical area, the midlatitude westerlies and the tropical easterlies induce southward and northward Ekman transport, respectively. Here, the Ekman transport is the net motion of ocean water within the Ekman layer as the result of a balance between Coriolis and turbulent drag forces exerted by the wind-stress. In the northern hemisphere, the Ekman transport is oriented about 90° from wind direction. For the balance of water mass, the convergence in the interior of the SPG determines downward motion, i.e. the squeezing of the water column (Ekman pumping). Being the forcing and the dissipation of PV zero (or almost) in the interior of the STG, the PV must be conserved (equation 2.1). Thus, the squashing of the water column is balanced by changes in planetary vorticity as the relative vorticity is small in the interior of the STG. This means that the water column moves southward. The reverse is true over the SPG, where the wind-stress curl induces divergence in the interior of the gyre, stretching of the water column and poleward motion. Generally speaking, this means that the large-scale wind-stress forcing tends to induce changes in the height of column water in the interior of ocean gyres, which are balanced by meridional displacement. This relation is referred to as Sverdrup balance and it is generally expressed as follows:

$$\beta \overline{v} = f \frac{\partial \omega}{\partial z} = f \frac{\partial (curl(\frac{\tau}{f\rho}))}{\partial z}$$
(2.2)

Here β is the linear variation of the Coriolis parameter f; $\overline{\nu}$ is the meridional velocity vertically integrated over the ocean depth; τ is the wind-stress exerted on the sea surface; ρ is the water density.

Once the water column has reached a lower latitude in the SPG, it must be recirculated back in order to balance the total mass. Given the Sverdrup balance is valid for the large-scale oceanic circulation, the return flow must be in a narrow poleward current. From the view of mass conservation, the poleward current can be either in the form of a western boundary current or an eastern boundary current. However, only the western boundary current can close the vorticity balance over the oceanic basin. Indeed, while the water column travels southward within the SPG, it acquires negative relative vorticity because of the large-scale wind forcing. Then, positive relative vorticity must be injected into the fluid in order to counterbalance the negative PV due to the wind-stress curl. The injection of positive relative vorticity is provided by the friction that acts as a dissipation term in the equation 2.1 (F different from zero). Specifically, Stommel (1948) has added friction to the bottom in the form of linear drag to balance the PV close to the boundaries. However, it is hard for the wind-driven oceanic circulation to reach the ocean bottom. For this reason, Munk (1950) has added friction along the side walls using a harmonic Newtonian viscosity. Because of the friction along the western sidewalls, the velocity parallel to the wall is zero along the coastlines and increases offshore. Then, an input of positive relative vorticity is provided to the western boundary of the basin, thus balancing the PV and recirculating the water column poleward. On the other hand, such a closure of the circulation could not be possible on the eastern boundary, because in this case the friction provides a further input of negative PV. Refer to Figure 2.2 for a schematic of the vorticity balance at the western and eastern boundary because of the sidewall friction, as in the Munk's model.

As specified above, the western boundary current intensification is present in all the main ocean gyres over the globe. This means that the oceanic currents intensify along the western portion of the ocean independently from the wind-stress curl sign. This is due to the Earth's sphericity (beta effect). Indeed, Stommel (1948) has shown that, in the case of a no-rotating or uniformly rotating ocean, the western intensification disappears and the ocean currents show an east-west symmetric circulation centered respect the oceanic basin.

2.3 The North Atlantic Oscillation

As a result of intrinsic oceanic variability as well as atmospheric forcing, the GS undergoes meridional shifts, meandering and changes in mass/heat transport (Kelly et al., 2010). Among other atmospheric drivers, the NAO has been shown to play a fundamental role in forcing GS variability on different timescales (Taylor et al., 1998; Joyce et al., 2000; Gangopadhyay et al., 2016). Being this PhD thesis focused on the ocean–atmosphere interaction associated with the GSF variability, the NAO represents a key atmospheric process to properly understand the results described



Figure 2.2: Vorticity balance at the western and eastern boundary because of the sidewalls friction (Munk's model). The figure is reproduced from Talley et al. (2011).

in following sections. For this reason, a brief description of the NAO is provided in this chapter.

The NAO is a large-scale seesaw of pressure between the high-latitude and subtropical North Atlantic and represents the dominant mode of atmospheric lowfrequency variability over the Euro-Atlantic region as it explains about the 30% of the annual atmospheric variability over that region (Jianping and Wang, 2003; Hurrell and Deser, 2010). This percentage increases up to about 36% during winter season, whereas it decreases to about 22% during summer (Barnston and Livezey, 1987; Hurrell et al., 2003).

The NAO is considered to be in its positive (negative) phase when the pressure gradient between the subtropical and the northern region is increased (reduced). Consistently, the positive NAO phase is associated with negative and positive mean sea-level pressure (SLP) anomalies in respect to the atmospheric mean state over the subpolar and subtropical North Atlantic, respectively (Figure 2.3). The reverse is true for the negative NAO phase. The pressure anomalies associated with the NAO are equivalent barotropic (high coherence along the vertical) and extend from the surface up to the lower stratosphere with amplitude increasing with height.



Figure 2.3: Leading empirical orthogonal function of the seasonal SLP anomalies in the North Atlantic sector $(20^{\circ}-70^{\circ}N, 90^{\circ}W-40^{\circ}E)$ during winter (DJF), spring (MAM), summer (JJA) and autumn (SON). The percentage of the total variability the SLP anomalies explain in North Atlantic sector is shown on the top of each subplot. The figure is reproduced from Hurrell et al. (2003).

The spatial pattern of the NAO has been shown to be non-stationary throughout the year, with different tilting and position of the centers of action (i.e. where the pressure anomalies are the most intense) during the four seasons (Figure 2.3; Hurrell and Deser, 2010). Furthermore, the spatial pattern of the NAO has been shown to be asymmetrical between its positive and negative phase (Hurrell et al., 2003; Fabiano et al., 2020). Specifically, the positive NAO phase exhibits a northwest– southeast tilting, whereas the negative NAO phase exhibits a northeast–southwest tilting (Figure 2.4).



Figure 2.4: Main patterns of the positive (left) and negative (right) NAO phase. These patterns have been extracted through a K-means clustering algorithm applied to winter (DJF) geopotential height anomalies at 500 hPa, using ERA40 (1957–1978) and ERAInterim (1979–2014 data. The figure is a partial reproduction of Figure 1 in Fabiano et al. (2020).

In describing the spatial pattern of the NAO, it is highlighted that the pressure anomalies extend also over the Aleutian Low (North Pacific). Because of that, it has been (and it is still) debated whether the NAO should be considered a specific mode of variability or a regional manifestation of the hemispheric mode of atmospheric variability known as the Arctic Oscillation (AO) or Northern Hemisphere Annular Mode (NAM; Thompson and Wallace, 2000; Deser, 2000; Ambaum et al., 2001). Apart from some subtle differences over the Euro-Atlantic region, the main difference between the AO/NAM and the NAO is the greatest amplitude of the SLP anomalies that characterise the AO/NAM over the North Pacific. These anomalies make the AO/NAM a (almost) zonally-symmetric mode of variability in the northern hemisphere. Some authors have suggested that the NAO should be considered as part of the AO/NAM, regionally modified by zonally-asymmetric forcings such as topography and land-sea contrast (Thompson and Wallace, 1998, 2000). Differently, other authors have suggested that the AO/NAM cannot be considered as a teleconnection pattern (Deser, 2000; Ambaum et al., 2001). For example, Deser (2000) has reached this conclusion showing that the Atlantic and Pacific centers of action are weakly correlated with each other and that the interannual SLP anomalies over the Arctic and Pacific sectors are not significantly correlated as they are between the Arctic and the North Atlantic.

The pressure anomalies associated with the NAO have a large impact on the Euro-Atlantic weather and climate, with important consequences for socio-economic and ecological systems (Drinkwater et al., 2003; Mysterud et al., 2003). For example, relevant in the context of this thesis are the NAO impacts on the SST, stormtrack and eddy-driven jet in North Atlantic region. Specifically, the NAO is associated with an SST tripolar pattern that extends to the entire North Atlantic (Figure 2.5). The SST tripole is characterized by negative (positive) SST anomalies in the SPG (STG), interpreted as the oceanic response to anomalous SHF and wind-driven oceanic circulation associated with NAO forcing. This SST tripole pattern represents an important portion of the extratropical SST variability on seasonal and interannual timescale (Cayan, 1992; Deser et al., 2010).

Furthermore, the NAO influences the position of stromtrack and eddy-driven jet in the North Atlantic. Very briefly, it is here indicated that the stormtrack is the region of high temporal variability on timescales of a few days, mainly linked with the synoptic systems that develop in the North Atlantic. The eddy-driven jet is a narrow band of strong westerly winds located around 30°–60°N, mainly driven by intense eddy activity along the stormtrack. The positive NAO is associated with enhanced and reduced eddy activity in the extratropical and subtropical North Atlantic, respectively (Figure 2.6, left panel). Consistently, the westerly winds are intensified at extratropics and reduced at tropics, thanks to the eddy momentum flux convergence/divergence due to transient eddies (Figure 2.6, right panel). The reverse



Figure 2.5: SST (°C; colour shadings), SLP (hPa; contours; contour interval is 1 hPa) and surface wind (m s⁻¹; arrows) anomalies associated with one positive standard deviation departure of the NAO index during winter. The NAO index has been defined as the station index of Hurrell et al. (2003). The figure is a reproduction of Figure 1a in Deser et al. (2010).

is true for the negative NAO. Such anomalies represent a northward (southward) shift of both stormtrack and eddy-driven jet during the positive (negative) NAO respect their climatological position.

The NAO has been shown to vary on a broad spectrum of timescales, showing enhanced variability on (sub-)monthly, interannual and decadal timescales (Feldstein, 2000; Bellucci et al., 2008; Reintges et al., 2017). As an example, I show here the power spectrum of the winter-mean NAO index defined as the principal component of the leading empirical orthogonal function of SLP data over the North Atlantic sector $(20^{\circ}-70^{\circ}N, 90^{\circ}W-40^{\circ}E;$ Hurrell et al., 2003, Figure 2.7). The wintermean NAO index shows a slightly red power spectrum, with power increasing with period. Specifically, there is an enhanced NAO variability at [2–3]-year timescale, followed by a reduced power at [3–6]-year timescale and a peak at the sub-decadal



Figure 2.6: Left panel: Regression of the variance of the meridional eddy velocity (as diagnostic for the stormtrack; $v'v'_{300}$; $m^2 s^{-2}$; color shadings) and geopotential height $(z_{300}; m; \text{ contours})$ at 300 hPa onto standardized NAO index during winter season (DJFM). The NAO index has been defined as the leading empirical orthogonal function of SLP data over the North Atlantic sector (20°-70°N, 90°W-40°E; green box). Contour intervals for the $v'v'_{300}$ and z_{300} regressions are $10 \text{ m}^2 \text{ s}^{-2}$ and $20 \,\mathrm{m}$ per NAO standard deviation, respectively. Positive (negative) z_{300} regression contours are black (gray) and the zero contour is bold. The magenta and cyan markers represent the position of the maximum and minimum in the z_{300} regressions, respectively. The magenta and cyan squares represent the position of the maximum and minimum in the z_{300} regressed onto the NAM index. The NAM index has been defined as the leading empirical orthogonal function of SLP data over the Northern Hemisphere (poleward of 20°N). Right panel: As in the left panel, but for $v'v'_{300}$ (m² s⁻²; color shadings) and zonal winds at 300 hPa (u₃₀₀; m s⁻¹; color shadings). Contour intervals for the $v'v'_{300}$ and u_{300} regressions are $10 \text{ m}^2 \text{ s}^{-2}$ and $2 \,\mathrm{m \, s^{-1}}$ per NAO standard deviation, respectively. Black (gray) contours in the u_{300} regressions represent westerly (easterly) anomalies. The black arrow indicates the latitude of maximum climatological u_{300} over a longitude range where the winter upper-tropospheric jet is well defined. The figure is a partial reproduction of Figure 7 and Figure 8 in Wettstein and Wallace (2010).

timescale. Some studies have also suggested that the NAO can vary at multidecadal timescales (Omrani et al., 2014; Peings and Magnusdottir, 2014; Woollings et al., 2015). However, the limited length of observational records makes it difficult to verify this peak.

For completeness, it is specified that the NAO index can also be defined with other approaches rather than the one presented above (Walker and Bliss, 1932; Wallace and Gutzler, 1981; Rogers, 1984; Hurrell, 1995; Jianping and Wang, 2003).



Power Spectrum NAO Index

Figure 2.7: Power spectrum of the winter NAO index over 1899–2022. The NAO index has been defined as the principal component of the leading empirical orthogonal function of SLP data over the North Atlantic sector $(20^{\circ}-70^{\circ}N, 90^{\circ}W-40^{\circ}E)$, as in Hurrell et al. (2003). The red noise spectrum and the 5% and 95% are also shown. The figure is reproduced from Hurrell and Deser (2010).

For example, another widely adopted approach in literature has been to define the NAO index as the normalized difference of SLP in two weather stations located at low- and high-latitudes (Azores–Iceland, Rogers (1984); Lisbon–Iceland, Hurrell (1995); Gibraltar–Iceland, Jones et al. (1997)). The NAO indices defined with these two approaches result highly comparable (Hurrell et al., 2003).

The NAO variability described above has been explained as the result of several processes, both internal and external to the atmosphere.

The high-frequency NAO variability has been suggested to rise from internal,

non-linear atmospheric processes. Feldstein (2000) has shown that the interaction between mean flow and transient eddies in North Atlantic region give rise to a peak in the NAO variability at sub-monthly timescale, with oscillations of about 10 days. Some authors have shown that the atmospheric random processes could also explain the low-frequency NAO variability (Stephenson et al., 2000; Thompson et al., 2003). In this context, the enhanced NAO variance at interannual and longer timescales should be understood as the remnant of an infinite number of highfrequency stochastic processes. This interpretation of the atmospheric variability has been referred to as the "climate noise paradigm" (Leith, 1973; Madden, 1976).

Another driver of NAO variability on short timescales has been suggested to be the stratosphere (Baldwin and Dunkerton, 1999, 2001). Indeed, it has been shown that the anomalies in the lower stratospheric circulation propagate downward and manage to affect the troposphere up to three weeks after. The stratospheric "downward control" on tropospheric variability is linked to the effect of stratospheric flow on the upward propagation of tropospheric planetary waves (Shindell et al., 2001; Ambaum and Hoskins, 2002). In this context, it is important to point it out that the lower stratospheric flow is affected by ozone, greenhouse gases and volcanic aerosol concentrations. These substances affect the meridional temperature gradient in the lower stratosphere and, ultimately, the polar vortex via radiative cooling/warming. Thus, changes in their concentration can play a role on the NAO variability by inducing stratospheric circulation anomalies that propagate downward in the troposphere (Graf et al., 1998; Shindell et al., 2001).

The spectral peak in NAO variability at interannual, decadal and longer timescales has been mainly interpreted as the result of air-sea interaction in the extratropics (Barsugli and Battisti, 1998; Rodwell et al., 1999; Czaja et al., 2003; Reintges et al., 2017). The huge thermal inertia of the ocean mixed layer provides a marked persistence to the SST anomalies induced by atmospheric forcing (Frankignoul and Hasselmann, 1977). Then, these anomalies can feed back on the atmospheric circulation via SHF (Hoskins and Karoly, 1981). In this context, it has been shown that the thermal damping of oceanic and atmospheric temperature anomalies is reduced when the atmosphere and ocean are coupled (Barsugli and Battisti, 1998). Thus, both the atmospheric and oceanic thermal variance are enhanced on timescales longer than the heat capacity of ocean mixed layer (several months). This paradigm has been referred to as the "reduced thermal damping" paradigm.

The SST anomalies forced by large-scale atmospheric circulation can also feed back on the atmospheric flow by changing the surface baroclinicity (Kushnir et al., 2002). Indeed, the SST patterns affect the surface temperature gradient in the atmosphere which is directly linked with baroclinicity. Thus, the changes in transient eddy activity force the atmospheric circulation via eddy—mean flow interaction (Hoskins et al., 1983; Novak et al., 2015).

The ocean plays an important role in enhancing the NAO variance on long timescales also through its dynamics. Several studies have shown that the NAO can force SST anomalies in the STG which are then transported northeastward along the NAC in a decade (Sutton and Allen, 1997; Krahmann et al., 2001). Furthermore, other studies have shown that the NAO influences the deep convection in the LS via anomalous SHF, with a consistent delayed response in the AMOC on the decadal timescale (Joyce et al., 2000; Eden and Willebrand, 2001; Bellucci et al., 2008; Reintges et al., 2017; Wills et al., 2019). Such dynamical modes of oceanic variability redden the spectrum of the SST in the North Atlantic and, ultimately, enhance the low-frequency NAO variability.

The NAO variability can be also driven by non-local SST anomalies (outside the extra-tropical North Atlantic; Trenberth et al., 1998; Robertson et al., 2000; Sutton et al., 2000; Czaja and Frankignoul, 2002). The SST patterns in tropical Pacific and Atlantic can affect the atmospheric convection and then the rate and the location of diabatic heating within the troposphere. Such diabatic heating force Rossby wave trains and/or changes in Hadley circulation which can condition the North Atlantic climate inducing atmospheric circulation anomalies partly projecting on the NAO (?).

Another external driver of the interannual NAO variability is the Northern Hemisphere cryosphere (Cohen, 2016; Wegmann et al., 2020). As an example, the sea-ice loss in the European Arctic during autumn season has been shown to be associated with upward Rossby wave propagation and a consistent weakening of stratospheric polar vortex (Cohen et al., 2014). These anomalies in the low-stratosphere propagate downward into the troposphere in following winter, determining negative NAO-like atmospheric circulation anomalies over the Euro-Atlantic sector (Kretschmer et al., 2018). A similar atmospheric response has been found for increased snow cover over eastern Eurasia during winter season (Ghatak et al., 2010). The changes in sea-ice and snow cover described above have been shown to be linked each other (Wegmann et al., 2015; Gastineau et al., 2017). Being the Barents-Kara sea-ice extension deeply affected by oceanic circulation in the North Atlantic, such anomalies in sea-ice extension over Euro-Atlantic sector and snow cover over Eurasia can also play a role on decadal NAO variability (Wegmann et al., 2020). Other studies have shown that anomalies in sea-ice over the LS and snow cover in North America also play a role on decadal NAO variability (Watanabe and Nitta, 1999; Kvamstø et al., 2004).

Finally, the decadal NAO variability has been shown to be also driven by the 11-year solar cycle, consistently with the stratospheric "downward control" on the tropospheric circulation discussed above (Thiéblemont et al., 2015).

2.4 The Gulf Stream SST front impact on the atmospheric circulation

The GS is a fundamental component of northern hemisphere climate because it strongly contributes to the meridional heat transport, partly balancing the differential solar input between tropics and high latitudes. Furthermore, the GS is fundamental in determining the horizontal distribution of surface diabatic heating, thus setting the midlatitude stationary and quasi-stationary atmospheric eddies. Finally, the GS plays an important role in northern hemisphere climate because it is characterized by a strong SST gradient (oceanic front). Indeed, there are pieces of evidence showing that the oceanic fronts, such as the one associated with the GS, affect the atmospheric circulation both on local- and large-scale and on different timescales (Minobe et al., 2010; Sampe et al., 2010; Brayshaw et al., 2011; Joyce et al., 2019). This fact is quite surprising because in contrast with the dominant paradigm of large-scale air-sea interaction at extratropics. Specifically, the low-frequency variability of extratropical large-scale SST has been mainly interpreted as passive oceanic response to stochastic high-frequency atmospheric forcing (Frankignoul and Hasselmann, 1977). At the same time extratropical large-scale SST anomalies have been shown to exert a weak impact on the atmosphere, with the ocean-induced atmospheric response essentially projecting onto intrinsic atmospheric variability modes (Kushnir et al., 2002). Several studies have shown that this interpretation of air-sea interaction in extratropics is not valid over meso-scale oceanic features such as the WBC (and their oceanic fronts) and the oceanic eddies (Chelton et al., 2001, 2004; Xie, 2004; Bellucci et al., 2021). Thus, taking into account the role of the atmospheric circulation in determining the existence of WBCs, a two-way interaction is expected between the atmosphere and the WBCs (oceanic fronts). Section 2.4.1 describes the main mechanisms through which the oceanic fronts force the atmospheric circulation. Furthermore, it shows the oceanic front impacts on the time-mean atmospheric circulation, with particular focus on the GSF. Then, section 2.4.2 provides some pieces of evidence of the GSF impact on the atmospheric variability.

2.4.1 Oceanic-forcing mechanisms and impact on the timemean atmospheric circulation

Vertical mixing mechanism

High-resolution satellite observations have highlighted small-scale patterns in the near-surface wind-stress curl and divergence co-located over meso-scale oceanic features such as the oceanic fronts (Figure 2.8; Chelton et al., 2001; Nonaka and Xie, 2003; Chelton et al., 2004; O'Neill et al., 2005; Small et al., 2008). The direct link between the near-surface wind and SST patterns over the oceanic fronts is better shown by the correspondence between the laplacian of the SST (or the SLP) and the wind divergence. Indeed, the laplacian operator acts as a spatial high-pass filter, unveiling the oceanic fronts effect which is generally masked by large-scale atmospheric circulations (Xie, 2004; Minobe et al., 2008a). In contrast to what is expected, the correlation between the near-surface wind speed and SST is positive over such oceanic fronts the SST patterns are driven by intrinsic oceanic processes rather than by heat fluxes from the atmosphere. Hence, over such oceanic features it is more the oceanic variability that affects the atmospheric one than the vice versa.

The mechanism through which oceanic fronts are able to induce zones of nearsurface wind-stress curl and divergence has been referred to as the vertical mixing mechanism (Wallace et al., 1989). The SST differences across the oceanic fronts induce distinct thermodynamic conditions in the near-surface atmosphere. Specifically, the near-surface atmosphere is less (more) stable over the warm (cold) side of the SST front and then the vertical mixing in the MABL is intensified (reduced). Consistently, the vertical gradient in the profiles of air temperature, humidity and velocity in unstable (stable) conditions is lower (greater) than the typical neutral logarithmic profile due to the turbulent fluxes (Stull, 1998). One effect of the vertical mixing within the boundary layer is to transfer horizontal momentum from the upper layers to the near-surface layers during the unstable conditions. Thus, because of the differential vertical mixing across the oceanic fronts, the near-surface winds over the warm side of an oceanic front tend to be stronger than the ones over its cold side (Wallace et al., 1989), creating areas of near-surface wind divergence and curl (Chelton et al., 2004). The wind divergence is linearly dependent on the component of the SST gradient parallel to the wind (downwind component), whereas the



Wind Stress Divergence

Figure 2.8: 4-year average of the near-surface wind-stress divergence (top) and curl (bottom), computed from 25-km-resolution wind measurements by the QuikSCAT scatterometer. The figure is reproduced from Chelton et al. (2004)

wind-stress curl is linearly dependent on the component of the SST gradient perpendicular to the wind (crosswind component; Chelton et al., 2001; O'Neill et al., 2003; Maloney and Chelton, 2006). The more intense the downwind and crosswind SST gradient, the more intense the wind-stress divergence/convergence and curl. Refer
to Figure 2.9 for a schematic of the impact of an oceanic front on the near-surface wind-stress due to the vertical mixing mechanism and how the differential vertical mixing across the front generates wind divergence and curl. The resulting areas of near-surface divergence and convergence due to vertical mixing induce anomalies in the vertical motion that can impact the atmospheric dynamics and thermodynamics also above the boundary layer (Xie, 2004; Small et al., 2008).



COOL

Figure 2.9: Schematic of the impact of an oceanic front on the near-surface windstress due to the vertical mixing mechanism and how the near-surface wind-stress variations across the front generate near-surface wind-stress divergence and curl. The schematic is reproduced from Maloney and Chelton (2006).

Atmospheric pressure adjustment mechanism

The GSF has been shown to be associated with time-mean upward motion along its own path extending all the way up to the upper-troposphere (Minobe et al., 2008a). The upward motion in the atmospheric boundary layer is induced by the atmospheric pressure adjustment to SST differences across the oceanic front. As shown in Figure 2.10, the GSF system is characterized by a low and high pressure system on the warm and cold side of the oceanic front, respectively. Thus, a crossfront secondary circulation is established (Small et al., 2008), anchoring a zone of convection on the southern side of the GSF because of wind-convergence over this side of the front. The extension of the upward motion in the free atmosphere is induced by the atmospheric instability due to the latent heat release during the ascent motion. Consistently, a narrow band of precipitation is found along the GSF (Pan et al., 2002). Being the GS region an area of intense cyclogenesis, the time-mean ascent motion should be understood as the cumulative effect of frontal systems passing over the GS region (Parfitt and Czaja, 2016; Parfitt and Seo, 2018) and not as the average daily circulation in that region. In the GSF system, highcloud formation can occur, with highest frequencies in the winter season (Minobe et al., 2008a, 2010). The upper-level divergence induced by the deep convection over the GS and the latent heat release in the troposphere represents ways through which the GS can affect the tropospheric circulation not only locally but also in remote areas by forcing stationary planetary waves (Hoskins and Karoly, 1981; Hoskins and Valdes, 1990; Held et al., 2002).

Before closing this section, we want to specify that the different positions of wind-stress divergence respect an oceanic front shown in Figure 2.9 and Figure 2.10 are not in contrast each other. Specifically, the area of wind-stress divergence occurs over the SST front in Figure 2.9, whereas it occurs north of the GSF in Figure 2.10. In this context, it should be taken into account that Figure 2.9 shows the impact of a time-varying oceanic front on the near-surface wind-stress, whereas Figure 2.10 shows the climatic responses to the GSF. The absence of wind-stress divergence over the climatological GSF position in Figure 2.10, as expected looking at Figure 2.9, should be understood as the effect of time averaging over an area of strong eddy activity.



Figure 2.10: Schematic of the climatic responses to the GSF. The green arrow indicates the warm core of the GS. The black dashed line indicates the GSF. The grey arrows indicate the cross-front near-surface winds. The blue region north of the GSF identifies the area of near-surface wind divergence and high pressure. The orange region south of the GSF identifies the region of near-surface wind convergence, low pressure and enhanced rain. The vertical yellow arrow represents the intense upward motion anchored to the GSF. The blue oval aloft identifies the area of upper-level divergence. The schematic is reproduced from Minobe et al. (2008b).

Oceanic baroclinic adjustment mechanism

The presence of the GSF has been shown to be important also in shaping the North Atlantic stormtrack and eddy-driven jet (Hoskins and Valdes, 1990; Nakamura et al., 2008; Brayshaw et al., 2011). Indeed, the GSF intensifies the near-surface atmospheric baroclinicity in the western portion of the North Atlantic, anchoring a zone of intense baroclinic eddy activity (Sampe et al., 2010; Brayshaw et al., 2011; Novak

et al., 2015) (Figure 2.11). Because of the latter, the southwest-northeast tilt of the stormtrack over the North Atlantic induced by the presence of the Rocky Mountains and the North American continent is further increased (Brayshaw et al., 2009). Consistently, the jet stream is tilted thanks to the eddy-mean flow interaction (Hoskins and Valdes, 1990; Brayshaw et al., 2011). The near-surface baroclinicity necessary for baroclinic instability growth is maintained by the GSF through the differential diabatic heating supplied across the front against the erosive effect due to the baroclinic eddies (Nakamura et al., 2004; Sampe et al., 2010). Hotta and Nakamura (2011) have shown that the sensible heating plays a more important role than the other diabatic heating sources in maintaining the near-surface baroclinicity.

The mechanism through which the sensible heat fluxes maintain the near-surface baroclinicity can be described as follows. The sensible heat fluxes establish an intense surface atmospheric temperature gradient (i.e. baroclinicity) across the oceanic front. As expected, the sharp gradient leads to the development of baroclinic eddies which tend to erode the atmospheric temperature gradient through the associated poleward eddy heat transport. This increases the differences between the SST and the temperature of the overlying air, leading to enhanced upward (downward) sensible heat fluxes south (north) of the GSF. In turn, this leads to a stronger across-front gradient of SHF, the near-surface baroclinicity is restored and the transient eddies can develop again. This mechanism has been referred to as "oceanic baroclinic adjustment" by Nakamura et al. (2008) and it differs from the "baroclinic adjustment" proposed by Stone (1978), in which it is the differential radiative heating between high and low latitudes to restore the near-surface baroclinicity. The restoration of the near-surface baroclinicity by an oceanic front through the sensible heat fluxes is schematized in Figure 2.12. More in detail, Ambaum and Novak (2014) and Novak et al. (2015) have shown that the relationship between the near-surface baroclinicity (also induced by the oceanic forcing) and the poleward eddy heat transport can be described by a nonlinear oscillator model.

Figure 2.11: Baroclinicity (0.5 and 0.6 day⁻¹; solid contours) at 775 hPa and meridional eddy heat flux (10 and 20 K m s⁻¹; dashed contours) between 700 and 925 hPa averaged over the 1957–2002 DJF winters. The figure is reproduced from Ambaum and Novak (2014). The black sector over the GS represents the area of interest for the analysis developed in that study.

Other aspects

Furthermore, the GSF has been shown to affect the time-mean stratospheric circulation. Results from Omrani et al. (2019) indicate that the presence of the GSF warms and weakens the stratospheric polar vortex. This is possible because the GSF enhances the upward wave propagation, leading to wave-activity convergence in the low stratosphere and then to a deceleration of the westerlies (Andrews et al., 1987). Recent studies have also suggested a possible effect of the GSF on the troposphere– stratosphere coupling. In particular, in aqua-planet experiments with an atmospheric GCM (AGCM), Ogawa et al. (2015) have shown that the stratospheric

Figure 2.12: Schematic diagrams showing the restoration of near-surface baroclinicity by an oceanic front through SHF from the ocean against the relaxation by poleward eddy heat transport, reproduced from Sampe et al. (2010).

changes associated with the ozone depletion in the southern hemisphere can propagate into the troposphere only if the SST front in the South Indian ocean is included. The signal can propagate downward because of the excitation of synoptic and planetary waves by the oceanic front. This means that a realistic representation of the eddy-driven tropospheric jet, only possible in the presence of the oceanic fronts, is important for the troposphere–stratosphere coupling. Consistently with Ogawa et al. (2015), Lubis et al. (2018) have demonstrated that the stratospheric changes associated with the polar vortex weakening requires the internal tropospheric activity wave forcing associated with eddy-driven jet dynamics to propagate into the troposphere. Since the GSF directly impacts the tropospheric eddy-driven jet and its dynamics, it is expected that the GSF can impact the troposphere–stratosphere coupling.

The influence of the GSF on the North Atlantic time-mean atmospheric circulation has been mainly analysed using atmosphere-only model simulations forced with realistic and smoothed SSTs in the GS region (e.g. O'Reilly et al., 2016, 2017; Omrani et al., 2019). The results show that the presence of a realistic GSF generates stronger baroclinicity over the North Atlantic than simulations with a smoothed or absent front. Consistently, the meridional eddy heat flux is stronger and the eddy-driven jet moves more polewards. Furthermore, the trimodal character of the observed eddy-driven jet latitude distribution is better represented when the GSF is not smoothed, increasing the occurrence of the northern jet position. This has been shown to also improve the representation of blocking frequency and position over Europe (O'Reilly et al., 2016). Looking at single atmospheric instabilities, Parfitt et al. (2016) have shown not only that the presence of the GSF induces higher occurrence of the atmospheric fronts (i.e. transient eddies) in the western North Atlantic, but also that these changes are mainly due to an higher occurrence of the cold atmospheric fronts. In a similar analysis, Sheldon et al. (2017) have shown that the upward motion in the warm conveyor belt of a cyclone is stronger and reaches a higher level of the troposphere when the GSF is realistically represented in GCMs. Consistently, Kuwano-Yoshida et al. (2010) have shown that the wintermean upward motion above the GSF is much weaker in an atmosphere-only model forced by smoothed SST front compared to the same model forced by realistic front condition. Taking into account the role of the atmospheric synoptic eddies on the North Atlantic climate, the latter studies provide further evidence of the GSF role in determining the stormtrack and eddy-driven jet characteristics.

2.4.2 Impact on the atmospheric variability

As a result of intrinsic oceanic variability as well as atmospheric forcing, the GSF undergoes meridional shifts and meandering, with distinct impacts on the atmospheric variability (Lee and Cornillon, 1995; Ducet et al., 2000; Kwon et al., 2010; Quattrocchi et al., 2012; Kwon and Joyce, 2013). Specifically, the SST anomalies corresponding to the GSF shifts are able to meridionally displace the area of intense synoptic variability in the western North Atlantic (Joyce et al., 2009). Nakamura and Yamane (2009) have shown that small-scale SST anomalies along the GSF may also affect the atmospheric circulation in larger portions of the North Atlantic. Indeed, positive (negative) SST anomalies along the GSF enhance (reduce) the local and downstream low-level baroclinicity. Then, the baroclinic activity along the North Atlantic stormtrack increases and the eddy-driven jet stream shifts poleward. Consistently, Joyce et al. (2009) and Joyce et al. (2019), using observational data, have presented evidence for significant meridional shifts in the North Atlantic eddy-driven jet and stormtrack as a homo-directional response to the GSF shifts, including dynamically consistent changes in the distribution of blocking frequency. Again, Sato et al. (2014) have shown that poleward GSF shifts are associated with abnormal seaice retreat over the Barent Sea via anomalous southerly warm advection and cold anomalies over the Eurasian continent via planetary waves triggered over the GS. In the context of the GSF influence on remote areas, we should also mention the work by Honda et al. (2001). Their results suggest that the low-boundary conditions over the GS region can affect the propagation of wave structures coming from the North Pacific basin and induced by oceanic forcing in that region. This means that the GSF variability can affect the atmospheric circulation in downstream areas not only by triggering wave-train as suggested by Sato et al. (2014), but also by modulating the North Pacific–North Atlantic interaction.

Differently from the atmospheric response to the GSF shifts, there are no studies directly dealing with the atmospheric response to meandering of the GSF. However, some authors have analyzed the atmospheric response to the meandering of the Kuroshio Extension SST front. For example, O'Reilly and Czaja (2015) have observed that periods of strong and elongated SST fronts are associated with an intensified low-level baroclinicity and eddy heat transport in the western North Pacific. These changes in the western oceanic basin force anomalous barotropic flow in the eastern North Pacific via eddy-mean flow interaction, with greater occurrence of blocked days. Ma et al. (2017) have analyzed the impact of SST front meandering on the atmospheric circulation forcing a Weather Research and Forecasting (WRF) model with solved and smoothed oceanic eddies along the Kuroshio Extension region. Their results show that the realistic representation of mesoscale SST anomalies leads to the local cyclogenesis enhancement and the northward shift of the stormtrack and jet stream in the eastern North Pacific. Even if the GS and the Kuroshio WBCs are very different systems, these results shed light on the possible impact of the GSF meandering on the atmospheric circulation.

Chapter 3

The atmospheric response to meridional shifts of the Gulf Stream SST front and its dependence on model resolution

3.1 Introduction

Recent observational and modelling studies have shown that the GSF variability affects both the local and large-scale atmospheric circulation (refer to section 2.4), acting as a source of atmospheric predictability especially on interannual and decadal time scales (Joyce et al., 2019; Athanasiadis et al., 2020).

Despite these advancements, the real character of the atmospheric response to oceanic front variability is still not well established. Some observational studies have shown that the GSF shifts increase the baroclinic activity in North Atlantic, inducing homo-directional shifts of the eddy-driven jet and the stormtrack and dynamically consistent changes in the distribution of blocking frequency (Nakamura and Yamane, 2009; Joyce et al., 2009; Sato et al., 2014; Kwon and Joyce, 2013; Joyce et al., 2019). However, such anomalous atmospheric circulation has been elusive to detect in other observational studies (Joyce et al., 2000; Frankignoul et al., 2001). For example, Wills et al. (2016) have shown that positive SST anomalies over the GS area determine negative NAO-like circulation anomalies, i.e. an opposite response compared to the one described above. Following the same methodological approach of this study, Yook et al. (2022) have extracted similar results for SST variability over the Kuroshio current.

Using atmosphere-only simulations could partly help to disentangle this issue as they allow to isolate the impact of the oceanic variability on the atmosphere. However, even if state-of-the-art AGCMs with horizontal resolutions close the oceanic deformation radius ($\sim 50 \,\mathrm{km}$) are more reliable than previous models and better represent the observed climate, important inter-model discrepancies exist (Czaja et al., 2019). Smirnov et al. (2015) have shown that, in an AGCM with horizontal resolution of 1°, the atmospheric response to shifts in the Oyashio Extension SST front results to be weak and exhibits features generally consistent with a steady linear response to a near-surface heat source (Hoskins and Karoly, 1981). On the other hand, in the AGCM with horizontal resolution of 0.25° the near-surface circulation is substantially weaker, while the vertical motion is stronger and deeper (affecting the upper-troposphere) and the surface meridional eddy heat transport largely balances the SST-induced diabatic heating anomalies. The differences in vertical motion between these two AGCMs recall results from Feliks et al. (2004). Using a very idealized framework, they showed that a narrow oceanic front is able to force the atmospheric circulation above the MABL through thermal pumping of vertical velocity. However, since oceanic fronts have a width of 100 km or less, only sufficiently high-resolution models can represent this mechanism. In line with this, Nakamura and Yamane (2009) argued that only models with a grid spacing no larger than 50 km are able to adequately resolve slight meridional shift of the GS and its impact on the large-scale atmospheric circulation. Ma et al. (2017) showed that a WRF model with a 27-km resolution is much more sensitive to mesoscale SST anomalies along the Kuroshio extension compared to the model with a 162-km resolution. The high-resolution model version exhibits a southward shift of Eastern North Pacific stormtrack and jet stream when the mesoscale SST anomalies are smoothed; in contrast, the low-resolution model is not able to resolve the small-scale diabatic processes associated with the mesoscale SST forcing and therefore shows a weak response in eddy activity and large-scale circulation for the same smoothing in oceanic boundary conditions. Looking at single atmospheric instabilities, Willison et al. (2013) have shown that a WRF model with a 20-km horizontal resolution exhibits enhanced frontal dynamics in North Atlantic mid-latitude cyclones compared to the same WRF model with a 120-km resolution. The authors have highlighted a more intense positive feedback between cyclone intensity and latent heat release in the high-resolution model, resulting in the intensification of the stormtrack as well as in the strengthening of the jet stream. Following these results, Sheldon et al. (2017) have argued that the diabatic heating in the warm conveyor belt of cyclones travelling close to the GSF is directly proportional to the number of air-parcel trajectories feeding the upward motion. The higher the model resolution, the higher the number of air-parcel trajectories reaching the upper-troposphere.

Despite these efforts, the number of studies dealing with the impact of horizontal resolution on the atmospheric response to oceanic forcing is still limited. In addition, previous analyses are based on idealised experimental frameworks forcing the atmosphere with fixed and unrealistic SST anomalies. Finally, such past studies have been limited to single-model assessments. A multi-model analysis to systematically investigate differences between the low-resolution and high-resolution atmospheric response to realistic SST variability linked to the meridional shifts of the GSF is still lacking.

The objective of this part of the PhD thesis is twofold: on the one hand it aims at assessing the dependence of the atmospheric response to GSF shifts on model resolution, in particular the atmospheric horizontal resolution, and on the other it aims at providing further evidence on the character of the large-scale atmospheric response by assessing the latter in isolation from the coupled variability in which is embedded, using atmosphere-only simulations. The atmospheric response has been investigated in the context of the High Resolution Model Intercomparison Project (HighResMIP), by analysing historical simulations performed with three AGCMs, each run at two different horizontal resolutions. The AGCMs have been forced with the same observed SSTs. Understanding the impact of horizontal resolution on air–sea interaction can shed light on the influence of extratropical oceanic variability on the atmosphere, with important implications for climate predictions and climate change studies.

The chapter 3 of this PhD thesis is structured as follows. In section 3.2 the HighresMIP dataset, the SST anomalies associated with the GSF shifts and the methodological approach are described. In section 3.3 the atmospheric response to the GSF shifts is presented. In section 3.4 we present a heat budget analysis along, across and above the GSF as well as in the North Atlantic basin, in order to investigate ways through which diabatic heating is balanced locally and at a wider scale. In section 3.5 we discuss some large-scale features of the atmospheric response to the GSF shifts. Finally, in section 3.6 we summarize the results, highlighting the most salient outcomes of this part of the PhD thesis.

3.2 Data and methodological approach

3.2.1 Data

The HighResMIP is part of the wider Coupled Model Intercomparison Project 6, and it was designed with the specific objective of investigating the impact of increasing model horizontal resolution on the representation of the observed climate and of an array of important physical processes (Haarsma et al., 2016). As such, it provides an ideal framework for a multi-model analysis of the impact of increasing model resolution on the atmospheric response to oceanic forcing. In this study, an ensemble of six atmosphere-only historical simulations have been analysed. Three different models have been used, each run with two configurations differing only in their horizontal resolution. Hereafter we will refer to model configurations with a

Table 3.1: HighResMIP models. Columns detail the institution name, the model name, the nominal resolution, the number of vertical levels, the number of members used for analysis and the model reference.

Institution	Model	Nominal Resolution (km)	Vertical Levels	Members	Reference
EC-Earth-Consortium	EC-Earth3P EC-Earth3P-HR	100 50	91 91	3 3	Haarsma et al. (2020)
Met Office Hadley Centre (MOHC)	HadGEM3-GC31-MM HadGEM3-GC31-HM	100 50	85 85	3 3	Roberts et al. (2018)
ECMWF	ECMWF-IFS-LR ECMWF-IFS-HR	50 25	91 91	8 6	Roberts et al. (2019)

nominal resolution coarser than 50 km as R100 models, and those with a nominal resolution finer than or equal to $50 \,\mathrm{km}$ as $R50 + \mathrm{models}$. Each model has been forced with the HadISST2 sea ice concentration and SST dataset, provided at daily frequency in the period 1950-2014 on a 0.25° grid (Kennedy et al., 2017). For each model a multi-member ensemble of simulations has been used, and the results in the following sections refer to the respective ensemble means. This specific experimental design and the use of multi-member ensembles allows a more robust identification of the atmospheric response forced by the observed oceanic variability as the ensemble averaging aids the forced response to emerge from the chaotic atmospheric variability, which is particularly strong at midlatitudes. The models considered in this study are documented as follows: EC-Earth3P (Haarsma et al., 2020), ECMWF-IFS (Roberts et al., 2018), HadGEM3-GC31 (Roberts et al., 2019). Table 3.1 shows the respective model configurations, nominal resolutions (km), number of vertical levels, number of members and reference (model documentation). Additional details about the experimental set-up can be found in Haarsma et al. (2016, see "Tier 1 - highresSST-present" experiment). Finally, model results have been compared to ERA5 reanalysis (Hersbach et al., 2020), here used as a surrogate of observations in the period 1950–2014.

Figure 3.1: SST (K; color shaded) anomalies associated to "North" (a) and "South" (b) phases of the GSF in winter (DJF) in HadISST2 dataset. The "North" ("South") phase has been defined as the upper (lower) tercile of GSF mean latitude. The respective climatological position of the SST front is indicated by the cyan dashed line. The black contours indicate winter SST climatology. The black rectangular frames indicate the longitudinal range over which the GSF mean latitude and subsequent diagnostics (Figure 3.9, Figure 3.10, Figure 3.11) have been calculated. Black dots denote anomalies that were found to be statistically significant at the 90% confidence level (details in section 3.2).

3.2.2 Gulf Stream sea surface temperature front shift

In this study, the atmospheric response to interannual GSF meridional shifts in the winter season (December–February) has been investigated. Previous studies have shown that the north–south shift of GSF represents the leading variability mode of SST variability in the GSF area on interannual and longer timescales (Joyce et al., 2009; Kwon and Joyce, 2013). The winter season has been selected because this is the

time of the year characterised by the most intense heat exchanges between ocean and atmosphere, resulting in a stronger impact of the ocean variability on the atmosphere (Kallberg et al., 2005). The GSF has been defined as the line of maximum SST gradient. The SST gradient magnitude has been calculated for winter mean SST fields, smoothed with a 2D spatial Gaussian–Kernel filter applied to a 7x7 gridpoint box, with standard deviation equal to 2. The smoothing has been applied in order to remove isolated points of strong SST gradient not representative of the GSF position. The latitude of the GSF has been averaged in the 50°–68°W longitudinal range, where the GS is more zonally oriented. Then, this zonally averaged latitude of the GSF has been used to define the "North" and "South" phases of the front shift via the respective upper and lower tercile categories.

Figure 3.2: Winter-mean latitude of the GSF averaged in the range $50^{\circ}-68^{\circ}W$ (solid black line). Red, yellow and blue stars represent years in which the GSF latitude falls, respectively, in the upper, middle and lower tercile categories (i.e. North, Middle and South position). The dashed black line is the 10-year running mean applied to GSF latitude timeseries. The solid and dashed black lines also represent the normalized interannual and smoothed GSF indices for the comparison with the indices suggested in other studies. The blue line represents the winter-mean normalized GSNW index as defined by Taylor and Stephens (1980). The orange line represents the normalized GSI as defined by Joyce et al. (2019). The GSI is averaged over January and March as representative of the winter season. Refer to the y-axis on the right for the normalized indices.

Figure 3.1 shows the SST composites obtained by averaging all years over the upper and below the lower terciles, i.e. the "North" and "South" phases of the

GSF, consisting of 22 and 21 years, respectively. The resulting SST pattern shows a tripolar structure extending to the entire North Atlantic in both GSF phases but of opposite sign (Figure 3.1). The "North" ("South") phase is associated dominantly with positive (negative) anomalies along the climatological position of the GSF in winter, as well as in the midlatitude North Atlantic to the south of the GSF and the NAC, and negative (positive) anomalies further north in the subpolar gyre region and south of 30°N. The anomalies are strongest close to the GSF with values exceeding 1 K (absolute departures). In the remaining part of the basin, the SST anomalies are much weaker and lower than 0.4 K. The corresponding SST anomalies in other parts of the global ocean have been assessed, and their significance in remotely forcing an atmospheric circulation response over the extratropical North Atlantic is discussed in the following.

In Figure 3.2 the GSF latitude time series (black line) in the period 1950–2014 is presented. The years corresponding to the "North" ("South") phase are highlighted with red (blue) stars. The index features an interannual variability component associated with SST front latitudinal shift of about 0.2°-0.5°. A lower frequency, decadal-scale component is also evident associated with weaker amplitude meridional displacements. The North Atlantic SST variability described above has been previously interpreted as the oceanic response to NAO forcing via SHF and Ekman currents, explaining an important portion of the extratropical SST variability on seasonal and interannual timescale (Cayan, 1992; Deser et al., 2010). However, it has been shown that close to the GSF the SST variability is primarily driven by oceanic processes, such as oceanic heat transport and diffusion, with the former admittedly remaining subject to the atmospheric forcing (Kelly and Qiu, 1995; Dong and Kelly, 2004; Kelly et al., 2010; Kwon et al., 2010; Minobe et al., 2010; Patrizio and Thompson, 2021). Therefore, the SST anomalies described here are the surface oceanic signature of both historical atmospheric forcing and intrinsic oceanic variability. This aspect must be taken into account to avoid erroneous considerations during the analysis of data, especially observations. A number of studies (e.g.

Czaja and Frankignoul, 2002; Ciasto and Thompson, 2004; Wills et al., 2016) present evidence that SST anomalies in the vicinity of the GS may affect the large-scale atmospheric circulation up to several months ahead. Some authors have argued that meridional shifts of the GSF may be even key in explaining decadal NAO variability and predictability (Feliks et al., 2011; Joyce et al., 2019; Athanasiadis et al., 2020).

Before closing this section, it is specified that other approaches to define the latitudinal position of the GS and its front have been proposed in literature. Amongst them, the GS North Wall (GSNW) index defined by Taylor and Stephens (1980) and the GS Index (GSI) defined by Joyce et al. (2000) have been widely used. The former defines the GS position through the principal component analysis of the surface temperature front position in the 79°-65°W longitudinal range. The latter defines the GS position through the principal component analysis of the 15°C isotherm position at 200 m depth in 9 fixed locations along the GS path and between 75°W and 55°W. Due to their widespread use, the GSF index is here compared with the GSNW index and the GSI. To do so, we have used the monthly GSNW data provided in McCarthy et al. (2018), extracting the index for the winter season (DJF). Furthermore, we have used the GSI data provided in Joyce et al. (2019). The approach to define the GS position adopted by Joyce et al. (2019) is slightly different compare to the one defined in Joyce et al. (2000). However, the two indices are very comparable being both based on the principal component analysis of the $15^{\circ}C$ isotherm position at 200 m depth in 9 fixed positions along the GS. Being the GSI data in Joyce et al. (2019) provided with a 3-month time step, we have adopted the GSI averaged over January and March as representative of the winter season. As shown in Figure 3.2, the GSF index (black line) captures the general trend of both GSNW index (orange line) and GSI (blue line), especially from 1975 onward. This is not always the case on the interannual timescale, where the indices can also be in anti-phase. Consistently, the zero-lag cross-correlation between the GSF index and the GSI (GSNW index) is not particularly high, with a value of 0.52 (0.39) (significance greater than 99%; the statistical significance of the correlations has been

assessed through bootstrap test in Ebisuzaki, 1997). The different trends of the indices on short (interannual) timescale may be linked to differences in the adopted longitudinal ranges, which capture sectors of the GS with distinct behaviours and characteristics. Gangopadhyay et al. (2016) have shown that the signature of the interannual variability in the GS path changes along the stream's path from 75°W to 55°W. Furthermore, it should be taken into account that the GSI is based on subsurface temperature data, which are largely unaffected by surface diabatic processes being the wintertime mixed layer west of 60°W less than 200 m (Qiu and Huang, 1995). This is not the case for the GSNW and the GSF indices which are based on surface temperature data. The added value of the GSF index here defined stays on the fact that it specifically captures the surface oceanic variability and that it does it in the GS region with the most intense oceanic-eddy activity (east of 70° W; Ducet et al., 2000). Both aspects are really important if we take into account that the air-sea interaction associated with small- and meso-scale oceanic features (such as oceanic fronts and eddies) has been shown to strongly influence the atmospheric circulation on the large spatial scale (Ma et al., 2017).

3.2.3 Methods

The atmospheric circulation anomalies associated with the GSF shifts have been analysed using composites of "North" minus "South" phase of the GSF. For assessing the statistical significance of the differences between the two GSF phases, a two-sided Student's t-test against the null hypothesis of no-difference has been applied at the 90% significance level. Specifically, the impact of the GSF meridional displacement on the atmospheric circulation has been characterised through the analysis of winter-mean near-surface winds, SLP, SHF (i.e. the sum of turbulent sensible and latent heat fluxes), zonal winds at 850 hPa (representing the eddy-driven jet, U850) and meridional temperature gradient (related to baroclinicity) at 925 hPa and, starting from daily data, [2–6]-day high-pass meridional eddy heat flux (MEHF) at 850 hPa, variance of [2–6]-day high-pass eddy meridional wind at 250 hPa and blocking frequency. The eddy-driven jet response has been analysed both in terms of the associated anomalous field and jet-latitude variability. As in Woollings et al. (2010), the daily zonal winds at 850 hPa have been zonally averaged in the 0°–60°W longitudinal range, masking out Greenland and the Atlas mountains that intersect the 850 hPa isobaric surface. Then a 10-day low-pass Lanczos filter with a window of 61 days (Duchon, 1979) has been applied to the resulting fields. The jet stream latitude has been defined as the latitude of maximum westerly wind speed in the 15°–75°N latitudinal range. To detect atmospheric blocking, the 2D large-scale blocking index defined in Davini et al. (2012) has been adopted including the condition to avoid the detection of false blocking events at low-latitudes (Athanasiadis et al., 2014). In addition to the above mentioned diagnostics, the monthly zonal-mean circulation and the atmospheric heat budget in the verticalmeridional cross-section in the 50°–68°W longitude range have been computed.

3.2.4 SST front-following coordinate system

For analyses in the vertical-meridional cross-section an SST front-following coordinate system has been devised. In order to define the SST-front coordinate system, the position of the GSF computed on the original HadISST2 0.25° grid has been assigned to the nearest grid point of the native grid of each AGCM. Since the oceanic grid is finer than those of AGCMs, the GSF position in each model is slightly different from the original one. However, this choice has no significant effect on atmospheric diagnostics calculated along the vertical-meridional plane because the AGCMs would not have been affected by subgrid SST front features anyway. Starting from the interpolated GSF position, a 3D space has been considered with the abscissa representing the displacement along the GSF, the second horizontal axis representing the meridional direction and the vertical axis representing pressure levels. Then, the vertical-meridional plane has been derived averaging the diagnostics in the along-front direction. The zonal average has been taken avoiding grid points over land north of the GSF. The SST front-following coordinate system has allowed for taking into account the poleward tilt of the SST front and carrying out the analysis purely along the cross-front direction. Such a system represents a novelty in the scientific literature dealing with the oceanic fronts variability and their impact on the atmospheric circulation.

3.3 Atmospheric response to Gulf Stream SST front shift

Figure 3.3 shows near-surface wind, SLP and SHF composite differences on the GSF meridional displacement in the models and the observations. R100 models exhibit a low-pressure anomaly downstream of the GSF that is largely consistent with meridional temperature advection in the area along the GSF tending to balance the anomalous diabatic heating. The general features of the atmospheric response in R100 models recall what one would expect at the surface for an extratropical shallow heat source in theoretical linear models (Hoskins and Karoly, 1981). On the other hand, R50+ models show a high-pressure anomaly downstream of SST anomalies, with winds blowing from the southern portion of North Atlantic poleward in the vicinity of the SST front. This is in contrast to what is expected from the lineartheory. The R50+ response resembles results from other studies, showing that the atmospheric response is strongly mediated by transient eddy fluxes balancing the anomalous diabatic heating (Peng et al., 1997; Peng and Whitaker, 1999; Watanabe and Kimoto, 2000). It is specified that in R50+ models the SLP anomalies are statistically significant over a large part of the North Atlantic at the 90% confidence level. This is not the case for R100 models, for which the statistically significant anomalies over the same region are more limited.

SHF anomalies are particularly intense close to the GSF climatological position both in R100 and R50+ models, reaching values that correspond to an important portion of winter climatology (about 15–20%). These largely coincide with the area of strongest along-front SST anomalies. For R100 models meridional temperature

Figure 3.3: SHF (W m⁻²; color shading), SLP (Pa; contours, solid for positive) and near-surface wind (m s⁻¹; vectors) response to the GSF shifts in winter (DJF). (a) ERA5. (b, c) EC-Earth. (d, e) MOHC. (f, g) ECMWF. Beside institution name, the model nominal resolution in km is reported. SHF are considered to be positive upwards, namely from the ocean to the atmosphere. For models the magenta contours represent SLP anomalies at -120, -60, -30, 30, 60 and 120 Pa; for ERA5 the magenta contours represent SLP anomalies at -180, -150, 150 and 180 Pa. Thick vectors indicate wind anomalies that were found to be significant at the 90% confidence level (details in section 3.2). The winter climatological position of the GSF is indicated by the cyan dashed line. The arrow in the upper-right corner of each panel is the unit vector for surface wind (m s⁻¹).

advection tends to balance this anomalous diabatic heating associated with the GSF shifts (Figure 3.12d). Indeed, the anomalous atmospheric circulation transports cold air from higher latitudes toward the GSF area. In contrast, in R50+ models meridional temperature advection does not tend to balance the anomalous diabatic heating but to exacerbate the induced temperature tendencies (Figure 3.13d). This

is consistent with the northward advection of warm air from lower latitudes. This fact indicates that other processes, different from horizontal temperature advection, have a key role in balancing heating anomalies above the GSF. This aspect will be discussed in more detail in section 3.4. Unlike in the frontal area, SHF anomalies differ in other portions of the ocean basin. R100 models show negative SHF anomalies south of Greenland and on the southern flank of the GSF and positive ones in the central North Atlantic. In contrast, R50+ models develop positive flux anomalies south of Greenland and on the eastern North Atlantic, with negative values mainly confined at the southern flank of the GSF. For the ECMWF model with a nominal resolution of 50 km (Figure 3.3f) and the Met Office Hadley Centre (MOHC) R50+ model (Figure 3.3e) the negative SHF anomalies extend more northward compared to the other two R50+ models, reaching the southern Greenland coast. These discrepancies are consistent with near-surface wind differences among R50+ models. In Figure 3.3c and Figure 3.3g, positive SHF anomalies close to the Greenland coast are associated with surface wind blowing from inland of North America and then transporting cold and dry continental air that is warmed by the ocean. In Figure 3.3e and Figure 3.3f, negative SHF anomalies close to Greenland are associated with near-surface wind blowing from the southern North Atlantic and transporting warm and wet air that reduces the thermal air-sea contrast. Apart from these discrepancies, R50+ models are more comparable to each other than R100 ones. Furthermore, R50+ models reproduce general features of surface atmospheric anomalies found in the ERA5 dataset. As depicted in Figure 3.3a, observations show a zonally elongated anticyclonic circulation anomaly in surface winds, consistent with the positive SLP anomaly downstream the heating source (statistically significant at 90% confidence level). SHF anomalies are positive far north of the GSF and negative far south as the respective wind anomalies suggest in an area of climatological westerly surface flow. Consequently, the SHF anomalies are spatially anti-correlated with the SST anomalies as expected for atmospheric forcing to the ocean. This implies that the large-scale SST anomalies seen in Figure 3.1 away from the GSF are the fingerprint of local atmospheric forcing that occurred in the real system during and prior to the GSF shifts.

The SHF, SLP and SST patterns in observations recall positive NAO-like forcing on the ocean. This is in line with previous studies showing that the GS meridional shifts are correlated with the low frequency NAO variability at a positive lag of about 1–2 years with the ocean following the atmospheric forcing (Taylor and Stephens, 1998; Frankignoul et al., 2001; Sanchez-Franks et al., 2016). However, the comparison between AGCMs and observations suggests that, as soon as the GSF shift is established, the ocean provides a positive feedback on the atmospheric circulation, and that the realistic representation of this feedback in AGCMs requires a sufficiently high horizontal resolution (as R100 models fail to reproduce it).

In Figure 3.4 zonal wind anomalies at 850 hPa are presented. R100 models generate a southward shift of the eddy-driven jet, with negative anomalies to the north of the climatological jet position and positive anomalies to the south, even though significant differences can also be seen between the two models. In contrast, R50+ models exhibit anomalies of the opposite sign, indicating a northward shift of the jet. These anomalies recall the pattern seen in the observations (ERA5), yet with a lower amplitude. Then, for both the R100 and the R50+ models, similar but stronger zonal wind anomalies were found aloft (not shown) indicating an equivalent barotropic structure extending throughout the troposphere.

To assess in more detail the response of the North Atlantic eddy-driven jet to the GSF shifts, the jet latitude distributions (probability density functions, PDFs) for each phase of the GSF position are shown for each model in Figure 3.5. As specified in section 3.2, it is recalled that the jet stream latitude has been defined through the zonal winds at 850 hPa, zonally averaged in the 0°-60°W longitudinal range and maximized in the 15°-75°N latitudinal range. Even though there are some differences between the two R100 models and between the four R50+ models, the common characteristics in each group are consistent with the results shown for the zonal wind anomalies at 850 hPa, indicating a northward shift in R50+

Figure 3.4: 850 hPa zonal wind (m s⁻¹; color shaded) response to the GSF shifts in winter (DJF). (a) ERA5. (b, c) EC-Earth. (d, e) MOHC. (f, g) ECMWF. Beside institution name, the model nominal resolution in km is reported. Black dots denote anomalies that were found to be statistically significant at the 90% confidence level (details in section 3.2). Green contours indicate the winter climatology of zonal wind at 850 hPa every 2 m s⁻¹ from 4 m s⁻¹. The winter climatological position of the GSF is indicated by the black dashed line. The grey masking corresponds to areas where the orography intersects the climatological 850 hPa isobaric surface.

models and a southward shift in R100 models. Again, only the anomalies in the jet latitude distribution of the R50+ models resemble the respective observed (ERA5) anomalies, being consistently weaker than the latter. Indeed, in ERA5 the GSF displacements are associated with anomalies of opposite sign in the jet latitude distribution north and south of the GSF position (approximately at 42°N) indicating shifts in the eddy-driven jet that are homo-directional to the GSF shifts. It should be noted that none of the six models can reproduce the trimodal character of the observed jet latitude distribution in wintertime, originally shown by Woollings et al. (2010) and seen here for ERA5. Particularly, the southern-jet regime is too weakly represented by the models, a problem that is arguably linked to the absence of coupled feedbacks in atmosphere-only simulations. Models unable to represent the observed circulation regimes (Madonna et al., 2017) are also expected to feature a "distorted" response to the GSF variability as it has been noted that the atmosphere tends to respond to oceanic and other moderate forcings by changes in the frequency of occurrence of its dominant circulation regimes (Palmer, 1993).

The jet latitude variability has been previously interpreted as a response to variations in the stormtrack activity, which influences the jet stream through an upstream baroclinic and a downstream barotropic effect (Novak et al., 2015; O'Reilly et al., 2017). The former is based on the non-linear oscillator relationship between low-level baroclinicity and MEHF (Ambaum and Novak, 2014); the latter is the downstream representation of the upstream low-level baroclinicity changes, which induce variations in eddy anisotropy and wave breaking, with direct impact on jet latitude (Orlanski, 2003). To assess whether the previously discussed homo-directional jet stream response to the GSF shifts is the result of changes in low-level baroclinicity and then in stormtrack activity, in Figure 3.6 and Figure 3.7 we show the meridional temperature gradient (here used as a proxy for baroclinicity) and MEHF anomalies at 925 hPa and 850 hPa respectively. As expected, the strong positive SHF anomaly (corresponding to the "North" minus "South" composites difference) along the GSF induces local changes in low-level baroclinicity both in R100 and R50+ models. Specifically, baroclinicity is found to increase at the northern flank of the GSF and to decrease at the southern flank. In the immediate vicinity of the GSF, these changes are understood as forced by the displacement of the SST front itself and are consistent with the meridional gradient of the above-mentioned SHF anomaly. Beyond these local changes, in R50+ models low-level baroclinicity undergoes significant changes also in a meridionally broader zone to the north of the GSF. As discussed later, these changes may be explained considering the meridional

Figure 3.5: Jet latitude distributions during winter (DJF). As specified in section 3.2 and following the method in Woollings et al. (2010), the jet latitude has been defined through the zonal winds at 850 hPa, zonally averaged in the 0°–60°W longitudinal range and maximized in the 15°–75°N latitudinal range. (a) ERA5. (b, c) EC-Earth. (d, e) MOHC. (f, g) ECMWF. Beside institution name, the model nominal resolution in km is reported. Red (blue) lines represent the distribution during the "North" ("South") phase of the GSF position in the respective model. Green lines represent the difference between the two distributions ("North" minus "South") and correspond to the right y-axis.

gradient in zonal temperature advection (dipole of opposite temperature tendencies) associated with the meridional shift in the low-level jet. In fact, stronger westerlies to north and weaker westerlies to the south, in an area where westerlies in winter are associated with cold temperature advection from the North American continent over the relatively much warmer ocean, tend to increase atmospheric baroclinicity in between. Similar large-scale anomalies in atmospheric baroclinicity have been detected also by Nakamura and Yamane (2009), who have argued that SST anomalies associated with the GS variability, despite being limited to the GS area, may be responsible for such broader-scale anomalies in atmospheric baroclinicity.

Figure 3.6: 925 hPa meridional air temperature gradient (K °lat⁻¹; color shaded) response to the GSF shifts in winter (DJF). The temperature gradient is reversed, i.e. equatorward. (a) ERA5. (b, c) EC-Earth. (d, e) MOHC. (f, g) ECMWF. Beside institution name, the model nominal resolution in km is reported. Black dots denote anomalies that were found to be statistically significant at the 90% confidence level (details in section 3.2). Green contours indicate the winter climatology every 0.8 K °lat⁻¹ from 0.8 K °lat⁻¹. The winter climatological position of the GSF is indicated by the black dashed line. The grey masking corresponds to areas where the orography intersects the climatological 925 hPa isobaric surface.

Given the changes in low-level baroclinicity in the broader area to the north of the GSF (around and to the east of Newfoundland, between 45°–55°N), R50+ models are found to respond with increased MEHF there as expected for baroclinic adjustment (Stone, 1978). This can be seen clearly in Figure 3.7, while it is not the case for R100 models. This discrepancy relates to how R100 models respond (differently, as shown in the next section) to the anomalous diabatic heating along the GSF.

Figure 3.7: MEHF anomalies (v'T'; m s⁻¹ K; color shaded) at 850 hPa induced by the GSF shifts in winter (DJF). (a) ERA5. (b, c) EC-Earth. (d, e) MOHC. (f, g) ECMWF. Beside institution name, the model nominal resolution in km is reported. Black dots denote anomalies that were found to be statistically significant at the 90% confidence level (details in section 3.2). Green contours indicate the winter climatology of eddy heat flux at 850 hPa every 3 m s⁻¹ K from 6 m s⁻¹ K. The winter climatological position of the GSF is indicated by the black dashed line. The grey masking corresponds to areas where the orography intersects the climatological 850 hPa isobaric surface.

In turn, the increased MEHF found for R50+ models in the broader area to the north of the GSF is considered to be part of the causal chain leading to the shift of the eddy-driven jet as it indicates westward acceleration in the lower troposphere following the E-vector formulation of eddy-mean flow interaction (Hoskins et al., 1983). In fact, the stormtrack (diagnosed via the variance of the meridional eddy velocity aloft, v'v' at 250 hPa; Figure 3.8) is found to move with the jet as it exhibits a meridional dipole pattern of anomalies similar to the one seen for the zonal wind in R50+ models. However the horizontal E-vector (v'² - u'²; -u'v') divergence at 250 hPa (not shown) did not reveal a significant forcing on the zonal mean flow.

The detected changes in low-level baroclinicity, MEHF and storm activity for R50+ models are found to be similar and more pronounced in the observations (ERA5, in Figure 3.6 and Figure 3.7), as it was the case for the zonal wind anomalies (Figure 3.4). In this context it must be specified that, while in the atmosphere-only simulations the atmospheric anomalies are only a response to the oceanic forcing, the respective anomalies detected in the observations represent also the atmospheric forcing that causes the GSF shifts in the first place (at negative lags but also at lag-0, i.e. concurrent with and preceding the changes in the GSF position). The study by Frankignoul et al. (2001) is quite insightful in this regard, showing that the GS axis moves to the north (south) following positive (negative) NAO-like forcing.

Finally, it is specified that the low-level baroclinicity anomalies have been computed taking into account also the changes in the static stability component, yet they show patterns quite comparable to what described above both for AGCMs and observations (not shown). This means that the low-level baroclinicity anomalies are mostly due to changes in the meridional temperature gradient rather than changes in static stability.

3.4 Heat budget

In the previous section it has been shown that R100 and R50+ simulations respond differently to the meridional displacements of the GSF, in terms of circulation and

Figure 3.8: Stormtrack anomalies (diagnosed via the variance of the meridional eddy velocity aloft, v'v' at 250 hPa; m² s⁻²; color shaded) induced by the GSF shifts in winter (DJF). (a) ERA5. (b, c) EC-Earth. (d, e) MOHC. (f, g) ECMWF. Beside institution name, the model nominal resolution in km is reported. Black dots denote anomalies that were found to be statistically significant at the 90% confidence level (details in section 3.2). Green contours indicate the winter climatology of stormtrack at 250 hPa every 30 m² s⁻² from 120 m² s⁻². The winter climatological position of the GSF is indicated by the black dashed line.

transient eddy activity. The resolution-dependent response of the atmospheric circulation to similar SHF anomalies close to the GSF raises the question of what might be different between R100 and R50+ models in terms of the primary, local response to the anomalous diabatic heating associated with the GSF shifts. Moreover, it was found that in R100 models cold meridional advection tends to balance the anomalous diabatic heating, whilst this was not found to be the case for R50+ models. Therefore, our study was naturally led to the analysis of the local thermodynamic balance.

Relevantly to this point, Smirnov et al. (2015), analyzing the response of low- and high-resolution atmosphere-only models to prescribed idealized SST forcing along an oceanic front, have found the local heat budget to depend on the horizontal resolution. Thus, to address the above-posed questions, a similar analysis was conducted for each of the examined models. Specifically, the terms of the time-mean thermodynamic equation have been computed as in Smirnov et al. (2015):

$$\overline{\dot{Q}} - \overline{u}\frac{\partial\overline{T}}{\partial x} - \frac{\partial\overline{u'T'}}{\partial x} - \overline{v}\frac{\partial\overline{T}}{\partial y} - \frac{\partial\overline{v'T'}}{\partial y} - (\overline{\omega}\frac{\partial\overline{T}}{\partial p} - \frac{k}{p}\overline{\omega}\overline{T}) - (\overline{u'}\frac{\partial\overline{T'}}{\partial p} - \frac{k}{p}\overline{\omega'T'}) = 0 \quad (3.1)$$
I II III IV V VI

Here u and v represent the zonal and meridional wind component respectively; ω is the pressure tendency, proportional to the vertical wind; p and T represent the pressure and the temperature, while \dot{Q} is the diabatic heating rate. It is specified that \dot{Q} is not provided as model output and then it has been calculated as a residual from the other heat budget terms. The overbars indicate climatological monthly means during winter, while the primes indicate departures thereof, k is R/C_p, with R equal to 287 J kg⁻¹ K⁻¹ and C_p equal to 1004 J kg⁻¹ K⁻¹. Terms I, III and V represent the zonal, meridional and vertical mean thermal advection, respectively. Terms II, IV and VI represent the respective zonal, meridional and vertical eddy heat flux convergences. Terms V and VI also include the mean and eddy component of the adiabatic heating rate, respectively. All terms have been calculated for each winter calendar month and both positions of the GSF (i.e., "North and "South"). Then, monthly differences between the "North" and "South" positions have been derived and averaged to form the respective winter means.

Figure 3.9 shows the cross-front vertical section of the diabatic heating term calculated in the thermodynamic budget (equation 3.1). At the vicinity of the GSF both R100 and R50+ models show strong diabatic heating anomalies in the lower troposphere, below 600–700 hPa, with maximum values near the surface. This is

expected considering the collocated positive SHF anomalies shown in Figure 3.3. It is noted, also, that R100 models exhibit significantly larger diabatic heating near the surface (Figure 3.9b, Figure 3.9c). Furthermore, it should be mentioned that the diabatic heating rate has been calculated as a residual from the thermodynamic equation and thus it includes other heating terms. In general, the diabatic heating in the boundary layer of a limited geographical area (as that straddling the GSF) depends on the underlying SHF (sensible and, if condensation occurs, also latent) distributed in the vertical by the boundary layer parameterization scheme, as well as to the locally absorbed radiation and the net heat fluxes entering through the lateral boundaries of the area. Considering these aspects, the diabatic heating rates shown in Figure 3.9 are not directly comparable with the SHF shown in Figure 3.3. Finally, it should be mentioned that the diabatic heating rate calculated as a residual is affected by any errors in computing all the terms in the thermodynamic budget. In ERA5 the diabatic heating in the lower troposphere is weaker than in the models (Figure 3.9a), something that is understood considering that ERA5 data come from a coupled world/model.

Figure 3.10 shows the heat budget as indicated above, zonally averaged along the SST front and meridionally in the range -1° and $+1^{\circ}$ north of the GSF position (using the previously described SST front-following coordinates). This latitude range has been chosen so as to match the meridional span of the intense SST anomalies induced by the meridional shifts of the GSF (Figure 3.1) and the induced SHF anomalies (Figure 3.3) in its vicinity.

Near the surface the diabatic heating anomalies described above are largely balanced by horizontal heat terms, with a dominance of the meridional components (terms III and IV) over the zonal ones (terms I and II) in all cases. However, while in ERA5 and R50+ models the anomalous diabatic heating at and near the surface is balanced exclusively by the meridional eddy heat flux divergence (term IV) with the meridional mean advection (term III) opposing this action, in R100 models the meridional mean advection co-operates with the meridional eddy heat flux diver-

Figure 3.9: Cross-front vertical section of diabatic heating term (K day⁻¹; color shading) in the thermodynamic budget (equation 3.1), calculated as a residual. (a) ERA5. (b, e) EC-Earth. (c, f) MOHC. (d, g) ECMWF. Beside institution name, the model nominal resolution in km is reported. The vertical dashed line represents the GSF position in relation to which the cross-front section has been constructed.

gence in balancing diabatic heating (Figure 3.10b, Figure 3.10f, Figure 3.10l). The different role played near the surface by the meridional mean advection in R50+ and R100 models is consistent with the near-surface wind anomalies presented in Figure 3.3, while similar differences in mechanisms balancing the heating at the surface emerged also in Smirnov et al. (2015). Vertical terms (terms V and VI) are negligible at the surface as expected for vanishing vertical motions at the lower

Figure 3.10: Vertical profiles of composited differences for zonal (terms I and II), meridional (terms III and IV) and vertical (terms V and VI) terms in the thermodynamic budget (equation 3.1) (K day⁻¹). Last column shows the vertical profile of the diabatic heating term (K day⁻¹), calculated as residual. (a)–(d) R100 models. (e)–(h) R50+ models. (i)–(n) ERA5. Each term has been calculated as a difference between the "North" and "South" phase of the GSF, averaged zonally along the front in the 50°–68°W longitudinal range and meridionally in the range -1° and +1° north of the SST front position (using the SST front-following coordinate system described in section 3.2). Blue lines refer to the EC-Earth model, green lines to the MOHC model, orange lines to the ECMWF model, and black lines refer to ERA5. Orange bold lines in (d)–(f) refer to the ECMWF model with a nominal resolution of 25 km. Solid lines refer to terms involving heat transport by time-mean fields (terms I, III and V), whilst dashed lines refer to the respective eddy components of heat transport (terms II, IV and VI).

boundary (Figure 3.10c, Figure 3.10g, Figure 3.10m). Instead, considering that the vertical profile of diabatic heating exhibits a gradual reduction from the surface up to about 600 hPa (Figure 3.9, Figure 3.10), it is noted that for all models and the observations (ERA5) the anomalous diabatic heating in the lower troposphere is partly balanced by mean vertical motion. In the vicinity of the GSF, which is an area of intense cyclogenesis and low-level convergence associated with the passage of frontal systems (Parfitt and Seo, 2018), the "mean vertical motion" should be understood as the aggregated effect of pulses in vertical motion. Zonal terms (terms I and II) seem to play a less important role, especially in the lower troposphere where the diabatic heating takes place (Figure 3.10a, Figure 3.10e, Figure 3.10i).

In order to further examine the atmospheric response in the vicinity of the GSF,

Figure 3.11: Zonally-averaged omega composited differences (Lagrangian pressure tendency in 10^{-2} Pa s⁻¹; color shading) and vertical-meridional winds (vectors) in response to the GSF shifts in wintertime (with the meridional wind component in m s⁻¹). Both the terms have been zonally averaged along the SST front in the 50°– 68°W longitudinal range, using the SST front-following coordinate system described in section 3.2. A vector scale for vertical-meridional winds is shown in the top-right corner of each panel. (a) ERA5. (b, e) EC-Earth. (c, f) MOHC. (d, g) ECMWF. Beside institution name, the model nominal resolution in km is reported. Thick vectors indicate wind anomalies that were found to be significant at the 90% confidence level (details in section 3.2). Negative (positive) omega values correspond to upward (downward) motion. Grey contours indicate the winter climatology of vertical motion in the vicinity of the GSF, contour interval: 10^{-2} Pa s⁻¹ with dashed contours for negative. The vertical dashed line represents the GSF position in relation to which the cross-front section has been constructed.

in Figure 3.11 we present the atmospheric circulation anomalies (composite differences: "North" minus "South") along the vertical-meridional cross-section in the
50° -68°W longitudinal range. All models show upward motion anomalies directly to the south of the GSF (warm sector) and downward motion anomalies to the north, thus indicating an anomalous cell-like circulation similar in character to the one seen for the time-mean circulation (contours). The upward motion anomalies are strongest in the vicinity of the GSF and in the lower troposphere (below 600 hPa) reaching their maximum at about 850 hPa. This is in agreement with the role the mean vertical motion (term V) was found to play for the local heat budget in all models and the observations. In contrast, R100 and R50+ models differ in the meridional component of the atmospheric circulation response in proximity to the GSF. R100 models exhibit equatorward motion, whilst R50+ models rather show a poleward anomaly. This finding is also in agreement with the different role that the meridional mean advection (term III) was found to play for the local heat budget in R100 and R50+ models.

Based on the results of the local thermodynamic budget and atmospheric circulation, we have extended the heat budget to the entire North Atlantic region, in order to understand more the atmospheric response on the basin-scale. Here we show the results for the multi-model ensemble means of R100 and R50+ models in Figure 3.12 and Figure 3.13 respectively. The results for each model confirm the general features of what is described in the following (not shown). Results for ERA5 are provided in Figure 3.14. It is specified that each heat term in the thermodynamic budget extended to the entire North Atlantic region has been averaged in 700–1000 hPa considering that the diabatic heating anomalies are maximum in the lower troposphere (Figure 3.9; Kallberg et al., 2005). Figure 3.12, Figure 3.13 and Figure 3.14 show that the effect induced by meridional heat terms is not limited to areas above the GSF. Indeed, the cooling (warming) effect induced by meridional mean advection (term III) in the R100 models (R50+ models and ERA5) extends farther north and downstream the GS (Figure 3.12d, Figure 3.13d and Figure 3.14d), consistent with the surface circulation anomaly shown in Figure 3.3. In R100 models, such large-scale temperature tendencies are largely balanced by zonal mean advec-



Figure 3.12: Horizontal distribution of composited differences for zonal (terms I and II), meridional (terms III and IV) and vertical (terms V and VI) terms in the thermodynamic budget (equation 3.1) (K day⁻¹), averaged in 700–1000 hPa, as ensemble mean of models with a nominal resolution of 100 km (R100 models). (a) Diabatic heating. (b) Zonal mean advection (term I). (c) Zonal eddy heat flux (EHF) convergence (term II). (d) Meridional mean advection (term III). (e) Meridional EHF convergence (term IV). (f) Vertical mean term (term V). (g) Vertical eddy term (term VI). The winter climatological position of the GSF is indicated by the black dashed line.

tion (term I), i.e. reduced cold advection from the continent associated with reduced westerlies (Figure 3.12b). On the other hand, in R50+ models and ERA5 the warming induced by meridional mean advection north and downstream the GSF is partly balanced by zonal advection of cold air coming from the inland of North America (term I; Figure 3.13b and Figure 3.14b) and partly by meridional eddy heat flux divergence (term IV; Figure 3.13e and Figure 3.14e). The former is consistent with



Figure 3.13: Horizontal distribution of composited differences for zonal (terms I and II), meridional (terms III and IV) and vertical (terms V and VI) terms in the thermodynamic budget (equation 3.1) (K day⁻¹), averaged in 700–1000 hPa, as ensemble mean of models with a nominal resolution greater than 50 km (R50+ models). (a) Diabatic heating. (b) Zonal mean advection (term I). (c) Zonal eddy heat flux (EHF) convergence (term II). (d) Meridional mean advection (term III). (e) Meridional EHF convergence (term IV). (f) Vertical mean term (Term V). (g) Vertical eddy term (term VI). The winter climatological position of the GSF is indicated by the black dashed line.

the zonal wind intensification north of the GSF shown in Figure 3.4; the latter is consistent with the northward shifts of the stormtrack shown in Figure 3.7. In the northern portion of the North Atlantic, the effect induced by zonal mean advection is particularly intense in both models and observations. However, in R100 models the zonal mean advection is largely counterbalanced by negative diabatic heating anomalies (Figure 3.12a), whereas in R50+ models and observations it is balanced



Figure 3.14: Horizontal distribution of composited differences for zonal (terms I and II), meridional (terms III and IV) and vertical (terms V and VI) terms in the thermodynamic budget (equation 3.1) (K day⁻¹), averaged in 700–1000 hPa, for ERA5. (a) Diabatic heating. (b) Zonal mean advection (term I). (c) Zonal eddy heat flux (EHF) convergence (term II). (d) Meridional mean advection (term III). (e) Meridional EHF convergence (term IV). (f) Vertical mean term (term V). (g) Vertical eddy term (term VI). The winter climatological position of the GSF is indicated by the black dashed line.

partly by positive diabatic heating anomalies (Figure 3.13a and Figure 3.14a) and meridional eddy heat flux convergence (Figure 3.13e and Figure 3.14e). Also Hotta and Nakamura (2011) have shown that the eddy heat flux convergence is an important heating source at high latitudes. For both models and observations, the diabatic heating is quite consistent with SHF shown in Figure 3.3. Finally, both in models and observations, it is still possible to see the cooling effect induced by vertical motion above the GSF, even if weaker because of the averaging on different vertical levels (Figure 3.12f, Figure 3.13f and Figure 3.14f).

Overall, the large-scale thermodynamic budget is consistent with the local heat budget. Its added value lies in highlighting which are the mechanisms maintaining the large-scale baroclinicity anomalies shown in Figure 3.6, shedding light on the large-scale atmospheric circulation as discussed in more detail in section 3.5.

3.5 Discussion

The results discussed in the previous sections show that R100 and R50+ models exhibit a substantially different atmospheric response to the GSF shifts. The primary forcing for the atmosphere is the anomalous diabatic heating arising from the GSF shifts (effectively "replacing" cold waters with significantly warmer waters). The north-south displacement of the maximum SST gradient is an additional forcing directly affecting low-level baroclinicity, yet only locally.

The different large-scale responses featured by R50+ (realistic response) and R100 models (unrealistic response) are linked to how these models locally react to the primary oceanic forcing (SHF anomalies) to maintain their thermodynamic balance above the GSF. In R100 models, the local diabatic heating anomalies are partly balanced by meridional mean advection, while in R50+ models the meridional mean advection plays the opposite role, thus requiring a stronger sub-monthly meridional eddy heat flux divergence to maintain local thermodynamic balance.

Then, in R50+ models the local intensification of baroclinic eddy activity (which resembles the observed one) extends farther north and downstream of the GSF leading to a poleward shift in the stormtrack and the eddy-driven jet that is homodirectional to the GSF shifts, as in the observations (ERA5). Instead, R100 models fail to reproduce this large-scale response because, in the first place, they do not have the correct local circulation response to the diabatic heating anomalies.

Our results show that the mechanisms maintaining the local- and the large-scale baroclinicity are different. As previous studies show (e.g. Nakamura et al., 2004; Nakamura and Yamane, 2010), differential surface heating across oceanic fronts is key in maintaining low-level baroclinicity close to the GSF against the erosive effect of baroclinic eddy fluxes. Given that the GSF shifts are quite limited in meridional distance (approximately 50–100 km as indicated by Figure 3.2 for the two tercile categories), the zone of maximum atmospheric baroclinicity is expected to shift with the GSF by a similar distance. Therefore, this effect alone cannot explain the detected large-scale baroclinicity anomalies. The detected large-scale changes in low-level baroclinicity in R50+ and in ERA5 can be explained by the differential zonal temperature advection induced by the detected changes in zonal wind near the east coast of the North American continent. Given that in that area (around and east of Newfoundland) in winter there is strong cold advection by the westerly flow, stronger westerlies to the north and weaker westerlies to the south (as in Figure 3.4) imply a differential temperature tendency (Figure 3.13, Figure 3.14) that tends to increase baroclinicity in between (Figure 3.6). We actually suggest that there is a positive feedback between the intensification of the MEHF and the large-scale zonal wind anomalies, triggered by the local atmospheric response in the vicinity of the GSF.

In this context, another point that requires discussion is what causes the detected shift in the stormtrack and the jet (referring to R50+ models and ERA5). The respective anomalies in MEHF shown in Figure 3.7 are, indeed, consistent with collocated changes in the low-level jet following eddy-mean flow interaction arguments (Hoskins et al., 1983; Novak et al., 2015). The former, however, cannot fully explain the pattern of the zonal wind anomalies (Figure 3.4), while the divergence of the horizontal E-vector components aloft (not shown) was not found to play the expected role aiding the understanding of the changes in the jet. In contrast, the analysis of atmospheric blocking was found to be insightful as the detected anomalies are dynamically consistent with the jet and stormtrack changes. Figure 3.15 shows blocking frequency anomalies associated with the GSF shifts (composite differences "North" minus "South", as for the other diagnostics). The reduced blocking frequency over Greenland seen for R50+ models and ERA5 is in agreement with a



Figure 3.15: Blocking frequency anomalies (% of blocked days on total days; color shaded) induced by the GSF shifts in winter (DJF). (a) ERA5. (b, e) EC-Earth. (c, f) MOHC. (d, g) ECMWF. Beside institution name, the model nominal resolution in km is reported. Black dots denote anomalies that were found to be statistically significant at the 90% confidence level (details in section 3.2). Green contours indicate the winter climatology of blocking frequency every 2% from 4.5%. The winter climatological position of the GSF is indicated by black dashed line.

northerly displaced jet and stormtrack, as Greenland blocking tends to displace the jet to the south leading to higher occurrences in the southern-jet regime (Woollings and Hoskins, 2008; Woollings et al., 2010; Madonna et al., 2017). Our findings are very much in agreement with Joyce et al. (2019) who showed that periods of northerly GS path are associated with a reduction in Greenland blocking frequency and increased excursions of the stormtrack to the northeast over the Labrador Sea (namely a poleward shift). Furthermore, our results are consistent with previous studies showing that increased horizontal resolution in climate models leads to reduced biases in blocking frequency (Matsueda et al., 2009; Jung et al., 2012; Anstey et al., 2013). Thus, a better representation of Greenland blocking in R50+ models could explain why those models realistically represent the jet stream response to GSF shifts, in contrast to R100 models. However, as in the typical chicken-egg problem, further analysis are necessary to finally disentangle if they are the westerly wind anomalies (induced by eddy-mean flow interaction) to determine the anomalies in the blocking frequency or vice versa. Apart from the reduction in high-latitude blocking frequency, the observations show opposite anomalies over the European continent than models. Such anomalies are dynamically consistent with the enhanced zonal winds over the central-western European coast (Figure 3.4a) and the higher occurrence of both the central- and the northern-jet regimes (Figure 3.5a) in ERA5.

Before closing this discussion it should be mentioned that although the methodology adopted in this study aimed at isolating the oceanic forcing related to the GSF meridional shifts, the atmospheric response to this variability may be contaminated by concurrent oceanic influences from other parts of the global ocean. To assess this possibility the authors first examined the global SST composites on the "North" and "South" tercile categories defined via the GSF position and found, in fact, anomalies recalling the typical patterns of the El Niño–Southern Oscillation (ENSO), with amplitudes reaching 0.4 K in the eastern equatorial Pacific (not shown). Consistent with these SST anomalies, Figure 3.16 shows negative (positive) SLP anomalies over the western (eastern) tropical Pacific in all model simulations. On the other hand, no significant SST and SLP anomalies are found over the tropical Pacific in the ERA5 dataset. To assess the possibility of a dominant ENSO influence, the zonal wind has been linearly regressed onto ENSO (NINO3.4 index) and then the composite differences for the zonal wind at 850 hPa (Figure 3.4) have been repeated after having removed the linearly regressed zonal wind from the original data. For the models it was found that the general character and the amplitude of the detected atmospheric response in the North Atlantic is largely insensitive to this test (not shown). In ERA5 the results indicate a northward shift of the eddy-driven jet, with an amplitude and statistical significance greater than the one in the original analysis. Since no ENSO-like anomalies are present in the tropical Pacific in ERA5, such differences are understood as the effect of other phenomena concurrent with ENSO.



Figure 3.16: Sea level pressure anomalies (Pa) induced by the GSF shifts in winter (DJF). (a) ERA5. (b, e) EC-Earth. (c, f) MOHC. (d, g) ECMWF. Beside institution name, the model nominal resolution in km is reported. Black dots denote anomalies that were found to be statistically significant at the 90% confidence level (details in section 3.2). The winter climatological position of the GSF is indicated by the black dashed line.

However, excluding completely all possible influences from remote parts of the global ocean is practically impossible as the historical atmosphere-only simulations are forced with global observed SST fields. There exist elaborate statistical methods for defining a GSF signal with minimal contamination by other specific processes (such as the ENSO variability), yet it is not possible to exclude every potential influence foreign to the GSF variability. As an example, Figure 3.16 shows SLP anomalies associated with the GSF shifts that are statistically significant over various areas of the northern hemisphere. For this, one would need to repeat all simulations with observed SSTs in the area of interest and climatological SSTs elsewhere.

3.6 Chapter conclusions

The objective of this part of the PhD thesis was twofold. First, the study aimed at assessing the atmospheric response to meridional shifts in the GSF via the use of multi-model atmosphere-only simulations forced with observed SSTs. Using multiple realizations for each model was key in letting the forced signal emerge from the chaotic atmospheric variability, while comparing results between different models and the observations enhanced our confidence in the interpretation of the findings. In turn, the use of atmosphere-only simulations allowed us to study the atmospheric response to the oceanic forcing in isolation, namely, focusing on a single direction of the two-way interaction between the two realms. These choices allowed the present study to provide robust evidence on how the observed GSF variability may be impacting the North Atlantic midlatitude atmospheric circulation. Secondly, the present study aimed at assessing the role of atmospheric horizontal resolution for the realistic simulation of the above-discussed response. In fact, past studies (e.g. Smirnov et al., 2015) have examined the dependence on model resolution of the atmospheric response to idealized SST forcings mimicking the meridional shift of an oceanic front, yet our study does so for realistic SST anomalies and in the more robust context of a multi-model, protocol-driven, coordinated framework. Here we refer to model configurations with a nominal resolution coarser than $50 \,\mathrm{km}$ as R100models, and those with a nominal resolution finer than or equal to $50 \,\mathrm{km}$ as $\mathrm{R50+}$ models.

The main findings can be summarized as follows. The interannual variability

in the meridional position of the GSF was found to relate to intense localized SST anomalies, positive for the "North" position and negative for the "South" position (Figure 3.1). These SST anomalies later induce collocated anomalies in SHF (sensible and latent) representing an important portion (15–20%) of the winter SHF climatology over a significant area (Figure 3.3). This anomalous diabatic heating forces a local atmospheric circulation response which is fundamentally different between R100 and R50+ models. The response in the latter was found to be similar to the respective observed anomalies, which is not the case for R100 models.

Discussing first the local response of the R50+ models, it was found that in the presence of the anomalous diabatic heating along the GSF, thermodynamic balance near the surface is maintained mainly by anomalous meridional eddy heat flux divergence (Figure 3.10f), while the anomalous meridional mean advection (term III in equation 3.1) tends to exacerbate the temperature tendency induced by the heating. Anomalous vertical motion is generated (Figure 3.11) and this contributes significantly to counter-balancing the anomalous heating in the lower troposphere (Figure 3.10g). Effectively, the baroclinic eddy activity is modified to equilibrate the anomalous baroclinicity close to the GSF, possibly intensifying the zonal winds north of the GS. This is consistent with the detected large-scale changes in lowlevel baroclinicity (Figure 3.6), which is indeed mostly maintained by advection of cold air from inland of North America (Figure 3.13, Figure 3.14). Given the latter, in response through baroclinic adjustment, the baroclinic eddy activity is further intensified. As a result, the eddy-driven jet is also modified (Figure 3.4) through the action of the synoptic eddy fluxes (vertical divergence of E-vectors). Finally, the detected changes in blocking frequency (Figure 3.15) are dynamically consistent with the changes in the stormtrack and the jet and in agreement with previous findings (Joyce et al., 2019).

To conclude, R50+ models produce a local as well as a large-scale atmospheric circulation response with patterns similar to the respective observed anomalies. The latter consist of a shift in the North Atlantic eddy-driven jet and stormtrack that is

homo-directional to the GSF shifts (Figure 3.5). Low-resolution (R100) models produce a strongly different local circulation response to the anomalous diabatic heating associated with the GSF shifts. It is argued that these differences in the local circulation response are key in understanding also the differences in the large-scale response detected in these models. Arguably, the most important of the above-mentioned aspects is that R100 models were found to maintain thermodynamic balance near the surface not through an intense anomaly in meridional eddy heat transport but through the action of anomalous meridional mean temperature advection (Figure 3.17). This does not agree with the circulation changes found in the observations and in high-resolution (R50+) models.

It is plausible that the improvement of the response in high-resolution models is conveyed by a better representation of mesoscale ocean-to-atmosphere forcing that is not resolved in their low-resolution counterparts and/or a more realistic representation of small-scale key atmospheric features, such as fronts and conveyor belts. After all, it should be remembered that the atmosphere does not interact with a smooth and quasi-linear time mean SST front but with a much more convoluted, time-evolving SST front with gradients that are even more pronounced locally.

In the comparison between AGCMs and observations, it should be pointed out that in ERA5 the atmospheric anomalies associated with concurrent GSF shifts are, at least in part, the representation of the atmospheric forcing that caused the GSF shifts. This is in line with previous studies (Taylor and Stephens, 1998; Frankignoul et al., 2001; Sanchez-Franks et al., 2016) showing that the GSF latitude is positively correlated with the NAO at zero and at negative lags (NAO leading). Therefore, in contrast to the AGCMs, the concurrent atmospheric anomalies in ERA5 cannot be interpreted purely as an atmospheric response to the GSF shifts. Considering this important disparity between AGCMs and ERA5, the larger amplitude of the atmospheric anomalies detected in ERA5 compared to the anomalies in the R50+ models (e.g. see Figure 3.4) is indicative of a positive feedback between ocean and atmosphere: as soon as the GSF shift is established the associated SST anomalies tend to force a NAO-like response (as the R50+ models indicate) that strengthens / prolongs the original atmospheric forcing (present only in ERA5). Notably, the existence of a positive feedback of this kind, i.e. between the NAO-induced tripole SST anomalies (recalling the SST anomalies associated with GSF shifts) and the NAO, has been proposed also by other authors (Czaja and Frankignoul, 2002; Joyce et al., 2019). However, such a positive feedback is not confirmed by Wills et al. (2016) who found that the atmospheric variability pattern forcing SST changes in the Gulf Stream Extension area and the atmospheric anomalies subsequently forced by the same SST changes are spatially anticorrelated and temporally distant. Such discrepancies show that the character of the atmospheric response to SST anomalies near the GSF should be further investigated in the observations. This is especially true for monthly and shorter timescales, when the atmospheric variability is expected to be dominated by internal atmospheric processes which could hamper the emergence of the atmospheric response to SST variability.



Figure 3.17: Schematic of the atmospheric response to the GSF shifts in winter for R100 models (top) and R50+ models and the ERA5 reanalysis (bottom). The winter climatological position of the GSF is represented by the black dashed line. (1) The brown zone straddling the GSF represents the area of intense SST anomalies induced by a shift of the oceanic front from its "South" to its "North" position. (2) The strong along-front SST anomalies induce intense anomalies in surface heat fluxes (SHF), represented by vellow circles with a black point in the center, meaning that the ocean is warming the atmosphere. (3) In R100 models the diabatic heating anomaly in the vicinity of the GSF is largely balanced by mean vertical advection (MVA), meridional eddy heat transport (MET, with "eddy" meaning monthly departures from climatology) and cold meridional advection (CMA). In R50+ models MVA has a similar role, but the warm meridional advection (WMA) induces a positive temperature tendency. Thus, to maintain balance, MET is significantly stronger than in R100 models. The direction of the arrows in respect to the GSF is supposed to indicate heat transport convergence (warming for WMA, yellow) and divergence (cooling for MET and CMA, blue). (4) Downstream of the GSF, the R100 (R50+) models exhibit surface cyclonic (anticyclonic) circulation anomalies consistent with the CMA and WMA, respectively. (5) In R100 models the GSF shifts are associated with a reduction in synoptic eddy heat fluxes (v'T') in most of the North Atlantic, whereas R50+ models exhibit positive (negative) anomalies north (south) of the GSF, represented by the red (blue) shadows. (6) The eddy-driven jet shifts poleward (equatorward) in R50+ (R100) models.

Chapter 4

Non-stationarity in the NAO-Gulf Stream SST front interaction

4.1 Introduction

Recent studies have shown that the extratropical atmospheric circulation in the Northern Hemisphere is partly controlled by the GSF (Joyce et al., 2019). Specifically, the presence of the GSF has been shown to shape the climatological structure of the North Atlantic storm-track and eddy-driven jet (Brayshaw et al., 2011; O'Reilly et al., 2016; Omrani et al., 2019). Furthermore, the meridional displacements of the GSF have been shown to affect the North Atlantic tropospheric variability, with atmospheric circulation anomalies extending up to the Eurasian continent (Joyce et al., 2009; Nakamura and Yamane, 2009; Kwon and Joyce, 2013; Sato et al., 2014; Seo et al., 2017). Therefore, studying the drivers of the GSF variability is important for understanding the associated coupled (ocean–atmosphere) variability and predictability.

The GSF latitudinal position has been linked to different climate modes, such as the Atlantic meridional mode (AMM; Hameed et al., 2018; Wolfe et al., 2019), the Atlantic multidecadal oscillation (AMO; Nigam et al., 2018), the AMOC (de Coëtlogon et al., 2006; Joyce and Zhang, 2010; Pérez-Hernández and Joyce, 2014; Ezer, 2015; Sanchez-Franks and Zhang, 2015; Goddard et al., 2015; Zeng and He, 2016), the ENSO (Taylor et al., 1998; Kwon et al., 2010; Pérez-Hernández and Joyce, 2014; Sanchez-Franks et al., 2016) and the NAO (Taylor and Stephens, 1998; Joyce et al., 2000; Frankignoul et al., 2001; Hameed and Piontkovski, 2004; Kwon et al., 2010; Pérez-Hernández and Joyce, 2014; Sanchez-Franks et al., 2016; Wolfe et al., 2019). In this context, particular attention has been directed onto the NAO, which is the dominant mode of variability of the surface atmospheric circulation in the North Atlantic region (Hurrell, 1995). For example, Taylor et al. (1998) have shown that the NAO explains a much larger portion of annual variance in the GSF latitudinal position than ENSO.

The NAO can affect the GSF latitudinal position through anomalous wind-stress over the North Atlantic, by directly changing the wind-driven oceanic circulation and by exciting westward propagating Rossby waves (Gangopadhyay et al., 2016). For example, the positive phase of the NAO is associated with anomalous northward Ekman transport over the GS region (Visbeck et al., 2003) and a northward shift of the zero wind-stress curl line. The resulting wind-driven oceanic circulation anomalies are generally largest at the confluence of the SPG and the STG (Marshall et al., 2001; Reintges et al., 2017), and may thus influence the position of the GSF. Indeed, previous studies have shown that a positive NAO results in a northward displacement of the GSF, which can lag the NAO forcing by less than one month up to 2–3 years (Taylor and Stephens, 1998; Joyce et al., 2000; Frankignoul et al., 2001; Taylor and Gangopadhyay, 2001; Joyce et al., 2019; Watelet et al., 2020). Similarly, a negative NAO results in a southward displacement of the GSF with some time lag. The time lags have been related to the time that fast barotropic and slow baroclinic planetary waves require to cross the ocean, travelling from the eastern to western North Atlantic (Veronis, 1973; Gangopadhyay et al., 1992; Marshall et al., 2001; Bellucci and Richards, 2006). As an example, in a numerical study, Sasaki and Schneider (2011) have shown that the NAO can affect the GS position through westward propagating sea-level height anomalies generated by anomalous Ekman convergence in the central North Atlantic, which reach the GS area after about 2 years. Consistently, in another numerical study, de Coëtlogon et al. (2006) have shown some evidence of NAO-forced baroclinic Rossby wave propagation in the southern part of the SPG, which affects the GS path and transport after about 2 years. However, there is still no consensus on the role of Rossby wave propagation on the GSF path. Indeed, several studies have documented the characteristics of oceanic Rossby waves in North Atlantic propagating with phase velocities in the 0,93–4,17 cm s⁻¹ range (Cipollini et al., 1997, 2001; Osychny and Cornillon, 2004; de Coëtlogon et al., 2006; Årthun et al., 2021; Watelet et al., 2020). These phase velocities imply a lag between the NAO and the GSF shifts greater than what is actually observed, even for waves that are generated in the mid-Atlantic (Frankignoul et al., 2001; Wolfe et al., 2019).

Furthermore, the NAO can indirectly affect the GSF position through buoyancy fluxes in the LS (Joyce et al., 2000; Peña-Molino and Joyce, 2008; Watelet et al., 2020). The NAO is associated with anomalous SHF over the SPG, with impacts on the deep convection and thus on the export of LS water (LSW) via the DWBC. Once the deep water is formed, it is partly recirculated within the LS and along the NAC and partly travels southward, flowing along the Slope Sea(i.e. the ocean band between the GS and the eastern continental shelf of North America from Cape Hatteras to the Grand Banks; Stommel, 1958; Csanady and Hamilton, 1988; Bower et al., 2009, 2011). Zhang and Vallis (2007) have shown that the bottom vortex stretching of the DWBC transport is able to affect the GS latitudinal position by changing the intensity of the NRG, a barotropic cyclonic circulation north of the GS (Hogg, 1992). The stronger (weaker) the DWBC transport close to the Grand Banks, the stronger (weaker) the NRG and the more southward (northward) the GS. The effect of the DWBC transport on the GS latitudinal position has been suggested to be particularly relevant on the decadal and multi-decadal timescales (Zhang and Vallis, 2006). Indeed, the DWBC feeds the deeper branch of the AMOC, which has been shown to be affected by the NAO on that range of timescales (Bellucci and Richards, 2006; Bellucci et al., 2008; Reintges et al., 2017; Omrani et al., 2022).

In the context of the NAO forcing changes in the GS latitudinal position through modifying the DWBC transport, Joyce et al. (2000) have speculated on the existence of a self-sustained decadal NAO-GS coupled oscillation using a simple oceanic model. In this oscillation, a positive NAO enhances the southward LSW transport in the DWBC. The anomalies propagate southward along the eastern continental shelf of North America and reach the GS area after some years. This propagation time lag depends on the values of model parameters. The enhanced southward LSW transport in response to the positive NAO acts to move the GS separation point close to Cape Hatteras southward (Pickart and Smethie, 1993; Spall, 1996a). Consequently, negative SST anomalies are established along the GS, reducing the atmospheric transient eddy activity and influencing the NAO phase. Finally, the reversal of the NAO phase changes the buoyancy forcing in the LS and the cycle starts again with the opposite sign. In agreement with this mechanism acting on decadal timescale, Peña-Molino et al. (2011) have shown in an observational study that anomalies in LSW require approximately 4–9 years to propagate from the LS to the north-east of Cape Hatteras. Similarly, Molinari et al. (1998) have estimated a transit time from the LS to 26.5°N of 10 years. Using chlorofluorocarbon and hydrographic data, Smethie (1993) has inferred an even longer transit time. However, it is unclear whether such a decadal NAO-GS oscillation operates in reality since observational support is lacking.

Furthermore, some studies have shown that a significant part of the LSW is not exported from the LS via the DWBC but it follows other, southward-oriented, interior pathways (Rossby, 1999; Rossby and Benway, 2000; Rossby et al., 2005; Bower et al., 2009, 2011). In agreement with these studies, analyzing observational and reanalysis data, Hameed and Piontkovski (2004) and Sanchez-Franks et al. (2016) have shown that the pressure and the longitude of the Icelandic Low (IL; i.e. one NAO center of action) are good predictors for the GS latitudinal position. In particular, enhanced (reduced) and westward (eastward) shifted IL intensifies (reduces) the transport of the surface LC, thus reducing (enhancing) the cold water sources along the Slope Water (i.e. the upper part of the water column in the Slope Sea – first 500 m depth) and shifting northward (southward) the GS (Peterson et al., 2017; Holliday et al., 2020; New et al., 2021). The highest prediction skill has been obtained when enhanced and westward shifted IL leads the GSF latitudinal position by 2–3 years. However, other studies have estimated that the temperature and salinity anomalies in the Slope Water would take about 4–18 months to travel from the Newfoundland to the GS separation point (Rossby et al., 2005; Peña-Molino and Joyce, 2008; Peterson et al., 2017; New et al., 2021). This suggests a shorter time lag between the NAO forcing and the GS shifts compared to the ones proposed by Hameed and Piontkovski (2004) and Sanchez-Franks et al. (2016).

Hameed and Piontkovski (2004) have also shown that the effect of the Azores High (i.e. the other NAO center of action) on the GS latitudinal position is insignificant during 1965–2000. For this reason, they have concluded that the fluctuations in the southward flow of the LSW is dominant compared to the influence of the NAO on the GS latitudinal position through Rossby waves excited in the extratropical North Atlantic. However, they have analyzed the time period 1965–2000 as a whole. On the other hand, Sasaki and Schneider (2011), looking at the non-stationarity of time series in an oceanic model forced with observed atmospheric fluxes, provide evidence of wave propagation during 1965–1985, affecting the GS path. The different methodological approach of these two studies show that the wave-structure propagation can only be seen by analyzing the non-stationarity of time series.

Previous studies have shown that a variety of mechanisms can contribute to interactions between the NAO and the GSF, but the primary mechanisms behind the response of the GSF to the NAO in observations remain unclear. The lack of consensus is emphasized by a wide range of time lags between meridional displacements of the GSF and the NAO reported among prior studies, ranging between 0 and 4 years (Table 4.1, Figure 4.1). The discrepancies between studies may be linked to differences in the definition of the GS latitudinal position, which has been based both on surface (Taylor and Stephens, 1998; Watelet et al., 2020) and subsurface data (Joyce et al., 2000; Hameed et al., 2018) and has adopted different space-domain (Hameed et al., 2018). These discrepancies might also be linked to differences in the datasets (models, observations and reanalyses) and methodological approaches. In particular, analyses that do not consider the non-stationarity of the mechanisms involved can miss capturing some of them (e.g. the wave propagation; Hameed and Piontkovski, 2004; Sasaki and Schneider, 2011). Finally, the discrepancies might be linked to differences in the analyzed time period. In some cases, this could reflect disparities in the quality and coverage of observations, but this could also suggest that the NAO–GSF interaction is non-stationary in time (Frankignoul et al., 2001; Hameed and Piontkovski, 2004; Sasaki and Schneider, 2011; Lillibridge and Mariano, 2013; Hameed et al., 2018). For example, using a revised index for the latitudinal position of the GS north wall, Hameed et al. (2018) showed that the NAO leads GS shifts by a lag of 1 year in 1940–2014, 0 year in 1961–2014 and no correlation in 2005 - 2014.

Table 4.1 shows some examples of the time lags proposed in literature, with the respective index for the GS latitudinal position, the atmospheric forcing, the typology of the analysed dataset, the seasonality and the time period. As shown in Figure 4.1, time lags of 0–2 years between the GS meridional displacements and the NAO-related forcing are those most common in literature.

To summarize, there is a lack of consensus on the key mechanisms underlying the response of the GSF to the NAO. There are also indications of non-stationarity in the NAO–GSF interaction. However, it is difficult to draw robust conclusions due to differences among methodologies used in prior studies, including the indices of GSF variability. The goal of this part of the PhD thesis is to assess the possible non-stationarity in the NAO–GSF interaction during the winter season and to help clarify the mechanisms underlying this interaction over the last few decades using ERA5 and ORAS5 reanalysis data. Table 4.1: Time lags between the GS meridional displacements and the NAO-related forcing. Columns detail the index to define the GS latitudinal position, the index to define the NAO-related forcing, the typology of the analysed dataset, the period over which the data have been averaged, the analyzed time period, the time lag and the reference. Only the time lags where the correlation between the GS index and the NAO-related index is statistically significant at 90% and 95% significance levels are shown in the table. Refer to the respective reference for more details.

$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	GS index	NAO-related index	Data	Time mean	Time period	Time lag	Reference
$ \begin{array}{ c c c c c c c c c c c c c c c c c c c$	GSNW	NAO	Observations	DJFM	1966 - 1996	2 years	Taylor and Stephens (1998)
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	GSI	NAO	Observations	JFM	1954 - 1989	0-1 year	Joyce et al. (2000)
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	Surface salinity data	NAO	Observations	DJF	1978 - 1998	1.5 years	Rossby and Benway (2000)
IL pressure 1-3 years IL latitude 1-2 years IL longitude 3-4 years AH pressure 0-2 years	GSI	NAO	Observations	Monthly	1955–1998	12–18 months	Frankignoul et al. (2001)
GSNW IL latitude I-2 years IL longitude Observations DJF 1966–2000 3–4 years AH pressure DJF 1966–2000 0–2 years Hameed and Piontkovski (2004)	GSNW	II. pressure		7 miliaar		1_3 years	- Hameed and Piontkovski (2004)
GSNW IL longitude Observations DJF 1966–2000 3-4 years Hameed and Piontkovski (2004)		II. latitude	Observations	DJF	1966–2000	1-2 years	
GSNW Observations DJF 1966-2000 Hameed and Piontkovski (2004)		IL longitude				3-4 years	
		AH pressure				0-2 years	
AH latitude No correlation		AH latitude				No correlation	
AH longitude 2 years		AH longitude				2 voore	
CSNW Observations (ocean); 1054 1008 0.2 years	CSNW	All longitude	Observations (ocean);		1054 1008	0 2 years	
GST index NAO reanalysis (atmosphere) Annual de Coëtlogon et al. (2006) Model (ocean); de Coëtlogon et al. (2006)	GST index	. NAO	reanalysis (atmosphere) Model (ocean):	Annual	1954-1998	0-5 years	de Coëtlogon et al. (2006)
(similar to GSNW) reanalysis (atmosphere) 1948–2000 0–2 years	(similar to GSNW)		reanalysis (atmosphere)		1948-2000	0–2 years	
GSNW NAO Observations DJFM 1980–1999 1 year Chaudhuri et al. (2009)	GSNW	NAO	Observations	DJFM	1980-1999	1 year	Chaudhuri et al. (2009)
SSH data NAO Model DJFM 1960-2003 2 years Sasaki and Schneider (2011)	SSH data	NAO	Model	DJFM	1960-2003	2 years	Sasaki and Schneider (2011)
GSI Synoptic transient Observations (ocean); JFM 1979–2009 2 years Kwon and Joyce (2013)	GSI	eddy heat flux	reanalysis (atmosphere)	JFM	1979-2009	2 years	Kwon and Joyce (2013)
SSH data NAO Observations Monthly 1992–2011 6 months Lillibridge and Mariano (2013)	SSH data	NAO	Observations	Monthly	1992 - 2011	6 months	Lillibridge and Mariano (2013)
16-point GS index NAO Model DJFM 1960–2003 2 years Pérez-Hernández and Joyce (2014 (based on SSH data)	16-point GS index (based on SSH data)	NAO	Model	DJFM	1960 - 2003	2 years	Pérez-Hernández and Joyce $\left(2014\right)$
IL pressure 0–2 years	GSNW	IL pressure	Observations (ocean); reanalysis (atmosphere)	Annual	1966-2014	0–2 years	Sanchez-Franks et al. (2016)
GSNW IL longitude Observations (ocean); reanalysis (atmosphere) Annual 1966–2014 3–4 years Sanchez-Franks et al. (2016)		IL longitude				3–4 years	
NAO 0–2 years		NAO				0–2 years	
GSNW-revised NAO Observations (ocean); reanalysis (atmosphere) Annual 1940–2014 1 year Watelet et al. (2017)	GSNW-revised	NAO	Observations (ocean); reanalysis (atmosphere)	Annual	1940-2014	1 year	Watelet et al. (2017)
eGSI-0m 1961-2015 0 year	eGSI-0m	_	Observations (ocean); reanalysis (atmosphere)	Annual	1961-2015	0 year	- Hameed et al. (2018)
2005–2015 No correlation					2005-2015	No correlation	
eGSI-200m Annual <u>1961–2015</u> 0–1 year	eGSI-200m (comparable to GSI)				1961-2015	0-1 year	
(comparable to GSI) 2005–2015 No correlation					2005-2015	No correlation	
GSNW NAO Observations (ocean); <u>1966–2014</u> 0–2 years Hameed et al. (2018)	GSNW	NAO			1966-2014	0–2 years	
reanalysis (atmosphere) 2055–2014 No correlation					2055 - 2014	No correlation	
Annual 1940–2014 1 year	wGSNW			Annual	1940 - 2014	1 year	
wCSNW DJF 1940-2014 0 year				DJF	1940 - 2014	0 year	
1961–2014 0 year				Annual	1961 - 2014	0 year	
2005–2014 No correlation					2005-2014	No correlation	
IL pressure 0–2 years	eGSI-0m	IL pressure	Observations (ocean); reanalysis (atmosphere)	DJF	1961-2015	0–2 years	- - Wolfe et al. (2019)
IL latitude 0–1 year		IL latitude				0–1 year	
IL longitude Observations (ocean); DIE 10c1 2015		IL longitude				1 year	
AH pressure reanalysis (atmosphere) DJr 1901–2013 — Wolfe et al. (2019)		AH pressure				0–2 years	
AH latitude No correlation		AH latitude				No correlation	
AH longitude No correlation		AH longitude				No correlation	



Figure 4.1: Histogram of the time lags between GS meridional displacements and the NAO-related forcing suggested in literature, as reported in Table 4.1. The stars identify the time lags where the correlation between the GSF and NAO indices is statistically significant at 95% significance level in the present work.

4.2 Data and methodological approach

4.2.1 Data

The interaction between the GSF and the NAO has been assessed using ERA5 (Hersbach et al., 2020) and ORAS5 (Zuo et al., 2019) reanalysis data in the period 1950–2020 and 1958–2018, respectively. The ORAS5 reanalysis has been used as a reference for oceanic variables no t available in ERA5 reanalysis, in order to assess the mechanisms through which the NAO drives the GSF latitudinal position (see below). The ORAS5 reanalysis is forced by ERA-40 and ERA-Interim reanalysis during 1958–1978 and 1979–2014, respectively. From 2015 onwards, the ORAS5 reanalysis is forced by ECMWF/IFS operational analysis. These atmospheric products are consistent with ERA5 reanalysis in terms of representation of large-scale modes of atmospheric variability such as the NAO. Thus, using the more recent and accurate ERA5 reanalysis as a reference for the atmospheric state instead of the other ECMWF products does not affect the NAO–GSF interaction and the interpretation of results presented here.

The GSF and its latitude have been defined as described in section 3.2.2. The

GSF latitude (i.e. the GSF index) has been computed using ERA5 data. SST data are also provided in ORAS5 reanalysis, however this reanalysis exhibits SST biases in the GS region (Zuo et al., 2019) which may affect the representation of the GSF latitudinal position.

Similarly to the approach followed by Hurrell et al. (2003), the NAO has been defined using a principal component analysis of winter SLP over the North Atlantic sector (0°–80°N, 90°W–40°E). Prior to the index calculation, SLP data have been scaled by a factor $\sqrt{cos(latitude)}$ so that different grid cells contribute to the variance of the field proportionally to the area they represent. Then, the NAO index has been defined as the leading principal component of SLP.

4.2.2 Methods

Cross-wavelet analysis

In order to assess the NAO–GSF interaction and its spectral features, a cross-wavelet analysis has been applied (Torrence and Compo, 1998; Grinsted et al., 2004). The NAO and GSF time series have been linearly detrended and standardized before performing the cross-wavelet analysis. The cross-wavelet analysis is a mathematical tool developed for the analysis of the relationship between two time series in the time-frequency space (Hudgins et al., 1993). This tool allows us to study how the cospectral features evolve in time, which is useful in assessing also the non-stationarity of the link between two variables. With this tool, the relation between two time series is assessed through the cross-wavelet power spectrum, which shows domains in the time-frequency space where the two time series share high power (Torrence and Compo, 1998). However, peaks in the cross-wavelet power spectrum can appear in a given frequency range when two time series are actually independent and only one of them features significantly high power in this range (Maraun and Kurths, 2004). For this reason, another common measure for the relationship between two time series is the squared wavelet coherence, i.e. the cross-wavelet power spectrum normalized by the wavelet power spectrum of both time series. This coefficient provides a good estimate of the correlation between two time series in the time– frequency space (Grinsted et al., 2004), although spurious peaks can occur in areas with low wavelet power. Finally, the time lag between two time series is estimated through the phase of the cross-wavelet power spectrum, defined as the arctan of the ratio between its real and imaginary part.

The Morlet function has been adopted as the wavelet basis function, with nondimensional frequency ω_0 equal to 6 to satisfy the admissibility condition (Farge, 1992). The statistical significance of the NAO–GSF cross-wavelet power spectrum has been assessed with the procedure detailed in Torrence and Compo (1998, refer to their section 6c), assuming the white- and the red-noise spectra as background spectra for the NAO and the GSF, respectively. The statistical significance of the squared wavelet coherence has been assessed through a Monte Carlo test, performing 300 simulations. In both cases, the 90% significance level has been used. The phase relationship between the NAO and the GSF has been considered statistically significant when the squared wavelet coherence exceeds the 90% significance level (Kohyama et al., 2021).

The spectral features of both time series have been further assessed through the wavelet transform, in order to verify whether peaks in the NAO–GSF crosswavelet power spectrum and squared wavelet coherence are associated with peaks in the NAO and GSF wavelet power spectra. Results from the wavelet analysis applied to the single time series are presented here as global wavelet power spectra, i.e. wavelet power spectra averaged along the time dimension. The global wavelet power spectrum provides a consistent estimate of the power spectrum of a time series (Percival, 1995) and it closely approximates the Fourier spectrum (Hudgins et al., 1993; Torrence and Compo, 1998). The NAO and GSF power spectra have been assessed also with the fast Fourier transform, in order to make possible the comparison between the global wavelet and the fast Fourier power spectra. The statistical significance of global wavelet power spectra has been assessed following the procedure described in Torrence and Compo (1998) (refer to their section 5a), assuming the white- and the red-noise spectra as background spectrum for the NAO and the GSF respectively.

Lanczos band-pass filter

As described later in section 4.3, the cross-wavelet analysis shows that the NAO–GSF covariability is particularly prominent at [6-11]-year period range. In order to assess the mechanisms through which the NAO may be forcing the GSF latitudinal position on the decadal timescale, data have been temporally filtered using a Lanczos [6-11]-year band-pass filter with a window of 12 coefficients (Duchon, 1979). The number of coefficients has been subjectively chosen to balance the filtering with the number of data lost at the beginning and end of the time series. However, the general features of the lead–lag relationships between time series discussed in later sections are not highly sensitive to the number of coefficients ranging from 5 to 25.

Indices

The NAO can affect the GSF latitudinal position through anomalous wind-stress over the North Atlantic, influencing the wind-driven oceanic circulation and exciting Rossby waves (Gangopadhyay et al., 1992; Taylor and Stephens, 1998; Joyce et al., 2000; Marshall et al., 2001; Taylor and Gangopadhyay, 2001; Gangopadhyay et al., 2016), as well as through buoyancy forcing over the LS, affecting the LSW formation (Hameed and Piontkovski, 2004; Zhang and Vallis, 2007; Peña-Molino and Joyce, 2008; Sanchez-Franks et al., 2016). In order to assess the mechanisms through which the NAO drives the GSF latitudinal position, the following approaches have been adopted.

Since the NAO affects the meridional Ekman transport over the GSF area (Visbeck et al., 2003; Deser et al., 2010), the response of the wind-driven oceanic circulation to the NAO has been assessed by averaging the winter meridional Ekman transport over the $50^{\circ}-68^{\circ}$ W longitudinal range (where the GSF is defined) and the $35^{\circ}-40^{\circ}$ N latitudinal range. A positive (negative) Ekman index indicates an

anomalous northward (southward) Ekman transport over the GS region.

The presence of Rossby wave-like patterns in the extratropical North Atlantic has been assessed through the Hovmöller diagram (Hovmöller, 1949) of the bandpass filtered SST and sea surface height (SSH) anomalies averaged in the 35°–38°N latitudinal range. This region has been selected in order to be close to the GS and, at the same time, avoid relatively high latitudes. Indeed, the Rossby waves detection is particularly challenging at those latitudes, as shown by the small number of studies dealing with the Rossby waves at latitudes higher than 39°N (e.g. Osychny and Cornillon, 2004; Sasaki and Schneider, 2011; Watelet et al., 2020). However, it is here specified that results about Rossby wave propagation are quite comparable to the ones discussed in later sections if SST and SSH anomalies are averaged in latitudinal range southward to 35°–38°N. This is also the case if SST and SSH data are averaged in smaller latitudinal range (e.g. latitudinal range of 1°N).

The NAO-related atmospheric forcing of Rossby waves has been assessed through the Hovmöller diagram of the band-pass filtered wind-stress curl anomalies averaged in the 35°–37°N latitudinal range. This latitudinal range is smaller than the one used for SST and SSH anomalies in order to avoid artecfact in the wind-stress curl pattern induced by the presence of Azores islands at 25°W.

Finally, the NAO influence on the GSF path through buoyancy forcing in the LS has been assessed by analyzing the zonal oceanic transport along the Slope Sea. First of all, the zonal oceanic transport has been averaged along the $50^{\circ}-68^{\circ}W$ longitudinal range, where the GSF index has been defined. Then, the zonal transport has been integrated in the $43.5^{\circ}-45^{\circ}N$ latitudinal range and over the 1000–3000 m depth interval (refer to the black box in Figure 4.2). The sign of the integrated transport has been inverted in order to have positive values for westward transport. The latitude–depth box has been selected in order to include the core of the sub-surface oceanic transport along the Slope Sea during the winter season. This core coincides with the core of the DWBC transport. This is shown by the location of the potential vorticity minima generally used as a tracer to distinguish the DWBC from

other water masses (Figure 4.2; Talley and McCartney, 1982). For this reason, the index has been referred to as the DWBC index. It is specified that the meridional component of the DWBC transport has not been taken into account because the oceanic currents along the continental shelf of North America north of the GSF ($50^{\circ}-68^{\circ}W$) are mainly zonally oriented (not shown). Furthermore, it is specified that the vertical patterns of the zonal transport and velocity do not perfectly coincide because of the different grid resolution along the vertical direction in ORAS5 reanalysis. Indeed, the vertical resolution in the upper oceanic levels is higher than the one in the deeper levels. Then, in the calculation of the zonal transport, the zonal velocity in the upper levels is multiplied by grid areas that are lower than the ones in the deeper levels. This gives rise to zonal transport values at the upper levels that are lower compared to the ones at greater depths.

Lead–lag relationship

The lead–lag relationships between the band-pass filtered NAO, GSF, Ekman and DWBC indices have been assessed performing the lead–lag cross-correlation analysis. The statistical significance of the lead–lag cross-correlations has been assessed through bootstrap test (Ebisuzaki, 1997) at the 95% significance level. The bootstrap test has been performed by randomly re-sampling one of the two time series 1000 times.

Finally, a lead–lag linear regression analysis based on the band-pass filtered NAO index has been applied to band-pass filtered Ekman transport and SHF data in the North Atlantic and westward transport data along the Slope Sea, in order to extract the respective response to the NAO-forcing. The statistical significance of lead–lag linear regression coefficients has been assessed through a two-tailed Student's t-test against the null hypothesis of no-correlations at the 90% significance level.



Figure 4.2: Latitude–depth cross section of the winter mean zonal transport (Sv; color shading) in ORAS5 dataset averaged in the 50° – 68° W longitudinal range. The black contours indicate the winter mean potential vorticity averaged in the 50° – 68° W longitudinal range every $0.5 \ 10^{-12} \ m^{-1} \ s^{-1}$ from $2.5 \ 10^{-12} \ m^{-1} \ s^{-1}$. It is specified that the potential vorticity is lower close to the coast and it increases towards the open sea. The vertical dashed line represents the GSF latitudinal position averaged in the 50° – 68° W longitudinal range during the winter season. The black dashed box represents the area where the zonal transport is integrated to define the DWBC index. For the DWBC index, the sign of the zonal transport is reversed in order to have positive (negative) DWBC index for anomalous westward (eastward) flow. The blue box in the bottom–right inset shows where the zonal transport, the potential vorticity and the GSF latitudinal position have been averaged along the longitudinal direction. The black dashed line in the inset indicates the winter climatological position of the GSF.

4.3 Results

Both NAO–GSF cross-wavelet power spectrum and squared wavelet coherence show statistically significant values in the [6–11]-year period range and during 1972–2018 (Figure 4.3b). The peak at decadal timescales is also present both in the single NAO and GSF wavelet (Figure 4.4, Figure 4.5) and global wavelet (Figure 4.3c) power spectra, providing further evidence of the NAO–GSF interaction. However, the peak in the GSF wavelet power spectrum is statistically significant only from 1985 onwards (Figure 4.5).



Figure 4.3: a) Detrended and standardized winter GSF (black) and NAO (orange) time series in ERA5 dataset (1950–2020). b) Cross-wavelet transform of the winter GSF and NAO time series. Thick black contours encircle the cross-wavelet power spectrum values that are statistically significant at the 90% confidence level. The phase relationship between the winter GSF and NAO time series is indicated by the vectors, following the convention in Torrence and Webster (1999). In case of in-phase signals, vectors point upwards; in case of anti-phase signals, vectors point downwards. If the GSF leads the NAO, vectors point to the right; if the NAO leads the GSF, vectors point to the left. Thick red vectors indicate phase relationship where squared wavelet coherence is statistically significant at the 90% confidence level (Kohyama et al., 2021). The lower and upper dotted-dashed yellow lines represent the 6-year and 11-year periods, respectively. The Lanczos band-pass filter has been defined to retain periods in the [6-11]-year range. c) Global wavelet (bold) and Fourier (thin) power spectrum of detrended and standardized winter GSF (black) and NAO (orange) time series. The bold dashed lines represent the 90% confidence level of time-averaged red- (black) and white-noise (orange) spectra.

In the context of the NAO–GSF decadal interaction during 1972–2018, the spectral features of the cross-wavelet power spectrum are quite homogeneous in time, whereas those of the squared wavelet coherence change around 1990. Indeed, before 1990 the squared wavelet coherence is statistically significant only for periods close to 11 years, with a phase relationship between the NAO and the GSF of about 80° on average. In contrast, after 1990 the squared wavelet coherence is statistically significant in the decadal, [6–11]-year range, with a phase relationship between the two time series of about 121° on average. These results clearly indicate that the lead– lag relationship between the NAO and the GSF at the decadal timescale changes in time. Specifically, the NAO leads the GSF shifts by about 3 and 2 years, before



Figure 4.4: a) Detrended and standardized winter NAO time series in ERA5 dataset (1950–2020). b) Wavelet transform of the winter NAO time series. Thick black contours encircle the wavelet power spectrum values that are statistically significant at the 90% confidence level. c) Global wavelet (bold) and Fourier (thin) power spectrum of detrended and standardized winter NAO time series. The bold dashed lines represent the 90% confidence level of time-averaged white-noise spectra.



Figure 4.5: a) Detrended and standardized winter GSF time series in ERA5 dataset (1950–2020). b) Wavelet transform of the winter GSF time series. Thick black contours encircle the wavelet power spectrum values that are statistically significant at the 90% confidence level. c) Global wavelet (bold) and Fourier (thin) power spectrum of detrended and standardized winter GSF time series. The bold dashed lines represent the 90% confidence level of time-averaged red-noise spectra.

and after 1990, respectively. Consistently, the lead–lag cross-correlation between the band-pass filtered NAO and GSF indices shows a peak when the former leads the latter by about 3 years during 1972–1990 and by about 2 years during 1990–2018





Figure 4.6: Lead-lag cross-correlation between the band-pass filtered NAO and GSF indices during 1972–1990 (solid) and 1990–2018 (dashed). The circles highlight the lead–lag cross-correlations that are statistically significant at the 95% confidence level.

In the last part of the analyzed record (2005–2015), both the NAO–GSF crosswavelet power spectrum and the squared wavelet coherence show statistically significant values at periods shorter than 6 years, with the two time series closely in anti-phase (Figure 4.3b). This covariability peak is associated with statistically significant values also in the individual wavelet power spectra for the NAO and GSF (Figure 4.4, Figure 4.5). These results show that the NAO and the GSF covary also at frequencies higher than the decadal one but only for a limited period of time. Consistently, the covariability peak at high frequencies during 2005–2015 is associated with low power in both the NAO and GSF global wavelet power spectra (Figure 4.4c, Figure 4.5c).

Finally, another peak of covariability emerges at periods longer than 16 years (Figure 4.3b). However, the portion of the time-frequency phase space for periods longer than 16 years is largely affected by edge-effects of the cross-wavelet analysis, so results in this time-frequency region should be considered with caution. Furthermore, this peak in the cross-wavelet power spectrum is associated with a peak in the wavelet power spectrum of the GSF time series, but not for the NAO (Figure

4.4, Figure 4.5). This suggests that the NAO–GSF covariability at periods longer than 16 years could be an artifact of the strong variability of the GSF alone at these timescales (Maraun and Kurths, 2004).

The cross-wavelet analysis has been repeated using ORAS5 (1958–2018; Zuo et al., 2019) and SODA3.4.2 (1980–2020; Carton et al., 2018) as alternative ocean reanalysis. The general features of the NAO–GSF covariability described above are captured by both datasets (Figure 4.7, Figure 4.8), despite the SST biases over the GS region in the ORAS5 reanalysis and the limited time period covered by the SODA3.4.2 reanalysis. Indeed, both reanalyses show a statistically significant peak in the NAO–GSF cross-wavelet power spectrum at the decadal timescale. SODA3.4.2 also shows a statistically significant peak in the NAO–GSF squared wavelet coherence at the same timescale, with the NAO leading the GSF by about 2 years. In the ORAS5 reanalysis, the NAO–GSF covariability is confined during 1965–2012, in agreement with the results from ERA5 reanalysis. Furthermore, both reanalyses show a peak in the NAO–GSF covariability for periods shorter than 6 years after 2000 (Figure 4.7, Figure 4.8). In the SODA3.4.2 reanalysis, the peak in the NAO–GSF cross-wavelet power spectrum for periods longer than 16 years is also present. Thus, the cross-wavelet analysis applied to ORAS5 and SODA3.4.2 reanalysis datasets provide further evidence of the decadal NAO-GSF covariability.

In order to assess the ocean-atmosphere interaction at the decadal timescale, the data and the selected indices have been band-pass filtered over the [6-11]-year interval. Note that we have only used the time period where the decadal peak is present in the ERA5 dataset (1972–2018) in the analyses described below.

As mentioned in section 4.1, the NAO can force the GSF latitudinal position through the anomalous basin-scale wind-forcing and the buoyancy forcing over the LS (Taylor and Stephens, 1998; Joyce et al., 2000; Rossby and Benway, 2000; Hameed and Piontkovski, 2004; Zhang and Vallis, 2006; Joyce et al., 2019; Watelet et al., 2020). However, it is unclear how these mechanisms interact to determine the time lag between the NAO forcing and the GSF shifts. Here, we assess the response of the



Figure 4.7: a) Detrended and standardized winter GSF (black) and NAO (orange) time series in ORAS5 dataset (1958–2018). b) Cross-wavelet transform of the winter GSF and NAO time series. Thick black contours encircle cross-wavelet power spectrum statistically significant at 90% level. Phase relationship between the winter GSF and NAO time series is shown as vectors, following the convention in Torrence and Webster (1999). In case of in-phase signals, vectors point upwards; in case of anti-phase signals, vectors point downwards. If the GSF leads the NAO, vectors point to the right; if the NAO leads the GSF, vectors point to the left. Thick red vectors indicate phase relationship where squared wavelet coherence is statistically significant at 90% level (Kohyama et al., 2021). c) Global wavelet (bold) and Fourier (thin) power spectrum of detrended and standardized winter GSF (black) and NAO (orange) time series. The bold dashed lines represent the 90% confidence level of time-averaged red- (black) and white-noise (orange) spectra. The x-axis range in a) and b) is 1950–2020, in order to be comparable with the ERA5 reanalysis dataset (Figure 4.3).

wind-driven oceanic circulation to the anomalous NAO wind-stress, the excitation of Rossby waves and the response of the westward flow along the Slope Sea to the NAO-related buoyancy forcing over the LS. The assessment is performed over the individual time intervals of 1972–1990 and 1990–2018, separately. The objective is to verify which mechanism can explain the lags between the GSF shifts and the NAO forcing on the decadal timescale before and after 1990.

The lead-lag cross-correlations between the band-pass filtered NAO and Ekman indices show maximum values at lag-0 in both intervals (Figure 4.9). Specifically, a positive (negative) NAO phase induces northward (southward) Ekman transport anomalies over the GS area. The effect of the NAO forcing on the Ekman transport extends over the whole North Atlantic basin, with particularly intense anomalies



Figure 4.8: a) Detrended and standardized winter GSF (black) and NAO (orange) time series in SODA3.4.2 dataset (1980–2020). b) Cross-wavelet transform of the winter GSF and NAO time series. Thick black contours encircle cross-wavelet power spectrum statistically significant at 90% level. Phase relationship between the winter GSF and NAO time series is shown as vectors, following the convention in Torrence and Webster (1999). In case of in-phase signals, vectors point upwards; in case of anti-phase signals, vectors point downwards. If the GSF leads the NAO, vectors point to the right; if the NAO leads the GSF, vectors point to the left. Thick red vectors indicate phase relationship where squared wavelet coherence is statistically significant at 90% level (Kohyama et al., 2021). c) Global wavelet (bold) and Fourier (thin) power spectrum of detrended and standardized winter GSF (black) and NAO (orange) time series. The bold dashed lines represent the 90% confidence level of time-averaged red- (black) and white-noise (orange) spectra. The x-axis range in a) and b) is 1950–2020, in order to be comparable with the ERA5 reanalysis dataset (Figure 4.3).

in the SPG and the subtropical North Atlantic (Figure 4.10, Figure 4.11). This is consistent with previous studies showing the instantaneous impact of the NAOrelated wind-stress on the wind-driven circulation in the North Atlantic (Visbeck et al., 2003; Deser et al., 2010). Since the wind-driven oceanic circulation adjusts quickly to the wind forcing, the negative peak in the NAO–Ekman transport crosscorrelation when the NAO leads the Ekman transport can be understood as an artifact of the periodicity due to the use of the band-pass filtering applied over the decadal range.

The fast response of the wind-driven oceanic circulation to the NAO forcing also determines that the Ekman transport leads the GSF shifts by the same number of years as the NAO. This is true for the 1972–1990 interval, during which the Ekman



Figure 4.9: Lead-lag cross-correlation between the band-pass filtered NAO, GSF and Ekman indices during 1972–1990 (a) and 1990–2018 (b): NAO–GSF (black), NAO–Ekman (yellow), Ekman–GSF (blue). The first (second) variable leads in the left (right) portion of the plot. The Ekman index has been defined as the meridional Ekman velocity averaged over the GSF area (50°–68°W; 35°–38°N). The circles highlight the lead-lag cross-correlations that are statistically significant at the 95% confidence level.

transport–GSF and NAO–GSF connections reveal the same lead–lag covariability pattern (Figure 4.9a). It is not completely so during 1990–2018, when the anomalous Ekman transport anticipates the NAO forcing by 1 year (Figure 4.9b). This aspect shows that the variability of Ekman transport over the GS area is also influenced by other factors besides the NAO. Apart from that, the Ekman transport–GSF crosscorrelation shows that the anomalous oceanic circulation forced by the anomalous wind-stress over the GSF area cannot directly account for the [2–3]-year time lag in



1972-1990

Figure 4.10: Lead–lag linear regression coefficients for band-pass filtered Ekman transport (m s⁻¹; arrows) anomalies on the band-pass filtered NAO index in the winter season during 1972–1990. Only regression coefficients for band-pass filtered Ekman transport anomalies that are statistically significant at the 90% confidence level are shown. The NAO leads (lags) at negative (positive) lags, as in Figure 4.9 (orange line). The winter climatological position of the GSF is indicated by the black dashed line.

the cross-correlation between the GSF and NAO indices.

The Hovmöller diagram (Hovmöller, 1949) of the band-pass filtered SST and SSH anomalies averaged in the 35°–38°N latitudinal range shows Rossby wave-like structures propagating from the eastern to western North Atlantic before 1990 (Figure 4.12). The signal is particularly clear in the SST anomalies, whereas it appears noisier in the SSH anomalies. Positive (negative) SST and SSH anomalies in eastern North Atlantic are associated with negative (positive) wind-stress curl anomalies in


1990-2018

Figure 4.11: Same as Figure 4.10 but during 1990–2018.

the same Atlantic region (20°–35°W; Figure 4.13 and Figure 4.14). Furthermore, negative (positive) wind-stress curl anomalies in the eastern North Atlantic are associated with the positive (negative) NAO phase. This is consistent with previous studies showing that baroclinic Rossby waves can be excited by the anomalous Ekman convergence/divergence due to large-scale atmospheric circulation anomalies, such as the ones associated with the NAO (Anderson and Gill, 1975; Sturges and Hong, 1995; Sturges et al., 1998; Fu and Qiu, 2002; Zhang et al., 2016; Årthun et al., 2021; Kowalski, 2022). In this context, it has been shown that the momentum transfer between the NAO and the ocean is particularly intense around 30°W (Visbeck et al., 1998; Esselborn and Eden, 2001). Consistently, the SST and SSH anomalies

show maximum values between 25°–35°W in the period where there is evidence of Rossby wave propagation.

Once excited in the eastern North Atlantic, the positive (negative) SST and SSH anomalies reach the western North Atlantic about 2–3 years later and the GSF shifts northward (southward). Taking into account the distance between 30°W (as reference for the NAO-related perturbation initiating the Rossby waves) and 59°W (the center of the longitudinal range used to define the GSF latitudinal position), the [2–3]-year time lag corresponds to a phase speed of about 2.6–4 cm/s. These values are consistent with the phase speeds of baroclinic Rossby waves in extratropical North Atlantic proposed by previous studies (0,93–4,17 cm/s; Cipollini et al., 1997, 2001; Osychny and Cornillon, 2004; de Coëtlogon et al., 2006; Watelet et al., 2020).

Results above support the idea of the NAO forcing the GSF latitudinal position through Rossby wave excitation, but only during 1978–1990. As specified before, the spectral features of the decadal NAO–GSF covariability in this period are different compared to those from 1990 onwards (Figure 4.3b). It is suggested here that the differences in the spectral features of the decadal NAO–GSF interaction before and after 1990 may be induced by the non-stationarity of the Rossby wave propagation. The evidence of Rossby wave propagation for a limited period of time is consistent with results by Sasaki and Schneider (2011), showing westward wave propagation only during 1965–1985. After 1990, the SST anomalies show no propagation, with anomalies changing their sign in agreement with the NAO forcing. The reasons behind the absence of Rossby wave propagation after 1990 are unclear.

However, it is interesting to highlight that the NAO index shows a peak of variability on decadal timescale that is stronger during 1970–1990 (Figure 4.4). The changes in the character of the surface atmospheric forcing in time could affect the way the ocean responds to that forcing and thus the Rossby wave propagation. Previous studies have already shown that decadal changes in the NAO-related wind patterns occurring around 1990 can affect the relationship between the NAO and the interannual sea level variability along the coast of North America (Kenigson



Figure 4.12: Left panel: Band-pass filtered winter GSF (black) and NAO (orange) time series between 1972–2014. Blue stars represent years in which the NAO falls in the lower tercile category. These years are used to perform the composite analysis described in section 4.4. Right panel: Hovmöller diagram (Hovmöller, 1949) of the band-pass filtered SST (K; color shading) and SSH anomalies (cm; black contours; contours every 0.5 cm from -1.75 cm) averaged in the 35°–38°N latitudinal range during 1972–2014. It is specified that 6 years are lost at each border of the time series because of the application of the Lanczos filter (see detail in section 4.2). For this reason, the y-axis of the plot ranges during 1972–2014 and not during 1972–2018, that is where the GSF and NAO covary at the decadal timescale (Figure 4.3). The SSH time series are even shorter because the original SSH data are provided until 2018.

et al., 2018; Diabaté et al., 2021). Furthermore, it should be mentioned that the atmosphere undergoes decadal variability also projecting on other patterns largely orthogonal or/and independent to the NAO, which may also affect the oceanic cir-



Figure 4.13: Left panel: Band-pass filtered winter GSF (black) and NAO (orange) time series between 1972–2014. Right panel: Hovmöller diagram (Hovmöller, 1949) of the band-pass filtered SST (K; color shading) and wind-stress curl anomalies $(10^{-7} \text{ N m}^{-3}; \text{ black contours; contours every } 0.1 \ 10^{-7} \text{ N m}^{-3} \text{ from } -1 \ 10^{-7} \text{ N m}^{-3})$ averaged in the 35°–38°N and 35°–37°N latitudinal range during 1972–2014, respectively. It is specified that 6 years are lost at each border of the time series because of the application of the Lanczos filter (see detail in section 4.2). For this reason, the y-axis of the plot ranges during 1972–2014 and not during 1972–2018, that is where the GSF and NAO covary at the decadal timescale (Figure 4.3).

culation in the North Atlantic.

Results above also show that, unlike the Ekman transport, the Rossby wave propagation mechanism can be important in setting the time lag between the NAO and the GSF position changes, particularly before 1990. This is supported because the 3-year time lag between the NAO and GSF shifts is about the same as the time



Figure 4.14: Left panel: Band-pass filtered winter GSF (black) and NAO (orange) time series between 1972–2014. Right panel: Hovmöller diagram (Hovmöller, 1949) of the band-pass filtered SSH (cm; color shading) and wind-stress curl anomalies $(10^{-7} \text{ N m}^{-3}; \text{ black contours; contours every 0.1 } 10^{-7} \text{ N m}^{-3} \text{ from -1 } 10^{-7} \text{ N m}^{-3})$ averaged in the 35°–38°N and 35°–37°N latitudinal range during 1972–2014, respectively. It is specified that 6 years are lost at each border of the time series because of the application of the Lanczos filter (see detail in section 4.2). For this reason, the y-axis of the plot ranges during 1972–2014 and not during 1972–2018, that is where the GSF and NAO covary at the decadal timescale (Figure 4.3). The SSH time series are even shorter because the original SSH data are provided until 2018.

required for SST and SSH anomalies generated in the central and eastern North Atlantic to propagate westward across the basin (Figure 4.12).

Finally, the NAO can influence the GSF path through buoyancy forcing over the LS, affecting the export of the LSW from the LS and thus the southwestward flow along the Slope Sea (Joyce et al., 2000; Rossby and Benway, 2000; Hameed and Piontkovski, 2004; Zhang and Vallis, 2007). Figure 4.15 shows that maximum positive correlation between the band-pass filtered NAO and DWBC indices is found when the former leads the latter by 4 years. This is true during both 1972–1990 and 1990–2018. Specifically, the positive (negative) NAO enhances (reduces) the westward volume transport along the Slope Sea after 4 years, with anomalies extending to a 3000 m depth and reaching the greatest amplitude at the DWBC depth (1000– 3000 m depth; Figure 4.16, Figure 4.17). These deep water transport anomalies can be associated with the heat loss (gain) over the LS during a positive (negative) NAO phase (Figure 4.18, Figure 4.19), which is expected to affect the formation (through deep convection) and, ultimately, the export of the LSW via the DWBC. In this context, it is specified that the anomalies in the zonal flow along the Slope Sea are more intense during 1990–2018 (Figure 4.17). Furthermore, statistically significant anomalies appear at depths greater than 3000 m during 1972–1990 (Figure 4.16). These water masses are generally referred to as Overflow Water, i.e. the densest branch of the DWBC, and previous studies have shown that they are formed in the Greenland and Norwegian Seas (Worthington, 1976; Pickart, 1992). Therefore, the NAO-related buoyancy forcing over the LS is expected to play no role for the transport anomalies deeper than 3000 m.

The 4-year time lag between the response of the DWBC and the NAO forcing is consistent with the range of values shown in previous observational and modelling studies. Zhang and Vallis (2006) have shown that changes in the deep convection in the LS lead anomalies in the NRG intensity by 4 years, that is the time for the DWBC to travel from the LS to the Grand Banks. In agreement with this, Peña-Molino et al. (2011) have shown that anomalies in the uppermost layer of the DWBC (500–1000 m depth approximately) take about 4–7 years to reach the GS area. Finally, Georgiou et al. (2020) have shown that most of the dense water formed by deep convection in the LS exits the subpolar basin through indirect routes that involve exchanges between the boundary currents and the ocean interior. Such



Figure 4.15: Lead–lag cross-correlation between the band-pass filtered NAO, GSF and DWBC indices during 1972–1990 (a) and 1990–2018 (b): NAO–GSF (black), NAO–DWBC (yellow), DWBC–GSF (blue). The first (second) variable leads in the left (right) portion of the plot. The DWBC index has been defined as the total oceanic westward transport in the 43.5°–45°N latitudinal range and 1000–3000 m depth (black dashed box). The circles highlight the lead-lag cross-correlations that are statistically significant at the 95% confidence level.

routes are associated with export timescales that are usually between 1–6 years.

Figure 4.15 shows a negative peak in the cross-correlation between the DWBC and GSF indices when the former leads the latter by 3 years during 1972–1990 and by 2 years during 1990–2018. This means that the GSF shifts northward (southward) when the westward DWBC transport is reduced (intensified). This is consistent with the fact that the stronger (weaker) DWBC transport induced by the NAO forcing is expected to be associated with a stronger (weaker) bottom vortex stretching along



1972-1990

Figure 4.16: Lead–lag linear regression coefficients for band-pass filtered zonal transport (10^{-1} Sv; color shading) anomalies on the band-pass filtered NAO index in the winter season during 1972–1990. The zonal transport anomalies have been averaged in the 50°–68°W longitudinal range before to perform the linear regression. Only regression coefficients that are statistically significant at the 90% confidence level are shown. Grey contours indicate the winter climatology of the zonal oceanic transport every 1 10^{-1} Sv from -3 10^{-1} Sv. The NAO leads (lags) at negative (positive) lags, as in Figure 4.15 (orange line).

the Slope Sea and thus with a more (less) intense NRG (not shown; Zhang and Vallis, 2006, 2007). The equivalent barotropic nature of the anomalous westward flow along the Slope Sea (Figure 4.16, Figure 4.17) is the signature of the changes in the NRG. Thus, the variability of the deep-ocean circulation (DWBC) and associated changes in the NRG also manifests as variability of the upper-ocean circulation (shifts of the GSF; Peña-Molino and Joyce, 2008). In this context, the fact that the anomalous



1990-2018

Figure 4.17: Same as Figure 4.16 but during 1990–2018

DWBC transport leads the GSF shifts by 3 years during 1972–1990, and by 2 years during 1990–2018, suggests that the DWBC–GSF relationship is different for the two time periods. Furthermore, it suggests that the anomalous DWBC transport forced by the NAO forcing can directly account for the time lag in the cross-correlation between the NAO and GSF indices only after 1990. Indeed, Zhang and Vallis (2006) have shown that the more (less) intense NRG due to the stronger (weaker) DWBC transport leads the northward (southward) GSF shift by 1 year. Being the DWBC–GSF correlation significant when the former leads the latter by [1–2]-year (and not only by 2 years), the maximum changes in the GSF index are expected to lag the DWBC as well as the NAO index by 2 years. This is not the case before 1990, since



1972-1990

Figure 4.18: Lead–lag linear regression coefficients for band-pass filtered SHF (W m⁻²; color shading) anomalies on the band-pass filtered NAO index in the winter season during 1972–1990. Only regression coefficients that are statistically significant at the 90% confidence level are shown. The NAO leads (lags) at negative (positive) lags, as in Figure 4.15. The winter climatological position of the GSF is indicated by the black dashed line.

the correlation between the GSF index and the DWBC and NAO indices peak when the latters lead the former by [2–4]-year.

4.4 Discussion

Results described in the previous section show that, in the ERA5 dataset, the NAO–GSF interaction is non-stationary over the 1950–2020 period. The NAO and the GSF



1990-2018

Figure 4.19: Same as Figure 4.18 but during 1990–2018.

indices covary in the decadal timescales but only during 1972–2018 (Figure 4.3). Over this time period, we have also found that the year 1990 marks a significant discontinuity in the decadal covariability of these two indices. Specifically, the NAO leads the GSF shifts by about 3 years before 1990 and by about 2 years after 1990 (Figure 4.6). It is here noted that other studies have also discussed a discontinuity in the air–sea interaction over North Atlantic occurring around 1990 (Andres et al., 2013; Kenigson et al., 2018; Diabaté et al., 2021). Specifically, the changes in the nature of NAO-related wind-stress during the 20th century have been suggested to cause a decadal shift of the relationship between the NAO and the interannual sea level variability along the eastern coast of North America. These results further confirm the existence of discontinuity in the decadal air-sea interaction over the GS area during the 1990s. Finally, Figure 4.3 shows that the NAO and GSF indices covary at multiannual timescales, but only for a limited period of time (2005–2015)

Based on these pieces of evidence, we have assessed several mechanisms through which the NAO may be forcing the GSF path on the decadal timescale. Specifically, we have inferred the responses of wind-driven Ekman transport, Rossby waves and the DWBC transport to the NAO through lagged correlation analyses. Results suggest that the response of Ekman transport to the NAO cannot directly account for the lagged response of the GSF to the NAO. Differently, the response of DWBC transport to the NAO forcing and the Rossby wave propagation mechanism appear to be consistent with the time lag between the NAO and the GSF response, but only after and before 1990, respectively.

In order to illustrate how these mechanisms interact to set the time lag between the NAO forcing and the GSF response, a lead-lag composite analysis has been performed along the life cycle of the NAO–GSF decadal fluctuation. The analysis has been performed for years (blue stars in Figure 4.12) when the NAO is smaller than its lower tercile (equal to about -0.22), so as to derive the average evolution of the selected indices associated with the NAO forcing. The average evolution of the selected indices has been obtained by averaging their values year-by-year in a time range of 9 years centered on those below the lower tercile of the NAO. The analysis has been applied for the 1972–1990 and 1990–2018 time periods, separately. In this way, the average evolution of the GSF shifts in response to NAO forcing has been assessed when there is evidence and no evidence of Rossby wave propagation. For the 1972–1990 time period, a Rossby wave index has been defined as the band-pass filtered SST anomalies averaged in the 50°-60°W longitudinal range and then normalized. The SST anomalies have been used to define the index instead of the SSH anomalies because they are less noisy closer to the GS region. It is here specified that the size of the selected sample is too small to draw definitive conclusions from the composite analysis. Despite this limitation, the composite analysis shown in Figure

4.20 helps illustrate more clearly the previously identified relationships between the NAO, GSF, Ekman transport, Rossby waves, and DWBC transport. Figure 4.20a and Figure 4.20b show the average evolution of the GSF shifts in response to the NAO forcing during 1972–1990 and 1990–2018, respectively.

The composite in Figure 4.20a shows that a positive NAO phase is associated with positive (northward) Ekman transport anomalies over the GSF region (Figure 4.9a, Figure 4.10), which tend to shift the GSF northward (year 1–3). At the same time, negative (or eastward) anomalies in the westward DWBC transport appear along the Slope Sea. These anomalies can be interpreted as the delayed response of the deep oceanic circulation to the previous negative NAO phase (Joyce et al., 2000). Indeed, the negative NAO phase is associated with less intense zonal winds and then with downward SHF anomalies over the LS (Figure 4.18; Eden and Willebrand, 2001; Omrani et al., 2022). The air-sea heat exchanges over the LS weaken the deep convection and the export of the LSW via the DWBC. The changes in the DWBC transit along the Slope Sea 4 years after the NAO peak and the GSF shifts northward (Figure 4.15a, Figure 4.16). Indeed, the weakened DWBC transport is expected to contract the NRG via the anomalous bottom vortex stretching (Zhang and Vallis, 2007). Finally, the GSF is pushed further northward by the Rossby waves excited by the NAO forcing on the eastern North Atlantic and reaching the GSF area after about 2 years (Figure 4.12; year 3–4). Due to the combined effects of all the mechanisms, the tendency of the GSF index shows a positive trend during the positive NAO phase (year 1–3), reaching its maximum value concurrently with the arrival of the Rossby waves in the GSF area.

At the same time, the NAO reverses the sign and the tendency of the GSF index starts to decrease (year 3–4). However, the GSF still shifts northward and reaches the maximum northern position 1 year later, with a lag of about 3 years respect the positive NAO peak (year 4–5). This is mainly due to the persisting effect of the Rossby waves that counteract the opposite effect of the anomalous Ekman and DWBC transport during the onset of the negative NAO phase. The negative NAO phase is associated with negative (southward) Ekman transport anomalies over the GSF region, which tend to shift the GSF southward (year 4–6). Contemporarily, the positive (westward) anomalies in the DWBC transport induced by the previous positive NAO phase (4 years before) transit along the Slope Sea, pushing further southward the GSF (year 4–6). Finally, the GSF moves further southward because of the sea level anomalies induced by the Rossby waves excited by the negative NAO about 2 years before (year 7–8). Consistently, the tendency of the GSF index shows a negative trend during the negative NAO phase (year 3–7), reaching its minimum value concurrently with the arrival of the Rossby waves in the GSF area. Thanks to their persisting effect, the GSF reaches the maximum southern position 1–2 years later, with a lag of about 3 years respect the negative NAO peak (year 8–9). At this point, the NAO reverses the sign and the cycle starts again.

The phase evolution of the GSF shifts in response to the decadal NAO forcing during 1972–1990 (Figure 4.20a) is quite comparable with the one during 1990– 2018 (Figure 4.20b). However, the GSF shifts lag the NAO forcing by 3 years during 1972–1990, whereas by 2 years during 1990–2018 (Figure 4.6, Figure 4.20). It is here suggested that the differences in the lead-lag NAO-GSF relationship between the two time periods are induced by the non-stationarity of Rossby wave propagation. Indeed, the Rossby waves excited by the NAO forcing on the eastern North Atlantic require about 2–3 years to reach the GS area. This means that the sea level anomalies associated with the wave propagation reach the GSF area when the anomalous Ekman and DWBC transports are in the transition phase from positive to negative values or from negative to positive values (year 3-4; year 7-8). Thus, the tendency of the GSF shifts is extended by about one year. On the other hand, in the absence of Rossby wave propagation, the tendency of the GSF shifts is maximum when the anomalous Ekman and DWBC transport reach their greatest intensity and it immediately reverses once the Ekman and DWBC transport change sign. In this context, the GSF reaches its maximum northern/southern position 2 years after the peak in the GSF tendency, with a lag of about 2 years respect the

NAO peak. This fact is interpreted as the combined effects of anomalous Ekman and DWBC transport. Thus, results above suggest that the Rossby wave propagation delays the shift of the front, changing its lag respect the NAO forcing during the two different periods of time.



Figure 4.20: a) The average evolution of the GSF shifts (solid black line), GSF tendency (dashed black line), the NAO (orange line), the Ekman index (green line), the DWBC index (blue line) and the Rossby waves index during 1972–1990. b) The average evolution of the GSF shifts (solid black line), GSF tendency (dashed black line), the NAO (orange line), the Ekman index (green line) and the DWBC index (blue line) during 1990–2018. The average evolution of the selected indices during the two time periods has been defined through a lead–lag composite analysis performed for years (blue stars in Figure 4.12) when the NAO is smaller than its lower tercile.

In the context of the GSF shifts evolution described above, it is specified that the drivers of the decadal NAO variability are outside the scope of present PhD thesis. However, the reversal of the NAO phases (Figure 4.20; year 3–4 and 7–8) seems to be consistent with the delayed negative feedback on the SST caused by changes in the AMOC, as proposed by previous studies (e.g. Reintges et al., 2017). Indeed, the positive NAO phase causes enhanced heat losses over the SPG (Eden and Willebrand, 2001; Omrani et al., 2022), which in turn leads to the delayed intensification of the meridional overturning through an increase of the DWBC transport (Figure 4.15). The AMOC intensification leads to a warmer SPG (through enhanced meridional heat transport), determining a phase reversal of the NAO in response to the reduced large-scale meridional SST gradient associated with the anomalously warm SPG (Marshall et al., 2001; Delworth et al., 2017). The reverse is true for the negative NAO phase. Furthermore, other processes could play a role in the decadal NAO variability associated with the GSF shifts. For example, there are some hints about the possible role played by the North Pacific sector on the decadal NAO variability. Indeed, large-scale SST and SLP anomalies are found in the North Pacific sector in association with the GSF shifts (not shown). Previous studies have shown that the North Pacific–North Atlantic interaction can account for a great percentage of the variance over the North Atlantic sector (Bjerknes, 1966; Honda et al., 2001; Honda and Nakamura, 2001). It is speculated that such interaction could have also a role on the decadal NAO variability.

To summarize, results in this part of the PhD thesis suggest that the decadal NAO variability can affect the decadal GSF shifts through changes in deep convection and thus by affecting the DWBC transport (Figure 4.15). The peak in the NAO–DWBC cross-correlation when the former leads the latter by 4 years is consistent with the transit time for anomalies in the LS to reach the GS area via the DWBC as suggested by previous observational and modeling studies (Peña-Molino et al., 2011; Georgiou et al., 2020). Furthermore, the role of the DWBC transport on the GSF path is suggested by the equivalent barotropic nature of the anomalous zonal flow along the Slope Sea (Figure 4.16, Figure 4.17; Peña-Molino and Joyce, 2008). However, the role of the LC cannot be excluded. The positive NAO is expected to

cause a northward expansion of the STG and a reduction of the cold water sources along the Slope Water (Peterson et al., 2017; Holliday et al., 2020; New et al., 2021). In agreement with this, the positive NAO phase is associated with reduced westward flow in the upper 500 m depth. The reverse is true for the negative NAO phase. The fast wind-driven oceanic circulation response to wind forcing could explain why the correlation between the NAO and the anomalous zonal flow at the Slope Water depth peaks at lag-0. It is here specified that the lead–lag cross-correlation between the NAO and the anomalous zonal flow at the Slope Water depth is strictly comparable to Figure 4.15, being the zonal flow along the Slope Sea equivalent barotropic. Hence, the changes in the zonal transport along the Slope Sea could be interpreted as the fast response of the LC and the lagged response of the DWBC to the NAO forcing. However, the barotropic nature of the flow along the Slope Sea makes it difficult to differentiate the role of the two currents.

Before closing this section, the possible causes of the non-stationarity in the NAO–GSF relation and associated mechanisms will be discussed. Firstly, we have to consider that the quality of observational data has greatly improved during the last decades. In particular, the absence of the decadal NAO–GSF covariability before 1972 could be an artifact due to a lack of satellite observations during that time period rather than an effective independence between the two time series. The quality of data could also cause the discrepancies in the lead–lag NAO–GSF relationship before and after 1990. The satellite observations provide highly resolved SST data, which are able to properly capture GSF shifts of approximately 50–100 km. This was hardly possible with observational datasets before 1979. However, the limited length of satellite observations can question the spectral features of the NAO–GSF covariability, especially over decadal and multidecadal timescales. In this context, the NAO–GSF covariability described in the current work could be seen as a part of a damped oscillation working on longer timescales, in which periods of weak/non-covariability alternate periods of strong covariability.

Furthermore, it should be taken into account that the GS latitudinal position is

potentially affected by a number of phenomena other than the NAO, such as the AMM (Hameed et al., 2018; Wolfe et al., 2019), the AMO (Nigam et al., 2018), the AMOC (de Coëtlogon et al., 2006; Joyce and Zhang, 2010) and the ENSO (Pérez-Hernández and Joyce, 2014; Sanchez-Franks et al., 2016). At the same time, the atmosphere undergoes decadal variability also projecting on patterns largely orthogonal or/and independent to the NAO. Thus, the decadal NAO–GSF covariability could be supported or not depending on the slowly varying atmospheric and oceanic background and the eventual phase-locking of the decadal NAO and GSF variability. This aspect could be relevant to explain the dependency of the NAO–GSF lead–lag relationship over time as well as the lack of decadal NAO–GSF covariability before 1972.

Further studies using longer time records of highly resolved SST data and model experiments are highly required for assessing the robustness of the non-stationarity in the NAO–GSF relation and deeper understanding of its possible causes and associated mechanisms.

4.5 Chapter conclusions

The interaction between the NAO and the GSF latitudinal position has been the subject of extensive investigations. In this context, there are indications of non-stationarity in their interaction but the differences among methodologies used in prior studies make it difficult to draw consistent conclusions. Furthermore, there is a lack of consensus on the key mechanisms underlying the response of the GSF to the NAO. The goal of this part of the PhD thesis was to assess the possible non-stationarity in the NAO–GSF interaction and to help clarify the mechanisms underlying this interaction over the last few decades, using a set of global atmosphere and ocean reanalysis.

Results show that the interaction between the NAO and the GSF latitudinal shifts during the winter season is non-stationary over the 1950–2020 time period. The NAO and the GSF indices covary on the decadal timescales (periods in [6–11]- year range) but only during 1972–2018 (Figure 4.3). A secondary peak in the NAO– GSF covariability emerges on multiannual timescales (periods shorter than 6 years) but only for a limited period of time (2005–2015). The detection of these decadal and multiannual peaks in different observational products provides robustness to this finding (Figure 4.7, Figure 4.8).

The non-stationarity in the decadal NAO–GSF covariability is also manifested through the dependency of their lead–lag relationship on the analyzed time period. Indeed, the NAO leads the GSF shifts by 3 years during 1972–1990 and by 2 years during 1990–2018 (Figure 4.6). Note that this may in part explain the discrepancies in the lead–lag relationships between the GSF and the NAO suggested in previous studies (Table 4.1). Depending on the analysed time period, the lead–lag relationship between the GSF latitudinal position and its drivers may be different.

Overall, the results show that the response of the GSF to the NAO on decadal timescales can be interpreted as the joint effect of several different mechanisms, which are not all stationary across the 1972–2018 period. Before 1990, we interpret the time lag between GSF shifts and NAO to reflect a combination of a quick Ekman response, a lagged response of the DWBC, and the propagation of Rossby waves in response to the NAO (Figure 4.20a). After 1990, we find an absence of Rossby wave propagation and thus the overall response of the GSF to the NAO is influenced only by the responses of Ekman and DWBC transports to the NAO (Figure 4.20b). Here it is suggested that the non-stationarity of the Rossby wave propagation causes the time lag between the NAO and the GSF latitudinal position on decadal timescales to differ before and after 1990.

Chapter 5

Conclusions

The present PhD thesis focused on the ocean–atmosphere interaction associated with the GSF variability, addressing the following scientific questions:

- 1. What is the character of the atmospheric response to the interannual GSF meridional shifts and its dependence on the model horizontal resolution?
- 2. What are the spectral features of the NAO–GSF interaction and the mechanisms through which the NAO forces the GSF meridional shifts on the decadal timescale?

The character of the atmospheric response to the interannual GSF meridional shifts was assessed in the ERA5 reanalysis dataset and in an ensemble of atmosphereonly historical simulations (HighResMIP dataset) forced with observed SSTs (1950– 2014). The role of the model horizontal resolution was assessed analyzing models with a nominal horizontal resolution from 25 km to 100 km. We referred to model configurations with a nominal resolution coarser than 50 km as R100 models, whereas to those with a nominal resolution finer or equal to 50 km as R50+ models. It is highlighted that the GSF variability was inspected through the definition of a new SST-based index (see section 3.2.2 for details). Furthermore, the analyses in the vertical–meridional cross-section were developed along an SST front following coordinate system, which represents a novelty in the scientific literature dealing with the oceanic fronts variability and their impact on the atmospheric circulation (see section 3.2.4 for details).

Results show that the interannual variability in the GSF meridional position is associated with intense SST anomalies straddling its climatological position. Specifically, a northward (southward) shift of the front induces positive (negative) SST anomalies (Figure 3.1). Such SST anomalies induce intense co-located SHF anomalies which represent a thermal forcing for the atmospheric circulation (Figure 3.3).

The local- and large-scale atmospheric response to the SST-induced diabatic heating anomalies is strongly resolution-dependent, with the response in R50+ models resembling the respective observed anomalies.

Discussing first the local response of the R50+ models, the thermodynamic balance in the lower troposphere is mainly maintained by anomalous meridional heat flux divergence (Figure 3.10f; term IV in equation 3.1) and the anomalous vertical thermal advection (Figure 3.10g; term V in equation 3.1). As expected for baroclinic adjustment, the local intensification of the baroclinic eddy activity is the response to the enhanced near-surface baroclinicity maintained by differential surface heating across the GSF (Figure 3.6). The eddy activity intensifies also north and downstream of the GSF but in this case it is the meridionally differential zonal temperature advection near the east coast of the North American continent that maintains the large-scale baroclinicity (Figure 3.13). The changes in eddy activity shift northward the eddy-driven jet (Figure 3.4), through the action of the synoptic eddy fluxes (Figure 3.7, Figure 3.8). The northward shift of the eddy-driven jet and stormtrack is consistent with large-scale anticyclonic circulation anomalies downstream the GSF (Figure 3.3).

The R100 models show a strongly different local circulation response, with the anomalous diabatic heating mainly balanced by anomalous meridional heat flux divergence, vertical mean advection and meridional mean advection (Figure 3.10b, Figure 3.10c). As in the R50+ models, the local near-surface baroclinicity is enhanced thanks to differential heating across the front. Differently, the large-scale baroclinicity is reduced, with an expected reduction in the eddy activity over the

North Atlantic and the southward shift of the eddy-driven jet. The changes in the eddy-driven jet and stormatrack are associated with large-scale cyclonic circulation anomalies.

Interestingly, R50+ models produce a local- as well as a large-scale atmospheric circulation response comparable to the respective observed anomalies. This finding provides important insight into the real character of the atmospheric response to the interannual GSF variability. The improvement in models with a resolution finer than 50 km may be conveyed by a better representation of mesoscale ocean-to-atmosphere forcing that is not resolved in their low-resolution counterparts and/or a more real-istic representation of small-scale key atmospheric features (e.g. atmospheric fronts and warm conveyor belts).

The similarity between R50+ models and observations also suggests the possible existence of a positive feedback between ocean and atmosphere. As soon as the NAO forces a GSF shift, the associated SST anomalies tend to force a NAO-like response that strengthens/prolongs the original atmospheric forcing. The existence of a similar positive feedback has been previously suggested by other studies (Czaja and Frankignoul, 2002; Joyce et al., 2019).

After having assessed the atmospheric response to interannual GSF variability, I moved on analyzing the other direction of ocean–atmosphere interaction, that is the role of the atmospheric forcing on the GSF front variability. In this context, I assessed both the possible existence of non-stationarity in the NAO–GSF interaction and the mechanisms through which the NAO may be forcing the GSF shifts on decadal timescales. The spectral features of the NAO–GSF interaction in the past decades and the mechanisms underlying this interaction were assessed in a set of global atmosphere and ocean reanalyses.

Results show that the interaction between the NAO and the GSF latitudinal position during the winter season is non-stationary over the 1950–2020 period. This is manifested through the presence of a covariability peak on decadal timescales only during 1972–2018 (Figure 4.3). A secondary peak in the NAO–GSF covariability emerges on multiannual timescales but only during 2005–2015. The detection of the decadal and multiannual peaks in different reanalysis products (ERA5, ORAS5 and SODA3.4.2) provides robustness to these findings (Figure 4.7, Figure 4.8). Finally, the non-stationarity in the decadal NAO–GSF covariability is also shown by the dependency of their lead–lag relationship during 1972–2018. Indeed, the NAO leads the GSF latitudinal shifts by 3 years during 1972–1990 and by 2 years during 1990–2018 (Figure 4.6).

Overall, the results show that the response of the GSF to the NAO on decadal timescales can be interpreted as the joint effect of several different mechanisms, which are not all stationary across the 1972–2018 period. Before 1990, we interpret the time lag between GSF shifts and NAO to reflect a combination of a quick Ekman response, a lagged response of the DWBC, and the propagation of Rossby waves in response to the NAO (Figure 4.20a). After 1990, we find an absence of Rossby wave propagation and thus the overall response of the GSF to the NAO is influenced only by the responses of Ekman and DWBC transports to the NAO (Figure 4.20b). Here it is suggested that the non-stationarity of the Rossby wave propagation causes the time lag between the NAO and the GSF latitudinal position on decadal timescales to differ before and after 1990.

To conclude, this PhD thesis helps to improve our understanding of extratropical climate variability. In fact, it provides important evidence on the role played by small-scale oceanic features, such as the oceanic fronts, on extratropical atmospheric variability.

In particular, results suggest the possible existence of a positive feedback between the GSF and the atmospheric circulation. As soon as the NAO forces a GSF shift, the associated SST anomalies tend to force a NAO-like response that strengthens/prolongs the original atmospheric forcing. This evidence suggests that the atmospheric predictability may be higher than expected, especially on interannual and longer timescales. Given the oceanic state is predictable to some extent, the present results suggest that there are higher chances to properly predict changes in the atmospheric circulation through state-of-the-art high-resolution models.

Furthermore, this PhD thesis shows that the ocean-atmosphere interaction associated with the GSF variability is non-stationary. The non-stationarity in the NAO-GSF covariability has implications for the predictability of the North Atlantic sector. Previous studies have proposed statistical models to predict the GS path based on the atmospheric state some years in advance (e.g. Hameed and Piontkovski, 2004; Sanchez-Franks et al., 2016). However, the relationship between the GS latitudinal position and the atmospheric indices has been generally considered stationary. Thus, results in the present thesis indicate the potential inadequacy of statistical models that do not take into account the non-stationary character of the NAO-GSF interaction. Furthermore, considering the impact that the GSF shifts have on the North Atlantic atmospheric circulation, errors in predicting the GSF latitudinal position can reduce our skills in predicting the atmospheric variability through statistical and dynamical predictions.

References

- Ambaum, M. H. P. and Athanasiadis, P. J. (2007). The response of a uniform horizontal temperature gradient to heating. *Journal of the Atmospheric Sciences*, 64(10):3708 – 3716.
- Ambaum, M. H. P. and Hoskins, B. J. (2002). The nao troposphere–stratosphere connection. *Journal of Climate*, 15(14):1969 – 1978.
- Ambaum, M. H. P., Hoskins, B. J., and Stephenson, D. B. (2001). Arctic oscillation or north atlantic oscillation? *Journal of Climate*, 14(16):3495 – 3507.
- Ambaum, M. H. P. and Novak, L. (2014). A nonlinear oscillator describing storm track variability. Quarterly Journal of the Royal Meteorological Society, 140(685):2680–2684.
- Anderson, D. L. and Gill, A. (1975). Spin-up of a stratified ocean, with applications to upwelling. *Deep Sea Research and Oceanographic Abstracts*, 22(9):583–596.
- Andres, M., Gawarkiewicz, G. G., and Toole, J. M. (2013). Interannual sea level variability in the western north atlantic: Regional forcing and remote response. *Geophysical Research Letters*, 40(22):5915–5919.
- Andrews, D. G., Holton, J. R., and Leovy, C. B. (1987). Middle atmosphere dynamics. Academic Press, page 489.
- Anstey, J. A., Davini, P., Gray, L. J., Woollings, T. J., Butchart, N., Cagnazzo, C.,

Christiansen, B., Hardiman, S. C., Osprey, S. M., and Yang, S. (2013). Multimodel analysis of northern hemisphere winter blocking: Model biases and the role of resolution. *Journal of Geophysical Research: Atmospheres*, 118(10):3956–3971.

- Athanasiadis, P., Yeager, S., Kwon, Y., et al. (2020). Decadal predictability of North Atlantic blocking and the NAO. *npj Climate and Atmospheric Science*, 3(20).
- Athanasiadis, P. J., Bellucci, A., Hermanson, L., Scaife, A. A., MacLachlan, C., Arribas, A., Materia, S., Borrelli, A., and Gualdi, S. (2014). The Representation of Atmospheric Blocking and the Associated Low-Frequency Variability in Two Seasonal Prediction Systems. *Journal of Climate*, 27(24):9082–9100.
- Baldwin, M. P. and Dunkerton, T. J. (1999). Propagation of the arctic oscillation from the stratosphere to the troposphere. *Journal of Geophysical Research: Atmospheres*, 104(D24):30937–30946.
- Baldwin, M. P. and Dunkerton, T. J. (2001). Stratospheric harbingers of anomalous weather regimes. *Science*, 294(5542):581–584.
- Barnston, A. G. and Livezey, R. E. (1987). Classification, seasonality and persistence of low-frequency atmospheric circulation patterns. *Monthly Weather Review*, 115(6):1083 – 1126.
- Barsugli, J. J. and Battisti, D. S. (1998). The basic effects of atmosphere–ocean thermal coupling on midlatitude variability. *Journal of the Atmospheric Sciences*, 55(4):477 – 493.
- Battisti, D. S., Bhatt, U. S., and Alexander, M. A. (1995). A modeling study of the interannual variability in the wintertime north atlantic ocean. *Journal of Climate*, 8(12):3067 – 3083.
- Bellucci, A., Athanasiadis, P. J., Scoccimarro, E., Ruggieri, P., Gualdi, S., Fedele, G., Haarsma, R. J., Garcia-Serrano, J., Castrillo, M., Putrahasan, D., et al. (2021). Air-sea interaction over the gulf stream in an ensemble of highresmip present climate simulations. *Climate Dynamics*, 56:2093–2111.

- Bellucci, A., Gualdi, S., and Scoccimarro, E. e. a. (2008). Nao–ocean circulation interactions in a coupled general circulation model. *Climate Dynamics*, 31:759–777.
- Bellucci, A. and Richards, K. J. (2006). Effects of nao variability on the north atlantic ocean circulation. *Geophysical Research Letters*, 33(2).
- Bjerknes, J. (1966). A possible response of the atmospheric hadley circulation to equatorial anomalies of ocean temperature. *Tellus*, 18(4):820–829.
- Bower, A., Lozier, S., Biastoch, A., Drouin, K., Foukal, N., Furey, H., Lankhorst, M., Rühs, S., and Zou, S. (2019). Lagrangian views of the pathways of the atlantic meridional overturning circulation. *Journal of Geophysical Research: Oceans*, 124(8):5313–5335.
- Bower, A., Lozier, S., and Gary, S. (2011). Export of labrador sea water from the subpolar north atlantic: A lagrangian perspective. Deep Sea Research Part II: Topical Studies in Oceanography, 58(17):1798–1818. Climate and the Atlantic Meridional Overturning Circulation.
- Bower, A. S., Lozier, M. S., Gary, S. F., and Böning, C. W. (2009). Interior pathways of the north atlantic meridional overturning circulation. *Nature*, 459(7244):243– 247.
- Brayshaw, D., Hoskins, B., and Blackburn, M. (2011). The Basic Ingredients of the North Atlantic Storm Track. Part II: Sea Surface Temperatures. *Journal of the Atmospheric Sciences*, 68(8):1784–1805.
- Brayshaw, D. J., Hoskins, B., and Blackburn, M. (2009). The Basic Ingredients of the North Atlantic Storm Track. Part I: Land–Sea Contrast and Orography. *Journal of the Atmospheric Sciences*, 66(9):2539–2558.
- Brown, O. B., Cornillon, P. C., Emmerson, S. R., and Carle, H. (1986). Gulf stream warm rings: a statistical study of their behavior. *Deep Sea Research Part A. Oceanographic Research Papers*, 33(11):1459–1473.

- Carton, J. A., Chepurin, G. A., and Chen, L. (2018). Soda3: A new ocean climate reanalysis. Journal of Climate, 31(17):6967 – 6983.
- Cassou, C., Deser, C., and Alexander, M. A. (2007). Investigating the impact of reemerging sea surface temperature anomalies on the winter atmospheric circulation over the north atlantic. *Journal of Climate*, 20(14):3510 – 3526.
- Cayan, D. R. (1992). Latent and Sensible Heat Flux Anomalies over the Northern Oceans: The Connection to Monthly Atmospheric Circulation. *Journal of Climate*, 5(4):354–369.
- Chassignet, E. P. and Marshall, D. P. (2008). Gulf Stream Separation in Numerical Ocean Models, pages 39–61. American Geophysical Union (AGU).
- Chaudhuri, A. H., Gangopadhyay, A., and Bisagni, J. J. (2009). Interannual variability of gulf stream warm-core rings in response to the north atlantic oscillation. *Continental Shelf Research*, 29(7):856–869.
- Chelton, D. B., Esbensen, S. K., Schlax, M. G., Thum, N., Freilich, M. H., Wentz, F. J., Gentemann, C. L., McPhaden, M. J., and Schopf, P. S. (2001). Observations of coupling between surface wind stress and sea surface temperature in the eastern tropical pacific. *Journal of Climate*, 14(7):1479 – 1498.
- Chelton, D. B., Schlax, M. G., Freilich, M. H., and Milliff, R. F. (2004). Satellite measurements reveal persistent small-scale features in ocean winds. *Science*, 303(5660):978–983.
- Ciasto, L. M. and Thompson, D. W. J. (2004). North Atlantic Atmosphere-Ocean Interaction on Intraseasonal Time Scales. *Journal of Climate*, 17(8):1617–1621.
- Cipollini, P., Cromwell, D., Challenor, P. G., and Raffaglio, S. (2001). Rossby waves detected in global ocean colour data. *Geophysical Research Letters*, 28(2):323–326.
- Cipollini, P., Cromwell, D., Jones, M. S., Quartly, G. D., and Challenor, P. G. (1997).

Concurrent altimeter and infrared observations of rossby wave propagation near 34°n in the northeast atlantic. *Geophysical Research Letters*, 24(8):889–892.

- Cohen, J. (2016). An observational analysis: Tropical relative to arctic influence on midlatitude weather in the era of arctic amplification. *Geophysical Research Letters*, 43(10):5287–5294.
- Cohen, J., Screen, J. A., Furtado, J. C., Barlow, M., Whittleston, D., Coumou, D., Francis, J., Dethloff, K., Entekhabi, D., Overland, J., and Jones, J. (2014). Recent arctic amplification and extreme mid-latitude weather. *Nature Geoscience*, 7(9):627–637.
- Cornillon, P. (1986). The effect of the new england seamounts on gulf stream meandering as observed from satellite ir imagery. *Journal of Physical Oceanography*, 16(2):386 – 389.
- Cornillon, P., Gilman, C., Stramma, L., Brown, O., Evans, R., and Brown, J. (1987). Processing and analysis of large volumes of satellite-derived thermal infrared data. *Journal of Geophysical Research: Oceans*, 92(C12):12993–13002.
- Csanady, G. and Hamilton, P. (1988). Circulation of slopewater. Continental Shelf Research, 8(5):565–624. Shelf Edge Exchange Processes of the Mid-Atlantic Bight.
- Czaja, A. and Frankignoul, C. (2002). Observed Impact of Atlantic SST Anomalies on the North Atlantic Oscillation. *Journal of Climate*, 15(6):606–623.
- Czaja, A., Frankignoul, C., Minobe, S., and other (2019). Simulating the Midlatitude Atmospheric Circulation: What Might We Gain From High-Resolution Modeling of Air-Sea Interactions? *Current Climate Change Reports*, 5:390–406.
- Czaja, A., Robertson, A. W., and Huck, T. (2003). The Role of Atlantic Ocean-Atmosphere Coupling in Affecting North Atlantic Oscillation Variability, pages 147–172. American Geophysical Union (AGU).

- Davini, P., Cagnazzo, C., Gualdi, S., and Navarra, A. (2012). Bidimensional Diagnostics, Variability, and Trends of Northern Hemisphere Blocking. *Journal of Climate*, 25(19):6496–6509.
- de Coëtlogon, G., Frankignoul, C., Bentsen, M., Delon, C., Haak, H., Masina, S., and Pardaens, A. (2006). Gulf stream variability in five oceanic general circulation models. *Journal of Physical Oceanography*, 36(11):2119 – 2135.
- Delworth, T. L., Zeng, F., Zhang, L., Zhang, R., Vecchi, G. A., and Yang, X. (2017). The central role of ocean dynamics in connecting the north atlantic oscillation to the extratropical component of the atlantic multidecadal oscillation. *Journal of Climate*, 30(10):3789 – 3805.
- Deser, C. (2000). On the teleconnectivity of the "arctic oscillation". *Geophysical Research Letters*, 27(6):779–782.
- Deser, C., Alexander, M. A., Xie, S. P., and Phillips, A. S. (2010). Sea surface temperature variability: patterns and mechanisms. *Annual review of marine science*, 2:115–143.
- Diabaté, S. T., Swingedouw, D., Hirschi, J. J.-M., Duchez, A., Leadbitter, P., Haigh,
 I. D., and McCarthy, G. D. (2021). Western boundary circulation and coastal sealevel variability in northern hemisphere oceans. *Ocean Science*, 17:1449–1471.
- Dong, S. and Kelly, K. A. (2004). Heat Budget in the Gulf Stream Region: The Importance of Heat Storage and Advection. *Journal of Physical Oceanography*, 34(5):1214–1231.
- Drinkwater, K. F., Belgrano, A., Borja, A., Conversi, A., Edwards, M., Greene,
 C. H., Ottersen, G., Pershing, A. J., and Walker, H. (2003). The Response of Marine Ecosystems to Climate Variability Associated with the North Atlantic Oscillation, pages 211–234. American Geophysical Union (AGU).

Ducet, N., Le Traon, P. Y., and Reverdin, G. (2000). Global high-resolution mapping

of ocean circulation from topex/poseidon and ers-1 and -2. *Journal of Geophysical Research: Oceans*, 105(C8):19477–19498.

- Duchon, C. E. (1979). Lanczos Filtering in One and Two Dimensions. Journal of Applied Meteorology and Climatology, 18(8):1016–1022.
- Ebisuzaki, W. (1997). A Method to Estimate the Statistical Significance of a Correlation When the Data Are Serially Correlated. *Journal of Climate*, 10(9):2147–2153.
- Eden, C. and Willebrand, J. (2001). Mechanism of interannual to decadal variability of the north atlantic circulation. *Journal of Climate*, 14(10):2266 2280.
- Esselborn, S. and Eden, C. (2001). Sea surface height changes in the north atlantic ocean related to the north atlantic oscillation. *Geophysical Research Letters*, 28(18):3473–3476.
- Ezer, T. (2015). Detecting changes in the transport of the gulf stream and the atlantic overturning circulation from coastal sea level data: The extreme decline in 2009–2010 and estimated variations for 1935–2012. *Global and Planetary Change*, 129:23–36.
- Ezer, T. and Mellor, G. L. (1992). numerical study of the variability and the separation of the gulf stream, induced by surface atmospheric forcing and lateral boundary flows. *Journal of physical oceanography*, 22(6):660–682.
- Fabiano, F., Christensen, H. M., Strommen, K., Athanasiadis, P., Baker, A., Schiemann, R., and Corti, S. (2020). Euro-atlantic weather regimes in the primavera coupled climate simulations: impact of resolution and mean state biases on model performance. *Climate Dynamics*, 54(11):5031–5048.
- Farge, M. (1992). Wavelet transforms and their applications to turbulence. Annual Review of Fluid Mechanics, 24(1):395–458.

- Fasullo, J. T. and Trenberth, K. E. (2008). The annual cycle of the energy budget. part ii: Meridional structures and poleward transports. *Journal of Climate*, 21(10):2313 – 2325.
- Feldstein, S. B. (2000). The timescale, power spectra, and climate noise properties of teleconnection patterns. *Journal of Climate*, 13(24):4430 – 4440.
- Feliks, Y., Ghil, M., and Robertson, A. W. (2011). The Atmospheric Circulation over the North Atlantic as Induced by the SST Field. *Journal of Climate*, 24(2):522– 542.
- Feliks, Y., Ghil, M., and Simonnet, E. (2004). Low-Frequency Variability in the Midlatitude Atmosphere Induced by an Oceanic Thermal Front. Journal of the Atmospheric Sciences, 61(9):961–981.
- Frankignoul, C., de Coëtlogon, G., Joyce, T. M., and Dong, S. (2001). Gulf Stream Variability and Ocean–Atmosphere Interactions. *Journal of Physical Oceanography*, 31(12):3516–3529.
- Frankignoul, C. and Hasselmann, K. (1977). Stochastic climate models, Part II Application to sea-surface temperature anomalies and thermocline variability. *Tellus*, 29(4):289–305.
- Fu, L.-L. and Qiu, B. (2002). Low-frequency variability of the north pacific ocean: The roles of boundary- and wind-driven baroclinic rossby waves. *Journal of Geophysical Research: Oceans*, 107(C12):13–1–13–10.
- Gangopadhyay, A., Chaudhuri, A. H., and Taylor, A. H. (2016). On the nature of temporal variability of the gulf stream path from 75° to 55°w. *Earth Interactions*, 20(9):1 17.
- Gangopadhyay, A., Cornillon, P., and Watts, D. R. (1992). A test of the parsons-veronis hypothesis on the separation of the gulf stream. *Journal of Physical Oceanography*, 22(11):1286 – 1301.

- Gastineau, G., García-Serrano, J., and Frankignoul, C. (2017). The influence of autumnal eurasian snow cover on climate and its link with arctic sea ice cover. *Journal of Climate*, 30(19):7599 – 7619.
- Georgiou, S., Ypma, S. L., Brüggemann, N., Sayol, J.-M., Pietrzak, J. D., and Katsman, C. A. (2020). Pathways of the water masses exiting the labrador sea: The importance of boundary–interior exchanges. *Ocean Modelling*, 150:101623.
- Getzlaff, K., Böning, C. W., and Dengg, J. (2006). Lagrangian perspectives of deep water export from the subpolar north atlantic. *Geophysical Research Letters*, 33(21).
- Ghatak, D., Frei, A., Gong, G., Stroeve, J., and Robinson, D. (2010). On the emergence of an arctic amplification signal in terrestrial arctic snow extent. *Journal* of Geophysical Research: Atmospheres, 115(D24).
- Goddard, P. B., Yin, J., Griffies, S. M., and Zhang, S. (2015). An extreme event of sea-level rise along the northeast coast of north america in 2009–2010. *Nature Communications*, 6:6346.
- Graf, H.-F., Kirchner, I., and Perlwitz, J. (1998). Changing lower stratospheric circulation: The role of ozone and greenhouse gases. *Journal of Geophysical Research: Atmospheres*, 103(D10):11251–11261.
- Grinsted, A., Moore, J. C., and Jevrejeva, S. (2004). Application of the cross wavelet transform and wavelet coherence to geophysical time series. *Nonlinear Processes* in Geophysics, 11(5/6):561–566.
- Haarsma, R., Acosta, M., Bakhshi, R., Bretonnière, P. A. B., Caron, L. P., Castrillo,
 M., Corti, S., Davini, P., Exarchou, E., Fabiano, F., Fladrich, U., Fuentes Franco,
 R., García-Serrano, J., von Hardenberg, J., Koenigk, T., Levine, X., Meccia,
 V., van Noije, T., van den Oord, G., Palmeiro, F. M., Rodrigo, M., RuprichRobert, Y., Le Sager, P., Tourigny, E., Wang, S., van Weele, M., and Wyser, K.

(2020). HighResMIP versions of EC-EARTH: EC-EARTH3P and EC-EARTH3P-HR. Description, model performance, data handling and validation. *Geoscientific Model Development Discussions*, pages 1–37.

- Haarsma, R. J., Roberts, M. J., Vidale, P. L., Senior, C. A., Bellucci, A., Bao, Q., Chang, P., Corti, S., Fučkar, N. S., Guemas, V., von Hardenberg, J., Hazeleger, W., Kodama, C., Koenigk, T., Leung, L. R., Lu, J., Luo, J.-J., Mao, J., Mizielinski, M. S., Mizuta, R., Nobre, P., Satoh, M., Scoccimarro, E., Semmler, T., Small, J., and von Storch, J.-S. (2016). High Resolution Model Intercomparison Project (HighResMIP v1.0) for CMIP6. *Geoscientific Model Development*, 9(11):4185– 4208.
- Halkin, D. and Rossby, T. (1985). The structure and transport of the gulf stream at 73°w. Journal of Physical Oceanography, 15(11):1439 1452.
- Hameed, S. and Piontkovski, S. (2004). The dominant influence of the Icelandic Low on the position of the Gulf Stream northwall. *Geophysical Research Letters*, 31(9).
- Hameed, S., Wolfe, C. L. P., and Chi, L. (2018). Impact of the atlantic meridional mode on gulf stream north wall position. *Journal of Climate*, 31(21):8875 – 8894.
- Hashizume, H., Xie, S.-P., Liu, W. T., and Takeuchi, K. (2001). Local and remote atmospheric response to tropical instability waves: A global view from space. *Journal of Geophysical Research: Atmospheres*, 106(D10):10173–10185.
- Held, I. M., Ting, M., and Wang, H. (2002). Northern winter stationary waves: Theory and modeling. *Journal of Climate*, 15(16):2125 – 2144.
- Hersbach, H., Bell, B., Berrisford, P., Hirahara, S., Horányi, A., Muñoz-Sabater, J.,
 Nicolas, J., Peubey, C., Radu, R., Schepers, D., Simmons, A., Soci, C., Abdalla,
 S., Abellan, X., Balsamo, G., Bechtold, P., Biavati, G., Bidlot, J., Bonavita, M.,
 De Chiara, G., Dahlgren, P., Dee, D., Diamantakis, M., Dragani, R., Flemming,
 J., Forbes, R., Fuentes, M., Geer, A., Haimberger, L., Healy, S., Hogan, R. J.,

Hólm, E., Janisková, M., Keeley, S., Laloyaux, P., Lopez, P., Lupu, C., Radnoti,
G., de Rosnay, P., Rozum, I., Vamborg, F., Villaume, S., and Thépaut, J.-N.
(2020). The ERA5 global reanalysis. *Quarterly Journal of the Royal Meteorological* Society, 146(730):1999–2049.

- Hogg, N. G. (1992). On the transport of the gulf stream between cape hatteras and the grand banks. *Deep Sea Research Part A. Oceanographic Research Papers*, 39(7):1231–1246.
- Hogg, N. G., Pickart, R. S., Hendry, R. M., and Smethie, W. J. (1986). The northern recirculation gyre of the gulf stream. *Deep Sea Research Part A. Oceanographic Research Papers*, 33(9):1139–1165.
- Holliday, N., Bersch, M., and Berx, B. e. a. (2020). Ocean circulation causes the largest freshening event for 120 years in eastern subpolar north atlantic. *Nature communication*, 11:585.
- Honda, M. and Nakamura, H. (2001). Interannual seesaw between the aleutian and icelandic lows. part ii: Its significance in the interannual variability over the wintertime northern hemisphere. *Journal of Climate*, 14(24):4512 – 4529.
- Honda, M., Nakamura, H., Ukita, J., Kousaka, I., and Takeuchi, K. (2001). Interannual seesaw between the aleutian and icelandic lows. part i: Seasonal dependence and life cycle. *Journal of Climate*, 14(6):1029 – 1042.
- Hoskins, B. J., James, I. N., and White, G. H. (1983). The Shape, Propagation and Mean-Flow Interaction of Large-Scale Weather Systems. *Journal of the Atmospheric Sciences*, 40(7):1595–1612.
- Hoskins, B. J. and Karoly, D. J. (1981). The Steady Linear Response of a Spherical Atmosphere to Thermal and Orographic Forcing. *Journal of the Atmospheric Sciences*, 38(6):1179–1196.
- Hoskins, B. J. and Valdes, P. J. (1990). On the Existence of Storm-Tracks. Journal of the Atmospheric Sciences, 47(15):1854–1864.

- Hotta, D. and Nakamura, H. (2011). On the Significance of the Sensible Heat Supply from the Ocean in the Maintenance of the Mean Baroclinicity along Storm Tracks. *Journal of Climate*, 24(13):3377–3401.
- Hovmöller, E. (1949). The trough-and-ridge diagram. Tellus, 1(2):62–66.
- Hudgins, L., Friehe, C. A., and Mayer, M. E. (1993). Wavelet transforms and atmospheric turbulence. *Physical Review Letters*, 71:3279–3282.
- Hurrell, J. W. (1995). Decadal trends in the north atlantic oscillation: Regional temperatures and precipitation. *Science*, 269(5224):676–679.
- Hurrell, J. W. and Deser, C. (2010). North atlantic climate variability: The role of the north atlantic oscillation. *Journal of Marine Systems*, 79(3):231–244. Impact of climate variability on marine ecosystems: A comparative approach.
- Hurrell, J. W., Kushnir, Y., Ottersen, G., and Visbeck, M. (2003). The North Atlantic Oscillation: Climatic Significance and Environmental Impact. Washington DC American Geophysical Union Geophysical Monograph Series, 134.
- Imawaki, S., Bower, A. S., Beal, L., and Qiu, B. (2013). Chapter 13 western boundary currents. In Siedler, G., Griffies, S. M., Gould, J., and Church, J. A., editors, Ocean Circulation and Climate, volume 103 of International Geophysics, pages 305–338. Academic Press.
- Jianping, L. and Wang, J. X. L. (2003). A new north atlantic oscillation index and its variability. Advances in Atmospheric Sciences, 20(5):661–676.
- Johns, W. E., Shay, T. J., Bane, J. M., and Watts, D. R. (1995). Gulf stream structure, transport, and recirculation near 68°w. Journal of Geophysical Research: Oceans, 100(C1):817–838.
- Jones, P. D., Jonsson, T., and Wheeler, D. (1997). Extension to the north atlantic oscillation using early instrumental pressure observations from gibraltar and south-west iceland. *International Journal of Climatology*, 17(13):1433–1450.
- Joyce, T. M., Deser, C., and Spall, M. A. (2000). The Relation between Decadal Variability of Subtropical Mode Water and the North Atlantic Oscillation. *Journal* of Climate, 13(14):2550–2569.
- Joyce, T. M., Kwon, Y.-O., Seo, H., and Ummenhofer, C. C. (2019). Meridional Gulf Stream Shifts Can Influence Wintertime Variability in the North Atlantic Storm Track and Greenland Blocking. *Geophysical Research Letters*, 46:1702–1708.
- Joyce, T. M., Kwon, Y.-O., and Yu, L. (2009). On the Relationship between Synoptic Wintertime Atmospheric Variability and Path Shifts in the Gulf Stream and the Kuroshio Extension. *Journal of Climate*, 22(12):3177–3192.
- Joyce, T. M. and Zhang, R. (2010). On the path of the gulf stream and the atlantic meridional overturning circulation. *Journal of Climate*, 23(11):3146 3154.
- Jung, T., Miller, M. J., Palmer, T. N., Towers, P., Wedi, N., Achuthavarier, D., Adams, J. M., Altshuler, E. L., Cash, B. A., Kinter, J. L., Marx, L., Stan, C., and Hodges, K. I. (2012). High-resolution global climate simulations with the ecmwf model in project athena: Experimental design, model climate, and seasonal forecast skill. *Journal of Climate*, 25(9):3155 – 3172.
- Kallberg, P., Berrisford, P., Hoskins, B. J., Simmons, A., Uppala, S., Lamy-Thepaut,S., and Hine, R. (2005). ERA-40 Atlas. ERA-40 Project Report Series. 19.
- Kelly, K. A. and Qiu, B. (1995). Heat Flux Estimates for the Western North Atlantic. Part II: The Upper-Ocean Heat Balance. *Journal of Physical Oceanography*, 25(10):2361–2373.
- Kelly, K. A., Small, R. J., Samelson, R. M., Qiu, B., Joyce, T. M., Kwon, Y.-O., and Cronin, M. F. (2010). Western Boundary Currents and Frontal Air–Sea Interaction: Gulf Stream and Kuroshio Extension. *Journal of Climate*, 23(21):5644–5667.
- Kenigson, J. S., Han, W., Rajagopalan, B., Yanto, and Jasinski, M. (2018). Decadal shift of nao-linked interannual sea level variability along the u.s. northeast coast. *Journal of Climate*, 31(13):4981 – 4989.

- Kennedy, J., Titchner, H., Rayner, N., and Roberts, M. (2017). input4MIPs.MOHC.SSTsAndSeaIce.HighResMIP.MOHC-HadISST-2-2-0-0-0, version 20170505. *Earth System Grid Federation*. accessed 5 May 2017.
- Kohyama, T., Yamagami, Y., Miura, H., Kido, S., Tatebe, H., and Watanabe, M. (2021). The gulf stream and kuroshio current are synchronized. *Science*, 374(6565):341–346.
- Kowalski, P. (2022). On the contribution of rossby waves driven by surface buoyancy fluxes to low-frequency north atlantic steric sea surface height variations. *Atmospheric and Oceanic Science Letters*, 15(3):100153.
- Krahmann, G., Visbeck, M., and Reverdin, G. (2001). Formation and propagation of temperature anomalies along the north atlantic current. *Journal of Physical Oceanography*, 31(5):1287 – 1303.
- Krauss, W. (1986). The north atlantic current. Journal of Geophysical Research: Oceans, 91(C4):5061–5074.
- Krauss, W., KäSe, R. H., and Hinrichsen, H.-H. (1990). The branching of the gulf stream southeast of the grand banks. *Journal of Geophysical Research: Oceans*, 95(C8):13089–13103.
- Kretschmer, M., Coumou, D., Agel, L., Barlow, M., Tziperman, E., and Cohen, J. (2018). More-persistent weak stratospheric polar vortex states linked to cold extremes. *Bulletin of the American Meteorological Society*, 99(1):49 – 60.
- Kushnir, Y., Robinson, W. A., Bladé, I., Hall, N. M. J., Peng, S., and Sutton, R. (2002). Atmospheric GCM Response to Extratropical SST Anomalies: Synthesis and Evaluation. *Journal of Climate*, 15(16):2233–2256.
- Kuwano-Yoshida, A., Minobe, S., and Xie, S.-P. (2010). Precipitation response to the gulf stream in an atmospheric gcm. *Journal of Climate*, 23(13):3676 3698.

- Kvamstø, N. G., Skeie, P., and Stephenson, D. B. (2004). Impact of labrador seaice extent on the north atlantic oscillation. *International Journal of Climatology*, 24(5):603–612.
- Kwon, Y., Alexander, M., Bond, N., Frankignoul, C., Nakamura, H., Qiu, B., and Thompson, L. (2010). Role of the Gulf Stream and Kuroshio–Oyashio Systems in Large-Scale Atmosphere–Ocean Interaction: A Review. *Journal of Climate*, 23(12):3249–3281.
- Kwon, Y.-O. and Joyce, T. M. (2013). Northern Hemisphere Winter Atmospheric Transient Eddy Heat Fluxes and the Gulf Stream and Kuroshio–Oyashio Extension Variability. *Journal of Climate*, 26(24):9839–9859.
- Lazier, J., Pickart, R., and Rhines, P. (2001). Chapter 5.5 deep convection. In Siedler, G., Church, J., and Gould, J., editors, *Ocean Circulation and Climate*, volume 77 of *International Geophysics*, pages 387–400. Academic Press.
- Lee, T. and Cornillon, P. (1995). Temporal variation of meandering intensity and domain-wide lateral oscillations of the gulf stream. *Journal of Geophysical Research: Oceans*, 100(C7):13603–13613.
- Leith, C. E. (1973). The standard error of time-average estimates of climatic means. Journal of Applied Meteorology and Climatology, 12(6):1066 – 1069.
- Lillibridge, J. L. and Mariano, A. J. (2013). A statistical analysis of gulf stream variability from 18+ years of altimetry data. *Deep Sea Research Part II: Topical Studies in Oceanography*, 85:127–146. Modern Physical Oceanography and Professor H.T. Rossby.
- Lozier, M. S. (2010). Deconstructing the conveyor belt. *Science*, 328(5985):1507–1511.
- Lubis, S. W., Huang, C. S. Y., Nakamura, N., Omrani, N.-E., and Jucker, M. (2018). Role of finite-amplitude rossby waves and nonconservative processes in downward

migration of extratropical flow anomalies. Journal of the Atmospheric Sciences, 75(5):1385 – 1401.

- Ma, X., Ping Chang, P., Saravanan, R., Montuoro, R., Nakamura, H., Wu, D., Lin, X., and L., W. (2017). Importance of Resolving Kuroshio Front and Eddy Influence in Simulating the North Pacific Storm Track. *Journal of Climate*, 30(5):1861– 1880.
- Madden, R. A. (1976). Estimates of the natural variability of time-averaged sea-level pressure. *Monthly Weather Review*, 104(7):942 952.
- Madonna, E., Li, C., Grams, C. M., and Woollings, T. (2017). The link between eddy-driven jet variability and weather regimes in the north atlantic-european sector. Quarterly Journal of the Royal Meteorological Society, 143(708):2960– 2972.
- Maloney, E. D. and Chelton, D. B. (2006). An assessment of the sea surface temperature influence on surface wind stress in numerical weather prediction and climate models. *Journal of Climate*, 19(12):2743 – 2762.
- Maraun, D. and Kurths, J. (2004). Cross wavelet analysis: significance testing and pitfalls. Nonlinear Processes in Geophysics, 11(4):505–514.
- Marshall, J., Johnson, H., and Goodman, J. (2001). A study of the interaction of the north atlantic oscillation with ocean circulation. *Journal of Climate*, 14(7):1399 – 1421.
- Marshall, J. and Schott, F. (1999). Open-ocean convection: Observations, theory, and models. *Reviews of Geophysics*, 37(1):1–64.
- Matsueda, M., Mizuta, R., and Kusunoki, S. (2009). Future change in wintertime atmospheric blocking simulated using a 20-km-mesh atmospheric global circulation model. *Journal of Geophysical Research: Atmospheres*, 114(D12).

- McCarthy, G. D., Joyce, T. M., and Josey, S. A. (2018). Gulf stream variability in the context of quasi-decadal and multidecadal atlantic climate variability. *Geophysical Research Letters*, 45(20):11,257–11,264.
- Minobe, S., Kuwano-Yoshida, A., Komori, N., et al. (2008a). Influence of the Gulf Stream on the troposphere. *Nature*, 452:206–209.
- Minobe, S., Kuwano-Yoshida, A., Komori, N., Xie, S.-P., and Small, J. (2008b). The gulf stream's pathway to impact climate. Newsletter of the International Pacific Research Center, 8(1):3 – 8.
- Minobe, S., Miyashita, M., Kuwano-Yoshida, A., Tokinaga, H., and Xie, S.-P. (2010). Atmospheric response to the gulf stream: Seasonal variations. *Journal of Climate*, 23(13):3699 – 3719.
- Molinari, R. L., Fine, R. A., Wilson, W. D., Curry, R. G., Abell, J., and McCartney,
 M. S. (1998). The arrival of recently formed labrador sea water in the deep western boundary current at 26.5°n. *Geophysical Research Letters*, 25(13):2249–2252.
- Munk, W. H. (1950). On the wind-driven ocean circulation. Journal of Atmospheric Sciences, 7(2):80–93.
- Mysterud, A., Stenseth, N. C., Yoccoz, N. G., Ottersen, G., and Langvatn, R. (2003). The Response of Terrestrial Ecosystems to Climate Variability Associated with the North Atlantic Oscillation, pages 235–262. American Geophysical Union (AGU).
- Nakamura, H., Sampe, T., Goto, A., Ohfuchi, W., and Xie, S.-P. (2008). On the importance of midlatitude oceanic frontal zones for the mean state and dominant variability in the tropospheric circulation. *Geophysical Research Letters*, 35(15).
- Nakamura, H., Sampe, T., Tanimoto, Y., and Shimpo, A. (2004). Observed Associations Among Storm Tracks, Jet Streams and Midlatitude Oceanic Fronts, pages 329–345. Amer. Geophys. Union.

- Nakamura, M. and Yamane, S. (2009). Dominant Anomaly Patterns in the Near-Surface Baroclinicity and Accompanying Anomalies in the Atmosphere and Oceans. Part I: North Atlantic Basin. *Journal of Climate*, 22(4):880–904.
- Nakamura, M. and Yamane, S. (2010). Dominant Anomaly Patterns in the Near-Surface Baroclinicity and Accompanying Anomalies in the Atmosphere and Oceans. Part II: North Pacific Basin. *Journal of Climate*, 23(24):6445–6467.
- New, A. L., Smeed, D. A., Czaja, A., Blaker, A. T., Mecking, J. V., Mathews, J. P., and Sanchez-Franks, A. (2021). Labrador slope water connects the subarctic with the gulf stream. *Environmental Research Letters*, 16(8):084019.
- Nigam, S., Ruiz-Barradas, A., and Chafik, L. (2018). Gulf stream excursions and sectional detachments generate the decadal pulses in the atlantic multidecadal oscillation. *Journal of Climate*, 31(7):2853 – 2870.
- Nonaka, M. and Xie, S.-P. (2003). Covariations of sea surface temperature and wind over the kuroshio and its extension: Evidence for ocean-to-atmosphere feedback. *Journal of Climate*, 16(9):1404–1413.
- Novak, L., Ambaum, M. H. P., and Tailleux, R. (2015). The Life Cycle of the North Atlantic Storm Track. *Journal of the Atmospheric Sciences*, 72(2):821–833.
- Ogawa, F., Omrani, N.-E., Nishii, K., Nakamura, H., and Keenlyside, N. (2015). Ozone-induced climate change propped up by the southern hemisphere oceanic front. *Geophysical Research Letters*, 42(22):10,056–10,063.
- Omrani, N., Keenlyside, N., and Matthes, K. e. a. (2022). Coupled stratospheretroposphere-atlantic multidecadal oscillation and its importance for near-future climate projection. *npj Climate and Atmospheric Science*, 5:59.
- Omrani, N.-E., Keenlyside, N. S., Bader, J., and Manzini, E. (2014). Stratosphere key for wintertime atmospheric response to warm atlantic decadal conditions. *Climate Dynamics*, 42(3):649–663.

- Omrani, N.-E., Ogawa, F., Nakamura, H., Keenlyside, N., Lubis, S. W., and Matthes, K. (2019). Key Role of the Ocean Western Boundary currents in shaping the Northern Hemisphere climate. *Scientific Reports*, 9(1):3014.
- O'Neill, L. W., Chelton, D. B., and Esbensen, S. K. (2003). Observations of sstinduced perturbations of the wind stress field over the southern ocean on seasonal timescales. *Journal of Climate*, 16(14):2340 – 2354.
- O'Reilly, C. H. and Czaja, A. (2015). The response of the Pacific storm track and atmospheric circulation to Kuroshio Extension variability. *Quarterly Journal of* the Royal Meteorological Society, 141(686):52–66.
- O'Reilly, C. H., Minobe, S., and Kuwano-Yoshida, A. (2016). Theincluence of the Gulf Stream on wintertime European blocking. *Climate Dynamics*, 47:1545–1567.
- O'Reilly, C. H., Minobe, S., Kuwano-Yoshida, A., and Woollings, T. (2017). The Gulf Stream influence on wintertime North Atlantic jet variability. *Quarterly Journal of the Royal Meteorological Society*, 143(702):173–183.
- Orlanski, I. (2003). Bifurcation in Eddy Life Cycles: Implications for Storm Track Variability. Journal of the Atmospheric Sciences, 60(8):993–1023.
- Osychny, V. and Cornillon, P. (2004). Properties of rossby waves in the north atlantic estimated from satellite data. *Journal of Physical Oceanography*, 34(1):61 76.
- O'Neill, L. W., Chelton, D. B., Esbensen, S. K., and Wentz, F. J. (2005). Highresolution satellite measurements of the atmospheric boundary layer response to sst variations along the agulhas return current. *Journal of Climate*, 18(14):2706 – 2723.
- Palmer, T. N. (1993). A nonlinear dynamical perspective on climate change. Weather, 48(10):314–326.
- Pan, J., Yan, X.-H., Zheng, Q., and Liu, W. T. (2002). Observation of western

boundary current atmospheric convergence zones using scatterometer winds. *Geo-physical Research Letters*, 29(17):13–1–13–4.

- Parfitt, R. and Czaja, A. (2016). On the contribution of synoptic transients to the mean atmospheric state in the gulf stream region. *Quarterly Journal of the Royal Meteorological Society*, 142(696):1554–1561.
- Parfitt, R., Czaja, A., Minobe, S., and Kuwano-Yoshida, A. (2016). The atmospheric frontal response to sst perturbations in the gulf stream region. *Geophysical Re*search Letters, 43(5):2299–2306.
- Parfitt, R. and Seo, H. (2018). A New Framework for Near-Surface Wind Convergence Over the Kuroshio Extension and Gulf Stream in Wintertime: The Role of Atmospheric Fronts. *Geophysical Research Letters*, 45(18):9909–9918.
- Parsons, A. T. (1969). A two-layer model of gulf stream separation. Journal of Fluid Mechanics, 39(3):511–528.
- Patrizio, C. R. and Thompson, D. W. (2021). Quantifying the Role of Ocean Dynamics in Ocean Mixed-Layer Temperature Variability. *Journal of Climate*, pages 1–63.
- Peings, Y. and Magnusdottir, G. (2014). Forcing of the wintertime atmospheric circulation by the multidecadal fluctuations of the north atlantic ocean. *Environmental Research Letters*, 9(3):034018.
- Peixoto, J. P. and Oort, A. H. (1992). Physics of climate / Jose P. Peixoto and Abraham H. Oort ; foreword by Edward N. Lorenz. American Institute of Physics New York.
- Peng, S., Robinson, W. A., and Hoerling, M. P. (1997). The Modeled Atmospheric Response to Midlatitude SST Anomalies and Its Dependence on Background Circulation States. *Journal of Climate*, 10(5):971–987.

- Peng, S., Robinson, W. A., and Li, S. (2003). Mechanisms for the nao responses to the north atlantic sst tripole. *Journal of Climate*, 16(12):1987 – 2004.
- Peng, S. and Whitaker, J. S. (1999). Mechanisms Determining the Atmospheric Response to Midlatitude SST Anomalies. *Journal of Climate*, 12(5):1393–1408.
- Percival, D. P. (1995). On estimation of the wavelet variance. *Biometrika*, 82(3):619–631.
- Peterson, I., Greenan, B., Gilbert, D., and Hebert, D. (2017). Variability and wind forcing of ocean temperature and thermal fronts in the slope water region of the northwest atlantic. *Journal of Geophysical Research: Oceans*, 122(9):7325–7343.
- Peña-Molino, B. and Joyce, T. M. (2008). Variability in the slope water and its relation to the gulf stream path. *Geophysical Research Letters*, 35(3).
- Peña-Molino, B., Joyce, T. M., and Toole, J. M. (2011). Recent changes in the labrador sea water within the deep western boundary current southeast of cape cod. Deep Sea Research Part I: Oceanographic Research Papers, 58(10):1019–1030.
- Pickart, R. S. (1992). Water mass components of the north atlantic deep western boundary current. Deep Sea Research Part A. Oceanographic Research Papers, 39(9):1553–1572.
- Pickart, R. S. and Smethie, W. M. (1993). How does the deep western boundary current cross the gulf stream? *Journal of Physical Oceanography*, 23(12):2602 – 2616.
- Pérez-Hernández, M. D. and Joyce, T. M. (2014). Two modes of gulf stream variability revealed in the last two decades of satellite altimeter data. *Journal of Physical Oceanography*, 44(1):149 – 163.
- Qiu, B. and Huang, R. X. (1995). Ventilation of the north atlantic and north pacific: Subduction versus obduction. Journal of Physical Oceanography, 25(10):2374 – 2390.

- Quattrocchi, G., Pierini, S., and Dijkstra, H. A. (2012). Intrinsic low-frequency variability of the gulf stream. *Nonlinear Processes in Geophysics*, 19(2):155–164.
- Reintges, A., Latif, M., and Park, W. (2017). Sub-decadal North Atlantic Oscillation variability in observations and the Kiel Climate Model. *Climate Dynamics*, 48:3475–3487.
- Roberts, C. D., Senan, R., Molteni, F., Boussetta, S., Mayer, M., and Keeley, S.
 P. E. (2018). Climate model configurations of the ECMWF Integrated Forecasting System (ECMWF-IFS cycle 43R1) for HighResMIP. *Geoscientific Model Development*, 11:3681–3712.
- Roberts, M. J., Baker, A., Blockley, E. W., Calvert, D., Coward, A., Hewitt, H. T., Jackson, L. C., Kuhlbrodt, T., Mathiot, P., Roberts, C. D., Schiemann, R., Seddon, J., Vannière, B., and Vidale, P. L. (2019). Description of the resolution hierarchy of the global coupled HadGEM3-GC3.1 model as used in CMIP6 High-ResMIP experiments. *Geoscientific Model Development*, 12:4999–5028.
- Robertson, A. W., Mechoso, C. R., and Kim, Y.-J. (2000). The influence of atlantic sea surface temperature anomalies on the north atlantic oscillation. *Journal of Climate*, 13(1):122 – 138.
- Rodwell, M. J., Rowell, D. P., and Folland, C. K. (1999). Oceanic forcing of the wintertime north atlantic oscillation and european climate. *Nature*, 398(6725):320– 323.
- Rogers, J. C. (1984). The association between the north atlantic oscillation and the southern oscillation in the northern hemisphere. *Monthly Weather Review*, 112(10):1999 – 2015.
- Rossby, T. (1999). On gyre interactions. Deep Sea Research Part II: Topical Studies in Oceanography, 46(1):139–164.
- Rossby, T. and Benway, R. L. (2000). Slow variations in mean path of the gulf stream east of cape hatteras. *Geophysical Research Letters*, 27(1):117–120.

- Rossby, T., Flagg, C. N., and Donohue, K. (2005). Interannual variations in upperocean transport by the gulf stream and adjacent waters between new jersey and bermuda. *Journal of Marine Research*, 63(1):203–226.
- Sampe, T., Nakamura, H., Goto, A., and Ohfuchi, W. (2010). Significance of a midlatitude sst frontal zone in the formation of a storm track and an eddy-driven westerly jet. *Journal of Climate*, 23(7):1793 – 1814.
- Sanchez-Franks, A., Hameed, S., and Wilson, R. E. (2016). The Icelandic Low as a Predictor of the Gulf Stream North Wall Position. *Journal of Physical Oceanography*, 46(3):817–826.
- Sanchez-Franks, A. and Zhang, R. (2015). Impact of the atlantic meridional overturning circulation on the decadal variability of the gulf stream path and regional chlorophyll and nutrient concentrations. *Geophysical Research Letters*, 42(22):9889–9887.
- Sasaki, Y. N. and Schneider, N. (2011). Interannual to decadal gulf stream variability in an eddy-resolving ocean model. *Ocean Modelling*, 39(3):209–219.
- Sato, K., Inoue, J., and Watanabe, M. (2014). Influence of the gulf stream on the barents sea ice retreat and eurasian coldness during early winter. *Environmental Research Letters*, 9(8):084009.
- Seo, H., Kwon, Y.-O., Joyce, T. M., and Ummenhofer, C. C. (2017). On the predominant nonlinear response of the extratropical atmosphere to meridional shifts of the gulf stream. *Journal of Climate*, 30(23):9679 – 9702.
- Sheldon, L., Czaja, A., Vannière, B., Morcrette, C., Sohet, B., Casado, M., and Smith, D. (2017). A 'warm path' for Gulf Stream-troposphere interactions. *Tellus* A: Dynamic Meteorology and Oceanography, 69(1):1299397.
- Shindell, D. T., Schmidt, G. A., Miller, R. L., and Rind, D. (2001). Northern hemisphere winter climate response to greenhouse gas, ozone, solar, and volcanic forcing. *Journal of Geophysical Research: Atmospheres*, 106(D7):7193–7210.

- Smagorinsky, J. (1953). The dynamical influence of large-scale heat sources and sinks on the quasi-stationary mean motions of the atmosphere. Quarterly Journal of the Royal Meteorological Society, 79(341):342–366.
- Small, R., deSzoeke, S., Xie, S., O'Neill, L., Seo, H., Song, Q., Cornillon, P., Spall,
 M., and Minobe, S. (2008). Air-sea interaction over ocean fronts and eddies.
 Dynamics of Atmospheres and Oceans, 45(3):274–319. OCEANIC FRONTS.
- Smethie, W. M. (1993). Tracing the thermohaline circulation in the western north atlantic using chlorofluorocarbons. *Progress in Oceanography*, 31(1):51–99.
- Smirnov, D., Newman, M., and Alexander, M. A. (2014). Investigating the Role of Ocean–Atmosphere Coupling in the North Pacific Ocean. *Journal of Climate*, 27(2):592–606.
- Smirnov, D., Newman, M., Alexander, M. A., Kwon, Y.-O., and Frankignoul, C. (2015). Investigating the Local Atmospheric Response to a Realistic Shift in the Oyashio Sea Surface Temperature Front. *Journal of Climate*, 28(3):1126–1147.
- Spall, M. A. (1996a). Dynamics of the gulf stream/deep western boundary current crossover. part i: Entrainment and recirculation. Journal of Physical Oceanography, 26(10):2152 – 2168.
- Spall, M. A. (1996b). Dynamics of the gulf stream/deep western boundary current crossover. part ii: Low-frequency internal oscillations. Journal of Physical Oceanography, 26(10):2169 – 2182.
- Stephenson, D. B., Pavan, V., and Bojariu, R. (2000). Is the north atlantic oscillation a random walk? *International Journal of Climatology*, 20(1):1–18.
- Stommel, H. (1948). The westward intensification of wind-driven ocean currents. Eos, Transactions American Geophysical Union, 29(2):202–206.
- Stommel, H. (1958). The abyssal circulation. Deep Sea Research (1953), 5(1):80–82.

- Stone, P. H. (1978). Baroclinic Adjustment. Journal of the Atmospheric Sciences, 35(4):561–571.
- Stull, R. B. (1998). An Introduction to Boundary Layer Meteorology. Springer Dordrecht.
- Sturges, W. and Hong, B. G. (1995). Wind forcing of the atlantic thermocline along 32°n at low frequencies. *Journal of Physical Oceanography*, 25(7):1706 – 1715.
- Sturges, W., Hong, B. G., and Clarke, A. J. (1998). Decadal wind forcing of the north atlantic subtropical gyre. *Journal of Physical Oceanography*, 28(4):659 – 668.
- Sutton, R. T. and Allen, M. R. (1997). Decadal predictability of north atlantic sea surface temperature and climate. *Nature*, 388(6642):563–567.
- Sutton, R. T., Norton, W. A., and Jewson, S. P. (2000). The north atlantic oscillation—what role for the ocean? *Atmospheric Science Letters*, 1(2):89–100.
- Sverdrup, H. U. (1947). Wind-driven currents in a baroclinic ocean; with application to the equatorial currents of the eastern pacific*. *Proceedings of the National Academy of Sciences*, 33(11):318–326.
- Talley, L. D. and McCartney, M. S. (1982). Distribution and circulation of labrador sea water. Journal of Physical Oceanography, 12(11):1189 – 1205.
- Talley, L. D., Pickard, G. L., Emery, W. J., and Swift, J. H. (2011). Chapter 7 dynamical processes for descriptive ocean circulation. In Talley, L. D., Pickard, G. L., Emery, W. J., and Swift, J. H., editors, *Descriptive Physical Oceanography* (Sixth Edition), pages 187–221. Academic Press, Boston, sixth edition edition.
- Tansley, C. E. and Marshall, D. P. (2000). On the influence of bottom topography and the deep western boundary current on gulf stream separation. *Journal of Marine Research*, 58(2):297–325.

- Taylor, A., Jordan, M., and Stephens, J. (1998). Gulf Stream shifts following ENSO events. *Nature*, 393:638.
- Taylor, A. and Stephens, J. A. (1980). Latitudinal displacements of the gulf-stream (1966 to 1977) and their relation to changes in temperature and zooplankton abundance in the ne atlantic. *Oceanologica Acta*, 3:145–149.
- Taylor, A. H. and Gangopadhyay, A. (2001). A simple model of interannual displacements of the gulf stream. *Journal of Geophysical Research: Oceans*, 106(C7):13849–13860.
- Taylor, A. H. and Stephens, J. A. (1998). The North Atlantic Oscillation and the latitude of the Gulf Stream. *Tellus A*, 50(1):134–142.
- Thiéblemont, R., Matthes, K., Omrani, N.-E., Kodera, K., and Hansen, F. (2015). Solar forcing synchronizes decadal north atlantic climate variability. *Nature Communications*, 6(1):8268.
- Thompson, D. W. J., Lee, S., and Baldwin, M. P. (2003). Atmospheric Processes Governing the Northern Hemisphere Annular Mode/North Atlantic Oscillation, pages 81–112. American Geophysical Union (AGU).
- Thompson, D. W. J. and Wallace, J. M. (1998). The arctic oscillation signature in the wintertime geopotential height and temperature fields. *Geophysical Research Letters*, 25(9):1297–1300.
- Thompson, D. W. J. and Wallace, J. M. (2000). Annular modes in the extratropical circulation. part i: Month-to-month variability. *Journal of Climate*, 13(5):1000 – 1016.
- Thompson, J. D. and Schmitz, W. J. (1989). A limited-area model of the gulf stream: Design, initial experiments, and model-data intercomparison. *Journal of Physical Oceanography*, 19(6):791 – 814.

- Torrence, C. and Compo, G. P. (1998). A practical guide to wavelet analysis. *Bulletin* of the American Meteorological Society, 79(1):61 – 78.
- Torrence, C. and Webster, P. J. (1999). Interdecadal changes in the enso-monsoon system. Journal of Climate, 12(8):2679 – 2690.
- Trenberth, K. E., Branstator, G. W., Karoly, D., Kumar, A., Lau, N.-C., and Ropelewski, C. (1998). Progress during toga in understanding and modeling global teleconnections associated with tropical sea surface temperatures. *Journal of Geophysical Research: Oceans*, 103(C7):14291–14324.
- Veronis, G. (1973). Model of world ocean circulation: I. wind-driven, two layer. Journal of Marine Research, 31:228–288.
- Visbeck, M., Chassignet, E. P., Curry, R. G., Delworth, T. L., Dickson, R. R., and Krahmann, G. (2003). The Ocean's Response to North Atlantic Oscillation Variability, pages 113–145. American Geophysical Union (AGU).
- Visbeck, M., Cullen, H., Krahmann, G., and Naik, N. (1998). An ocean model's response to north atlantic oscillation-like wind forcing. *Geophysical Research Letters*, 25(24):4521–4524.
- Walker, G. T. and Bliss, E. W. (1932). World weather v. Memoirs of the Royal Meteorological Society, 4:53–84.
- Wallace, J. M. and Gutzler, D. S. (1981). Teleconnections in the geopotential height field during the northern hemisphere winter. *Monthly Weather Review*, 109(4):784 – 812.
- Wallace, J. M., Mitchell, T. P., and Deser, C. (1989). The influence of sea-surface temperature on surface wind in the eastern equatorial pacific: Seasonal and interannual variability. *Journal of Climate*, 2(12):1492 – 1499.

Watanabe, M. and Kimoto, M. (2000). Atmosphere-ocean thermal coupling in the

North Atlantic: A positive feedback. *Quarterly Journal of the Royal Meteorological Society*, 126(570):3343–3369.

- Watanabe, M. and Nitta, T. (1999). Decadal changes in the atmospheric circulation and associated surface climate variations in the northern hemisphere winter. *Journal of Climate*, 12(2):494 – 510.
- Watelet, S., Beckers, J.-M., and Barth, A. (2017). Reconstruction of the gulf stream from 1940 to the present and correlation with the north atlantic oscillation. *Journal of Physical Oceanography*, 47(11):2741 2754.
- Watelet, S., Beckers, J.-M., Molines, J.-M., and Troupin, C. (2020). Properties of baroclinic rossby waves in the north atlantic from eddy-resolving simulations of ocean circulation. Ocean Science Discussions, 2020:1–18.
- Wegmann, M., Orsolini, Y., Vázquez, M., Gimeno, L., Nieto, R., Bulygina, O., Jaiser, R., Handorf, D., Rinke, A., Dethloff, K., Sterin, A., and Brönnimann, S. (2015). Arctic moisture source for eurasian snow cover variations in autumn. *Environmental Research Letters*, 10(5):054015.
- Wegmann, M., Rohrer, M., Santolaria-Otín, M., and Lohmann, G. (2020). Eurasian autumn snow link to winter north atlantic oscillation is strongest for arctic warming periods. *Earth System Dynamics*, 11(2):509–524.
- Wettstein, J. J. and Wallace, J. M. (2010). Observed patterns of month-to-month storm-track variability and their relationship to the background flow. *Journal of* the Atmospheric Sciences, 67(5):1420 – 1437.
- Willison, J., Robinson, W. A., and Lackmann, G. M. (2013). The Importance of Resolving Mesoscale Latent Heating in the North Atlantic Storm Track. *Journal* of the Atmospheric Sciences, 70(7):2234–2250.
- Wills, R. C. J., Armour, K. C., Battisti, D. S., and Hartmann, D. L. (2019). Ocean–atmosphere dynamical coupling fundamental to the atlantic multidecadal oscillation. *Journal of Climate*, 32(1):251 – 272.

- Wills, S. M., Thompson, D. W. J., and Ciasto, L. M. (2016). On the Observed Relationships between Variability in Gulf Stream Sea Surface Temperatures and the Atmospheric Circulation over the North Atlantic. *Journal of Climate*, 29(10):3719–3730.
- Wolfe, C. L. P., Hameed, S., and Chi, L. (2019). On the drivers of decadal variability of the gulf stream north wall. *Journal of Climate*, 32(4):1235 1249.
- Woollings, T., Franzke, C., Hodson, D. L. R., Dong, B., Barnes, E. A., Raible, C. C., and Pinto, J. G. (2015). Contrasting interannual and multidecadal nao variability. *Climate Dynamics*, 45(1):539–556.
- Woollings, T., Hannachi, A., and Hoskins, B. (2010). Variability of the North Atlantic eddy-driven jet stream. Quarterly Journal of the Royal Meteorological Society, 136(649):856–868.
- Woollings, T. and Hoskins, B. (2008). Simultaneous Atlantic–Pacific blocking and the Northern Annular Mode. Quarterly Journal of the Royal Meteorological Society, 134(636):1635–1646.
- Worthington, L. V. (1976). On the North Atlantic Circulation. Vol. 6. The Johns Hopkins University Press.
- Xie, S.-P. (2004). Satellite observations of cool ocean–atmosphere interaction. Bulletin of the American Meteorological Society, 85(2):195 – 208.
- Yang, H., Li, Q., Wang, K., Sun, Y., and Sun, D. (2015). Decomposing the meridional heat transport in the climate system. *Climate Dynamics*, 44(9):2751–2768.
- Yook, S., Thompson, D. W. J., Sun, L., and Patrizio, C. (2022). The simulated atmospheric response to western north pacific sea surface temperature anomalies. *Journal of Climate*, 35(11):3335 – 3352.
- Yu, L. and Weller, R. A. (2007). Objectively analyzed air-sea heat fluxes for the

global ice-free oceans (1981–2005). Bulletin of the American Meteorological Society, 88(4):527 – 540.

- Zeng, X. and He, R. (2016). Gulf stream variability and a triggering mechanism of its large meander in the south atlantic bight. *Journal of Geophysical Research*, 121:8021–8038.
- Zhang, J., Kelly, K. A., and Thompson, L. (2016). The role of heating, winds, and topography on sea level changes in the north atlantic. *Journal of Geophysical Research: Oceans*, 121(5):2887–2900.
- Zhang, R. and Vallis, G. K. (2006). Impact of great salinity anomalies on the lowfrequency variability of the north atlantic climate. *Journal of Climate*, 19(3):470 – 482.
- Zhang, R. and Vallis, G. K. (2007). The role of bottom vortex stretching on the path of the north atlantic western boundary current and on the northern recirculation gyre. *Journal of Physical Oceanography*, 37(8):2053 – 2080.
- Zuo, H., Balmaseda, M. A., Tietsche, S., Mogensen, K., and Mayer, M. (2019). The ecmwf operational ensemble reanalysis–analysis system for ocean and sea ice: a description of the system and assessment. Ocean Science, 15(3):779–808.
- Årthun, M., Wills, R. C. J., Johnson, H. L., Chafik, L., and Langehaug, H. R. (2021). Mechanisms of decadal north atlantic climate variability and implications for the recent cold anomaly. *Journal of Climate*, 34(9):3421 – 3439.